



Article Prices and Taxes in a Ramsey Climate Policy Model under Heterogeneous Beliefs and Ambiguity

Peter von zur Muehlen †

Federal Reserve Board, Washington, DC 20551, USA; pmuehlen@verizon.net † Retired.

Abstract: In a Ramsey policy regime, heterogeneity in beliefs about the potential costs of climate change is shown to produce policy ambiguities that alter carbon prices and taxation. Three sources of ambiguity are considered: (i) the private sector is skeptical, with beliefs that are unknown to the government, (ii) private agents have pessimistic doubts about the model, or (iii) the policy authority itself does not trust the extant scientific climate model and fears the worst. These three sources of ambiguity give rise to four potential belief regimes characterized by differentials between the government's and the private sector's inter-temporal rates of substitutions, with implications for the prices of carbon and capital, framed in terms of distorted Arrow–Debreu pricing theory that establishes an equivalence between the optimal carbon tax and the permit price of an underlying asset—the government-imposed limit on emissions in economies with cap and trade. This paper shows that in most instances, skeptical beliefs and resulting ambiguities justify higher carbon taxes and lower capital taxes to offset the private sector's increased myopia compared with rational expectations. Conversely, ambiguities created by worst-case fears in either the private sector or in government tend produce forces in the opposite direction.

Keywords: Knightian uncertainty; multiplier preferences; Ramsey planner; social planner; carbon tax; capital tax; ambiguity premium; dynamic stochastic integrated general equilibrium (DSIGE); robust Arrow–Debreu asset prices

JEL Classification: C61; C73; D80; E62; H21; H23; H63; Q32; Q38; Q54

1. Introduction

"The trouble with the world is that the stupid are cocksure and the intelligent are full of doubt." Bertrand Russell

That global warming poses an existential threat to the future of humanity and the planet is now a universally accepted scientific fact. Yet significant segments of society still harbor doubt and skepticism about what scientists tell us, with some, including a previous US Administration, either minimizing the threat or denying it altogether, and others fearing the worst.¹

In the minds of many, doubt or skepticism is justified by uncertainties surrounding climate science, including the extent of terrestrial carbon uptake, mankind's role in this, the relationship between carbon and temperature, and, ultimately any claimed economic damages.² The nature and causes of resistance to accepting bad news and environmental threats have been much discussed in the literature (see Meyer and Kunreuther (2017), Kunreuther et al. (1978), Kaufmann et al. (2017)). At the simplest level, disbelief may be motivated by myopic economic self interest, such as a refusal to contemplate the possibly enormous costs of carbon abatement, and an unwillingness to accept significant taxation on fossil energy (see Lifton (2017)). Alternatively, the much publicized threats of pending climate-related disasters have likely contributed to a fear in some sectors of society that the worst of climate catastrophes will ensue. For the analyst, some humility may be in order,



Citation: von zur Muehlen, Peter. 2022. Prices and Taxes in a Ramsey Climate Policy Model under Heterogeneous Beliefs and Ambiguity. *Economies* 10: 257. https://doi.org/10.3390/ economies10100257

Academic Editor: Ralf Fendel

Received: 2 July 2022 Accepted: 8 October 2022 Published: 17 October 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the author. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). because, as Millner et al. (2010) have noted, "our knowledge of the climate system is not of sufficiently high quality to be described with unique probability distributions, and that formal frameworks that account for aversion to ambiguity are normatively legitimate". In a recent paper, Pindyck (2017) warned that excessive reliance on Integrated Assessment Models (IAMs), first introduced by Nordhaus (1993), "create a perception of knowledge and precision that is illusory, possibly fooling policy-makers into thinking that the forecasts that models generate have some kind of scientific legitimacy. The same goes for any illusion that the probability distributions underling uncertainty can even be known". It is this last thought that mostly motivates the present paper.

For purposes of exposition, I shall give specific meaning to the terms *skepticism* and doubt. Skeptics are said to be individuals who distort their probability assessments in favor of different outcomes than predicted by science. Because such beliefs are arbitrary, they are also, in principle, unknown to the government, thereby creating an ambiguity for the policy maker (see Hansen and Sargent (2012)). By doubt I shall mean the kind of pessimism regarding the trustworthiness of a given model that Hansen and Sargent (2012) have described as emanating from a belief that nature is likely to produce worst-case outcomes. Although the literature has focused solely on ambiguities inhabiting the policy authority, both, government and private agents, may be pessimistic regarding the climate model. Pessimists in this paper make decisions by playing a game against an imagined malevolent opponent. A powerful and rational motive for pessimism is the prospect of tipping points that Dietz et al. (2021) have described as the most important source of uncertainty, one capable of throwing off all modeling descriptions and justifying preparedness for the worst.³ In this paper, all pessimistic players use Hansen and Sargent's (2008) min-max strategies to compute their policies, meaning they are revealed to the authority, assumed to be a Ramsey planner, in the form of additional implementability constraints under which such a planner must operate.

The preceding sources of ambiguity give rise to four potential belief regimes:

- 1. Unknown private beliefs
 - (a) Political planner: The government strategically assumes private beliefs to be true. Its ignorance of private beliefs produces ambiguity.
 - (b) Paternalistic planner: The government trusts and adheres to the true model. However, the presence of unknown private beliefs creates ambiguity for policy.
 - (c) Pessimistic planner: Having doubts about the model and facing unknown private beliefs, the planner confronts two sources of ambiguity.
- 2. Known pessimistic private beliefs
 - (a) Pessimistic planner: The planner's own doubts doubts about the model and the private sector's pessimistic doubts that constrain the Ramsey planner's policy produce two sources of ambiguity.

The framework for studying policy with ambiguity is based on work by Hansen and Sargent (2005, 2007, 2008), a general rationale for pursuing robust climate policy under deep uncertainty having been provided by Workman et al. (2021). In their theory, deviations of private-sector beliefs from some approximating or reference distribution—the true scientific distribution—are represented as martingale multiplier distortions *M*, later defined as likelihood ratios having well defined properties. A justification for this approach to modeling ambiguity is based on a theorem by Strzalecki (2011), who axiomatized the robust control criterion of multiplier preferences introduced by Hansen and Sargent (2001), relating them to other classes of preferences studied in decision theory, in particular, the variational preferences introduced by Maccheroni et al. (2006), and proving them to be equivalent to multiplier distortions of probabilities.

As indicated earlier, the study of ambiguity in the context of climate policy is not without precedent, but discussions have generally been limited to cases when only the planner has doubts about the approximating climate-economic model. Such doubts include concerns about potential mis-specification of alternative models and ambiguity over how much weight to assign to each of these models, while agents themselves are usually assumed to have rational beliefs (see Millner et al. (2012), Brock and Durlauf (2015), Cai et al. (2013), Cai and Lontzek (2019), Anderson et al. (2013), Berger et al. (2016), Li et al. (2016), Lemoine and Traeger (2016), Rezai and van der Ploeg (2017), and Barnett et al. (2020)). Hennlock's (2009) is no exception in that, although he attributes deep uncertainty to the consumer, the government, being a direct extension of the consumer, remains the consumer's sole agent, so that, in effect, it is the planner who is modeled as having doubts about the model. In a bit of a twist to the approaches taken by other researchers, Rezai and van der Ploeg (2017) studied the implications of adopting max-min, max-max, and min-max regret policies when the planner faces alternative models ranging from science-based paradigms to denialist imaginings, concluding that max-min or min-max regret climate policies that rely on a non-skeptic view of global warming lead to a substantial and moderate amount of caution, respectively, while max-max policies produce policies that do not match the beliefs of climate skeptics. Later, Rezai and van der Ploeg (2019) applied a version of *Pascal's wager* and asked: what would an agnostic but rational planner—one who does not know or care which model is correct—do when faced with some probability that the approximating model, adhered to by so-called deniers, is false? Their conclusions are briefly described in Section 13.

The literature on optimal climate policy has generally followed the tradition of welfare analysis based on expected utility maximization within the framework of an integrated climate assessment model, with government defined as a social planner (vid. Golosov et al. 2014) seeking to maximize the expected welfare of society unconstrained by private decisions or market outcomes. An alternative is to assume that the government is a Ramsey planner, likewise seeking to maximize consumer welfare, akin to the authority introduced by Chari et al. (1994) to study optimal dynamic capital taxation, but under constraints imposed by market equilibrium.⁴ This paper studies both versions of government, in which each type of planner must acknowledge the possibly distorted beliefs held by the private sector.

In its essentials, the description of the economy here follows the recent literature on optimal carbon taxation, foremost among them Nordhaus (2008), Acemoglu et al. (2012), von Below (2012), Golosov et al.'s (2014), van der Ploeg and Withagen (2014), and Belfiori (2017, 2018). The analytical framework is a by now familiar dynamic stochastic integrated general equilibrium (DSIGE) model similar to those in Anderson et al. (2013) and Golosov et al. (2014), which in turn are based on RICE—Regional Dynamic Integrated Model of Climate and the Economy—developed by Nordhaus (1993, 2008, 2007).

The government's fiscal policy tools include bond finance and taxes on carbon and capital. I include a tax on capital because within a Ramsey planning framework, capital and Pigouvian carbon taxation are tightly linked: (1) the government's stochastic discount factor for the return to capital and for the expected social cost of carbon (SCC) is the same, and (2) the SCC and taxes on capital and carbon are influenced by the same shadow price.⁵

In essence, this paper will show how the planner's ignorance of the model or of skeptical private beliefs creates an endogenous gap between the government's and the household's discount factors, leading to a gap in their respective Arrow–Debreu pricing of carbon and capital that contributes to an *ambiguity premium* over the standard certainty-equivalent formulation of the expected social cost of carbon.⁶ This finding is related to recent papers by von Below (2012), Barrage (2018), and Belfiori (2017), who proved in different contexts that the optimal tax on capital is negative, and the optimal tax on carbon is higher than the standard Pigou rate, if the government's subjective discount rate is *exogenously* lower than the public sector's.⁷ The underlying reason is that climate change decreases the returns to capital, so that individuals, who are too impatient from a social point of view, i.e., skeptical, do not save enough without a capital subsidy and, at the same time, burn too much fossil fuel unless the latter is taxed sufficiently. The optimal policy response is therefore to tax capital less and to tax carbon more.⁸ In the reverse case, a pessimistic public motivated to over-invest and to under-utilize carbon may justify

lower carbon and higher capital taxation, unless the government is also pessimistic. In this paper, any disparity in discount factors between the government and the private sector is endogenously driven, in this instance by heterogeneity in beliefs, fear of mis-specification, and ambiguity.

The next section provides formal definitions of pessimism and skepticism as understood in this paper, using multiplier preferences introduced by Hansen and Sargent (2001, 2005, 2008). Section 3 derives the Euler conditions for a consumer who may be skeptical (Section 3.5) or pessimistic (Section 3.4), as defined in Section 2. Section 4 presents a three-factor production function subject to damaging climate-related total productivity shocks in a model of a firm renting capital and purchasing energy from the household. Section 5 uses results from Sections 3.3, 3.6, and 4 to derive two versions of Hotelling's rule. Subject to constraints derived in Section 6, Section 7 presents the Ramsey planner's Euler conditions for the three belief regimes under study. Section 8 derives the possibly distorted equilibrium prices of carbon damage and capital. In anticipation of the main results, Section 9 comments on this paper's methodology and approach, which contain some innovations. Section 10 derives the expected social cost of carbon, including an ambiguity premium that governments in all four policy/belief regimes will implement. Section 11 establishes the conditions under which the planner may or may not impose an additional ambiguity-related carbon tax premium over the social cost of carbon. Section 12 presents a number of conditions under which a planner may or may not raise the subsidy rate on capital, where it will become apparent that such conditions mirror those that drive results for the carbon tax premium. Finally, before the paper's conclusion, Section 14 describes a reverse feedback from taxation to beliefs whereby a planner facing an economy with pessimistic agents is able to manipulate debt and taxes to affect pessimistic beliefs.

Throughout, references to state-conditioned distorted (including robust) Arrow–Debreu prices reflect the basic theme in this paper: that a Ramsey planner's Pigouvian tax policy under ambiguity is able to implement allocations via equilibrium pricing of an underlying asset with unknown returns that have an equivalence in a cap and trade economy. In this respect, this paper is most closely related to Barnett et al. (2020), who use asset pricing methods not only to impute market valuations but also to ascertain social valuations, as this paper intends. As in Barnett et al. (2020), the asset prices in this paper can be viewed as equivalent to shadow prices of the expected discounted values of stochastic processes impinging on the economy.

Climate policy that is motivated by a planner's own deep or Knightian uncertainty is an interesting and important topic and has been fairly exhaustively treated in the literature. Furthermore, as has also been shown elsewhere in the literature, the effects of ambiguities originating from the planner turn out not always to be clear-cut, depending on specific features related to preferences and returns in the economy. More importantly, as maintained in this paper, an equally or possibly more urgent issue for policy must be the role of beliefs held in the private sector, because they affect consumption and investment decisions, where heterogeneity between private and government beliefs will surely impact policy. That public acceptance of climate science and its policy prescriptions are not unanimous is uncontroversial and has been well documented. However, little attention has been paid to its implications, particularly its effects on ambiguity in optimal climate policy. By altering inter-temporal rates of substitution or pricing kernels that determine consumption decisions and household wealth, belief distortions in the private sector impact a planner's implementability constraints and become more salient by the addition of ambiguities that arise when those beliefs are unknowable to the government. Whatever deep uncertainty may or may not already inhabit the mind of a planner, a welfare maximizing policy authority would be remiss in ignoring the effects of private-sector belief heterogeneity and associated ambiguities on its own policy decisions.

This paper then is the first systematic attempt to analyze the policy implications of ambiguities arising from belief heterogeneity in the private sector regarding the nature of anthropomorphic climate change. Significantly, this paper distinguishes between mere skepticism producing myopic behavior in the economy and true doubt as manifested by worst-case fears leading to increased foresight. The analytic choice to investigate the implications of belief heterogeneity and associated ambiguities by evaluating the planner's Euler conditions follows the example of Anderson et al. (2013). However, the particular approach leading to a role for second-order moments as factors in optimal policy is an innovation of this paper. The covariances between multiplier distortions and variables in the economy that arise as relevant to policy will be shown to allow more detailed descriptions of the effects of ambiguity on the social cost of carbon and on carbon and capital taxation. As will become clear later, ambiguities resulting from belief heterogeneity and distortions in the private sector produce ambiguity-related premiums on the social cost of carbon and, separately, on the carbon tax and on capital subsidies.

2. Multiplier Preferences

The representative consumer and the government share a reference probability model, given a joint density $\pi(x^t)$ of the history of shocks $x^t = x_0, \dots, x_t$, where, as detailed later in Section 4 (see Equation (21)), x_t is a climate-induced damage shock to the production economy at time *t*. The consumer and the government do not necessarily agree that π is the true probability. The government may doubt the model fearing the worst, and the private sector may either have similar doubts or be skeptical in some arbitrary way. Either may then choose an alternative model via some distortion of π in a manner described by Hansen and Sargent (2001, 2005, 2008), who invoke the Radon-Nikodym theorem to express any alternative model as as a non-negative measurable mapping $\hat{\pi}(x^t) = M_t(x^t)\pi(x^t)$, with $\mathbb{E}_t M_t = 1$, where, since uncertainty is realized in t = 0, $M_0 = 1$.⁹ Being the unconditional likelihood ratio $M_t(x^t) = \frac{\hat{\pi}_t(x^t)}{\pi_t(x^t)}$ of an alternative density to $\hat{\pi}_t(x^t)$, M_t , is a martingale with respect to the reference model π , $\mathbb{E}_t M_{t+1} = M_t$, with the interpretation of a change in measure. The *distorted expectation* of x_{t+1} , given history x^t , is ¹⁰

$$\tilde{\mathbb{E}}_t[x_{t+1}|x^t] = \mathbb{E}_t[\frac{M_{t+1}}{M_t}x_{t+1}].$$

As described in more detail below, disbelief may take two forms, either as skepticism or as pessimism, the latter being a manifestation of worst-case fears. Importantly, throughout, any alternative model $\hat{\pi}(x^t)$ is assumed to be absolutely continuous with respect to the reference model $\pi(x^t)$.¹¹

Later, it will be convenient to decompose M_t by defining the conditional likelihood ratio $m_t(x^{t+1}) \equiv \frac{M(x^{t+1})}{M(x^t)}$, such that $\mathbb{E}_t m_{t+1} = 1$.

2.1. Skepticism (Random Belief Distortion)

As indicated in the introduction, for the purpose of this paper, skepticism refers to an arbitrary rejection of the extant approximating model $\pi(x^t)$ in favor of some other model $M_t \pi(x^t)$, where M_t is a random variable with properties previously set out, including the assumption of absolute continuity with respect to the true distribution $\pi_t(x^t)$, which means that households can be skeptics but not outright climate change deniers.

2.2. Pessimism (Ambiguity Aversion)

Pessimism or model doubt refers to a worst-case belief distortion $\hat{\pi}_t(x^t) = M_t \pi_t(x^t)$ derived from the consumer's having solved a min-max problem shown later.

Following Hansen and Sargent (2008), define discounted *relative entropy* conditional on date zero information as the distance $v_0(\hat{\pi}_t, \pi_t)$ between $\hat{\pi}_t$ and π_t associated with M_t over time-*t* information and over an infinite horizon as

$$\begin{aligned} v_0(\hat{\pi}_t, \pi_t) &= (1 - \beta) \mathbb{E}_0 \sum_{t=0}^{\infty} \beta^t M_t \log M_t \\ &= \beta \mathbb{E}_0 [\sum_{t=0}^{\infty} \beta^t M_t \mathbb{E}_t \frac{M_{t+1}}{M_t} (\log M_{t+1} - \log M_t)], \\ &= \beta \mathbb{E}_0 [\sum_{t=0}^{\infty} \beta^t M_t \mathbb{E}_t m_{t+1} \log m_{t+1}], \end{aligned}$$

where β is the subjective discount factor. With this definition, an agent's ambiguity about π is represented by a set of joint densities $\{M_t\}_{t=0}^{\infty}$ satisfying the constraint,

$$\beta \mathbb{E}_0 M_t \mathbb{E}_t[m_{t+1} \log m_{t+1}] \le \eta, \tag{1}$$

where $\eta > 0$. In the following, a pessimistic consumer will choose a consumption plan subject to the constraint in (1).¹²

3. Households

3.1. CRRA Preferences

Consumers derive utility from consumption c_t , given a constant-elasticity preference function $u(c_t)$,¹³

$$u(c_t) = \frac{c_t^{1-\gamma}}{1-\gamma}, \ 0 < \gamma < 1,$$

= log $c_t; \ \gamma \to 1,$

with elasticities $\epsilon_{cc} = -u_{cc}c/u_c = \gamma$ (implying constant relative risk aversion).

C

If $\gamma = 1$ (logarithmic preferences), $\epsilon_{cc} = 1$. For the record, the constant intertemporal elasticity of substitution for consumption is $1/\gamma$. I will assume that $\gamma \leq 1$, which accords with much of the literature on long-run risk—the kind this paper is most concerned with.¹⁴

3.2. The Household's Budget Constraint

The household owns three assets: (1) the stock of depreciating capital with a net yield $[(1 - \tau_t^k(x^t))(r_t(x^t) - \delta)]k_t(x^{t-1})$, where r_t is the real rental rate on capital $k_t(x^{t-1})$ left over from last period, and τ_t^k is the tax on capital; (2) a government bond $b(x^t)$, defined as an Arrow–Debreu security promising one unit of consumption in period t + 1, if the state is x_{t+1} and zero otherwise, and (3) the resource Q_t of fossil fuels from which it draws E_t units every period, according to the law of motion¹⁵

$$Q_{t+1} = Q_t - E_t, \tag{2}$$

which, by the assumption of exhaustibility, implies

$$\sum_{t=0}^{\infty} E_t \le Q_0. \tag{3}$$

The household sells E_t to the firm at a price p_t^e with after-tax revenue $(p_t^e(x^t) - \tau_t^e(x^t))E_t(x^t)$, where τ_t^e is an excise (carbon) tax per unit of energy.

The household receives income from (1) inelastically supplied labor H_t at the competitive wage $w_t = 1$, (2) rent from capital, (3) revenues from the sale of fossil energy, and (4) a lump-sum transfer from the government g_t . It spends its resources on consumption c_t , new capital k_{t+1} , and the purchase of a new Arrow security b_{t+1} , trading at the state-contingent price $\hat{p}_{t+1}(x_{t+1}|x^t)$ to be defined later. Summarizing, the household's one-period budget constraint is

$$0 \leq H_{t}(x^{t}) + b_{t}(x^{t}) + g_{t}(x^{t}) + R_{t}^{k}(x^{t})k_{t}(x^{t-1}) + (p_{t}^{e} - \tau_{t}^{e})[Q_{t}(x^{t}) - Q_{t+1}(x^{t})] - c_{t}(x^{t}) - k_{t+1}(x^{t}) - \mathbb{E}_{t}\hat{p}_{t+1}b_{t+1}(x^{t+1}) \equiv \mathcal{L}_{t}(x^{t}),$$
(4)

where $R_t^k(x^t) = 1 + (1 - \tau_t^k)(r_t(x^t) - \delta)$ is the after-tax gross return to capital.

3.3. The Consumer's Maximization Problem

This section studies the two types of consumer introduced earlier: those who are skeptical of the model and form some arbitrary belief distortion M to the true distribution π according to Section 2.1, and those who are pessimistic and play a game against a malevolent force to determine a worst-case value for M following Section 2.2. It is convenient to set this problem up for the latter and then show the former to be a special case.

3.4. Pessimistic Consumer

The general framework for the consumer's problem is a game against some malevolent force representing extreme uncertainty about the model. Given the resource constraint in (2) and the budget constraint (4), the representative consumer solves the Lagrangian

$$\max_{\{c_t, E_t, b_{t+1}, k_{t+1}, Q_{t+1}\}} \min_{\{M_t, m_{t+1}\}} \qquad \mathbb{E}_0 \sum_{t=0}^{\infty} \beta^t M_t \left(u(c_t) + \lambda_t^c \mathcal{L}_t - \frac{\beta}{\sigma^c} m_{t+1} \log m_{t+1} \right),$$

subject to $M_{t+1} = m_{t+1}M_t$, $M_0 = 1$, $\mathbb{E}_t m_{t+1} = 1$, where λ_t^c is a time-varying Lagrangian multiplier associated with the household budget constraint (4), and $0 \ge -\frac{1}{\sigma^c} < \infty$ denotes a Lagrangian shadow cost for the penalty of deviating from rational expectations represented by the approximating distribution π , also known as the Kullback-Leibler distance (see Kullback and Leibler 1951) between the two probability measures π and $\hat{\pi}$; so $0 > \sigma^c > -\infty$ may be considered the consumer's *parameter of ambiguity aversion*.¹⁶ As in all of this paper, expectations \mathbb{E}_t are taken over the measure π .

The preceding criterion has the Bellman recursion

$$\mathcal{U}_{t}(k_{t}, b_{t}, Q_{t}) = \max_{c_{t}, E_{t}, k_{t+1}, b_{t+1}, Q_{t+1}} \min_{m_{t+1}} u(c_{t}) + \lambda_{t}^{c} \mathcal{L}_{t}(x^{t}) + \beta \mathbb{E}_{t} \left(m_{t+1} \mathcal{U}_{t+1}(k_{t+1}, b_{t+1}, Q_{t+1}) - \frac{\beta}{\sigma^{c}} m_{t+1} \log m_{t+1} \right) + \varphi_{t}^{c} (Q_{t} - E_{t} - Q_{t+1}),$$
(5)

where φ_t^c is the Lagrangian shadow price associated with the resource constrains (2).

3.4.1. Inner Minimization

The optimal conditional likelihood ratio that minimizes (5) has the familiar exponentially twisting form¹⁷

$$m_{t+1}^c = \frac{e^{\sigma^c \mathcal{U}_{t+1}}}{\mathbb{E}_t e^{\sigma^c \mathcal{U}_{t+1}}}$$

or, equivalently,

$$M_{t+1}^c = \frac{e^{\sigma^c \mathcal{U}_{t+1}}}{\mathbb{E}_t e^{\sigma^c \mathcal{U}_{t+1}}} M_t^c.$$
(6)

The worst-case martingale distortion in (6) is *pessimistic* in that it attaches higher probabilities to histories with low continuation utilities and lower probabilities to histories with high continuation utilities. Notably, because of its dependence on continuation values, it is also a function of future distortions m_{t+j}^c and future decisions by both the consumer and the government that, as shown later, determine equilibrium prices in the economy.

Substituting m_{t+1}^c from (6) into (5) produces the *risk-sensitive* recursion¹⁸

$$\mathcal{U}(k_t, b_t, Q_t) = \max_{c_t, E_t, k_{t+1}, b_{t+1}, Q_{t+1}} u(c_t) + \frac{\beta}{\sigma^c} \mathbb{E}_t \log e^{\sigma^c \mathcal{U}(k_{t+1}, b_{t+1}, Q_{t+1})} + \lambda_t^c \mathcal{L}_t.$$
(7)

The first-order conditions (FONCs) for the flow variables $\{c_t, E_t\}$ are

$$\lambda_t^c = u_{c_t}, \tag{8}$$

$$(p_t^e - \tau_t^e)\lambda_t^c = \varphi_t^c.$$
(9)

The envelope conditions are,

$$\mathcal{U}_{k_t} = \lambda_t^c R_t^k = u_{c_t} R_t^k, \qquad (10)$$
$$\mathcal{U}_{b_t} = \lambda_t^c = u_{c_t}, \qquad (11)$$

$$\mathcal{U}_{h_t} = \lambda_t^c = u_{c_t}, \tag{11}$$

$$\ell_{O_t} = \varphi_t^c. \tag{12}$$

In conjunction with these envelope conditions, the first-order conditions for the stock variables b_{t+1} , k_{t+1} , and Q_{t+1} are, respectively,

$$\hat{p}_{t+1} = \beta m_{t+1}^c \frac{u_{c_{t+1}}}{u_{c_t}}, \tag{13}$$

$$1 = \beta \mathbb{E}_{t} m_{t+1}^{c} \frac{u_{c_{t+1}}}{u_{c_{t}}} R_{t+1}^{k} = \mathbb{E}_{t} \hat{p}_{t+1} R_{t+1}^{k}, \qquad (14)$$

$$\varphi_t^c = \beta \mathbb{E}_t m_{t+1}^c \varphi_{t+1}^c. \tag{15}$$

Note that \hat{p}_{t+1} is the one-period worst-case equilibrium price of a state-contingent claim as shown later in (18). According to the preceding conditions, the price of such a claim is determined by continuation utilities and the random climate cost shock x_{t+1} . This is information that the government can exploit to determine optimal fiscal policy.

3.5. Skeptical Consumer

Let M^s denote a skeptical consumer's belief distortion. As stated earlier, skepticism represents an arbitrary random distortion $\hat{\pi}_t \equiv M_t^s \pi_t$ of the true distribution π_t . From the point of view of society (or the government), M_{t+1}^s is an unknowable exogenous variable. However, this household is sure of its beliefs and evaluates (5) by disabling the penalty on belief distortions and letting $\sigma^c \to 0$.

The Euler equations are the same as before, except that $m = m^s$ is random and unknown to the authorities, and the equilibrium price \hat{p} of a state-contingent claim based on skepticism as defined here is unrelated to the consumer's continuation utility. Additionally, since m_{t+1}^s is random, it is unrelated to x_{t+1} .

3.6. Arrow–Debreu Prices under Belief Distortions

The belief distortions of probabilities, ranging from skepticism to deep uncertainty, treated in this paper, add an important dimension to stochastic discounting. This section shows how to construct their corresponding asset-price measures.

For preferences $u(c_t)$ distorted by a martingale process M_{t+i} (either M_{t+i}^s or M_{t+i}^c), the *j*-period-ahead stochastic discount factor (MSDF) is,

$$\hat{\varrho}_{t+j,t} = \beta^{j} \frac{M_{t+j}}{M_{t}} \frac{u_{c_{t+j}}}{u_{c_{t}}} \equiv \frac{M_{t+j}}{M_{t}} \varrho_{t+j,t},$$
(16)

where $\varrho_{t+i,t}$ is the pricing kernel under rational expectations. When j = 1, this becomes the familiar one-period-ahead stochastic discount factor SDF.

Let $\hat{q}_{t+j}(x^{t+j})$ be the *j*-period numeraire

$$\hat{q}_{t+j}(x^{t+j}) \equiv \beta^{j} M_{t+j} \pi_{t+j}(x^{t+j}) \frac{u_{c_{t+j}}(x^{t+j})}{u_{c_{0}}(x^{0})}, \quad \hat{q}_{0} = 1, M_{0} = 1,$$
(17)

and define

$$\hat{p}_{t+j,t}(x_{t+j}|x^t) \equiv \frac{\hat{q}_{t+j}(x^{t+j})}{\hat{q}_t(x^t)} = \beta^j \left(\prod_{i=1}^j m_{t+i} \frac{u_c(x^{t+i})}{u_c(x^t)}\right) \pi_{t+j}(x^{t+j})$$

$$= \frac{M_{t+j}}{M_t} \varrho_{t+j,t} \pi_{t+j}(x^{t+j})$$

as the market's distorted t + j equilibrium price in (13) of an Arrow–Debreu security in terms of consumption at history x^t , or equivalently,

$$\hat{p}_{t+j,t}(x_{t+j}|x^t) = \hat{q}_{t+j,t}\pi_{t+j}(x^t),$$
(18)

which, for j = 1, also corresponds to the first-order condition for capital in (14). For future reference, denote the undistorted rational expectations price by

$$p_{t+j,t}(x_{t+j}|x^t) = \varrho_{t+j,t}\pi_{t+j}(x^t).$$
(19)

4. Firms

The economy's output is produced by a continuum of atomistic firms with a Cobb-Douglas production technology:

$$Y_t = (1 - D_t(T_t - T_0))F(k_t, H_t, E_t, Q_t) = (1 - D_t(Q_t))k_t^{\alpha} E_t^{\nu} H_t^{1 - \alpha - \nu},$$
(20)

where k_t is the stock of capital, H_t is hours of labor, E_t is the flow of fossil energy, and Q_t is the remaining stock of carbon energy in the ground. As explained presently, $D_t(Q_t)$ is a damage function measuring the proportion of GDP lost due to the change in average global temperatures $T_t - T_0$ since the beginning of the Industrial Revolution at t = 0.

Dietz and Venmans (2019) observe that "climate has delivered two important and related insights. First, global warming appears to be approximately linearly proportional to cumulative emissions of carbon dioxide. Second, the temperature response to an emission of CO_2 appears to be approximately instantaneous and then constant as a function of time". This conclusion accords with Matthews et al. (2012), and Collins et al. (2013), who earlier defined the *Transient Climate Response to Cumulative Carbon Emissions* (TCRE)¹⁹

$$\lambda_t^{TC} = \frac{T_t - T_0}{Q_0 - Q_t}$$

where $0 \le Q_0 - Q_t = \sum_{i=0}^{t-1} E_i$ is accumulated carbon emissions since the beginning of the Industrial Revolution.²⁰ The TCRE parameter is generally assumed to be a stochastic variable, due to uncertainties surrounding climate modeling.

The preceding motivates a damage function having the following exponential form

$$D_t(Q_t) = e^{\zeta_t(T_t - T_0)} = e^{\zeta_t \lambda_t^{TC}(Q_0 - Q_t)} \equiv e^{x_t(Q_0 - Q_t)}, \ \frac{\partial D_t}{\partial Q_t} = -x_t e^{x_t(Q_0 - Q_t)} < 0,$$
(21)

where ζ_t is a stochastic parameter that translates the damaging effects of temperature changes into units of GDP, and $x_t = \zeta_t \lambda_t^{TC}$ combines the damage parameter with the TCRE parameter λ_t^{TC} .²¹ As posited in Section 2, I shall consider x_t to be a random variable with either known distribution $\pi_t(x_t)$ or unknown distribution giving rise to ambiguities described earlier. Confining the source of economic damage to a single catch-all variable follows the practice of a number of authors, including Li et al. (2016), who create a single

source of model uncertainty with a stochastic variable γ that reduces their end-of-period capital stock. Others who have used similar exponential formulations include Golosov et al.'s (2014), and Anderson et al. (2013).²² To the extent that there exist possibilities of catastrophic tipping points that would likely upend all calculations and planning, the underlying distribution of x_t may be taken as Knightian and unknowable by either the private sector or the government or both. Tipping points, analyzed by Lemoine and Traeger (2016), and Cai et al. (2013), are abrupt nonlinear climate disruptions that pose a potentially existential threat to humanity in ways that may override concerns with belief and skepticism. A further source of extreme uncertainty is *polar amplification* analyzed by Brock and Xepapadeas (2017).

The typical firm rents capital k_t from consumers and buys energy E_t from households in order to maximize an expected infinite stream of profits. Hours of labor H_t are supplied inelastically at the going wage 1—a standard assumption in this literature (see Golosov et al. (2014) and Li et al. (2016)).

$$\begin{aligned} \max_{k_t, H_t, E_t} \mathbb{E}_o \sum_{t=0}^{\infty} \hat{q}(x_t) [Y_t - r_t k_t - H_t - p_t^e E_t], \\ = \max_{k_t, H_t, E_t} \mathbb{E}_o \sum_{t=0}^{\infty} \hat{q}_t(x_t) \Big((1 - e^{-x_t(Q_t - Q_0)}) k_t^{\alpha} E_t^{\nu} H_t^{1 - \alpha - \nu} - r_t k_t - H_t - p_t^e E_t \Big), \end{aligned}$$

where $\hat{q}(x_t|x_0)$ is the belief-distorted and possibly robust numeraire defined earlier in (17). The first-order conditions with respect to $\{k_t, H_t, E_t\}$ are

$$Y_t = Y_{k_t} = \alpha \frac{Y_t}{k_t}, \tag{22}$$

$$1 = Y_{H_t} = (1 - \alpha - \nu) \frac{Y_t}{H_t},$$
 (23)

$$p_t^e = Y_{E_t} = \nu \frac{Y_t}{E_t}.$$
(24)

5. Two Versions of Hotelling's Rule

Denote by R_{t+1}^e the rate of return to energy, net of the carbon tax τ_t^c ,

r

$$R_{t+1}^{e} = \frac{p_{t+1}^{e} - \tau_{t+1}^{e}}{p_{t}^{e} - \tau_{t}^{e}}.$$
(25)

Additionally, since, for either $m = m^s$ or $m = m^c$, (9) and (15) imply

$$p_t^e - \tau_t^e = \beta \mathbb{E}_t m_{t+1} \frac{u_{c_{t+1}}}{u_{c_t}} (p_{t+1}^e - \tau_{t+1}^e),$$
(26)

it follows that

$$\mathbb{E}_t \hat{p}_{t+1} R^e_{t+1} = 1, \tag{27}$$

where expectations are taken with respect to $\pi(x^t)$, as before. Substituting (24) for period t and t + 1 in (26) and using the result $p^e = Y_E$ from (24), yields the firm's dynamic rule for optimal energy use, given taxes,

$$Y_{E_t} - \tau_t^e = \mathbb{E}_t \hat{\varrho}_{t+1} (Y_{E_{t+1}} - \tau_{t+1}^e), \tag{28}$$

which is a version of Hotelling's rule that will become relevant later for determining the social cost of carbon.

If $m_{t+1} = 1$, the preceding equation confirms Hotelling's original formula that with zero or constant taxes, energy consumption falls over time at the subjective rate of discount. If consumers are climate-skeptic, a multiplier $m_{t+1}^s < 1$ effectively lowers the private-sector's discount factor for future benefits of fossil fuel in favor of current returns,

leading to increased current consumption relative to rational expectations. If the consumer is pessimistic with $m_{t+1} = m_{t+1}^c$, energy expenditures are delayed, following the same argument.

Hotelling's (1931) original rule states that the price of an exhaustible resource net of extraction costs should rise at the rate of interest, which, on average, is above the rate of real GDP growth. By Equation (14), the *t*-period Arrow–Debreu price of a claim on a unit of capital is

$$P_t^k = \mathbb{E}_t \sum_{j=0}^{\infty} \hat{p}_{t+j,t} R_{t+j}^k, \qquad (29)$$

while Equation (27) implies that the *t*-period Arrow–Debreu price of a claim on a unit of carbon energy is

$$P_t^e = \mathbb{E}_t \sum_{j=0}^{\infty} \hat{p}_{t+j,t} R_{t+j}^e.$$
 (30)

From Equations (14) and (27) follows a version of Hotelling's rule in terms of returns:

Version 1 :
$$R_{t+1}^e = R_{t+1}^k, \forall t,$$
 (31)

while Equations (29) and (30) restate Hotelling's rule in terms of Arrow–Debreu contingent prices:

$$Version \ 2: \qquad P_{t+1}^e = P_{t+1}^k, \ \forall t.$$
(32)

Note that in each case, returns and prices are distorted by the consumer's beliefs, be they skeptical m^s or pessimistic m^c .

These two versions of Hotelling's rule (31) and (32) constitute binding no-arbitrage conditions that require prices and returns to capital and fossil fuel to be equal: in equilibrium, the return to fossil fuel left in the ground for one more period equals the return to the next unit of capital. This rule illuminates two important features of a competitive market for exhaustible energy when there is uncertainty and skepticism regarding the underlying model: (i) the pricing of energy resources continues to obey the laws of asset markets requiring equality of returns to all activities, including capital, bonds, and energy stores, but (ii) the market now uses a martingale-distorted and possibly robust stochastic discount factor to evaluate the expected future returns to all assets.

Remark 1. Hotelling's rule assures efficient allocation but not necessarily socially optimal outcomes if it fails to internalize costs to society created by private economic activity. Later, Section 11 shows how the government can remedy this failure with a public version of Hotelling's rule that includes a social accounting of all costs.

The next few sections discuss how a Ramsey planner implements competitive equilibrium depending on assumptions about heterogeneity in beliefs, the degree of doubt about the model by either the government or the public, and by how ignorant the planner is about private beliefs.

6. The Ramsey Planner's Constraints

6.1. National Income Identity and the Government Budget Constraint

In all belief regimes considered here, a Ramsey planner commits to policy in period 0 by choosing a competitive equilibrium that maximizes the consumer's expected utility over time. This means the government chooses allocations that satisfy the natural resource constraint (2) and (3) and the national income resource constraint that output exhausts

consumption plus investment plus government spending, which, to keep things simple, is assumed to be entirely devoted to a lump-sum transfer g_t ,

$$Y_t = c_t + k_{t+1} - (1 - \delta)k_t + g_t.$$
(33)

The government's budget constraint requires that spending and the redemption of bonds from the preceding period be covered by tax receipts, lump-sum transfers, and new issuance of bonds:

$$b_t = \mathbb{E}_t \hat{p}_{t+1} b_{t+1} - g_t + \tau_t^e E_t + \tau_t^k (r_t - \delta) k_t.$$

When $b_t < 0$, the government is a lender. Solved forward, the preceding equation becomes the government's dynamic budget constraint,

$$b_t \le \mathbb{E}_t \sum_{j=0}^{\infty} \hat{q}_{t+j} [\tau_{t+j}^k (r_{t+j} - \delta) k_{t+j} - \tau_{t+j}^e (Q_{t+j+1} - Q_{t+j}) - g_{t+j}].$$
(34)

Note the added term involving receipts from the carbon tax.

6.2. The Ramsey Planner's Implementability Constraints

In solving for optimal taxation in the case of a Ramsey planner, I use a so-called primal approach due to Chamley (1986) that searches directly for allocations by solving the government's problem subject to an *implementability* constraint. A starting point is the household's dynamic budget constraint (4), which, when solved forward for b_t , utilizing the no-arbitrage condition (14) and a no-Ponzi game condition, yields the intertemporal budget constraint²³

$$b_t \geq \mathbb{E}_t \lim_{T \to \infty} \sum_{j=0}^{T-1} \hat{p}_{t+j} [c_{t+j} - H_{t+j} - g_{t+j}].$$
 (35)

Let W_t denote household wealth in period t, composed of government bonds, the after-tax equity value of physical capital, and fossil fuels still in the ground, valued at current after-tax energy prices:

$$\mathcal{W}_t \equiv b_t + R_t^k k_t + (p_t^e - \tau_t^e) Q_t. \tag{36}$$

Appendix **B** shows that

$$\mathcal{W}_t \geq c_t - H_t - g_t + \beta \mathbb{E}_t m_{t+1} \frac{u_{c_{t+1}}}{u_{c_t}} \mathcal{W}_{t+1}$$
$$= c_t - H_t - g_t + \mathbb{E}_t \hat{p}_{t+1} \mathcal{W}_{t+1}, \qquad (37)$$

where $\hat{p}_{t+j,t}$ is the market's previously defined distorted t + j equilibrium price of an Arrow–Debreu security in terms of consumption at history x^t . Solved forward, household wealth is

$$\mathcal{W}_t \geq \mathbb{E}_t \sum_{j=0}^{\infty} \hat{p}_{t+j,t} [c_{t+j} - H_{t+j} - g_{t+j}].$$
 (38)

For later use, it is convenient to define the marginal-utility-of-consumption-scaled market value of wealth $Y_t = u_{c_t} W_t$, so that the Ramsey planner's implementability constraint (37) becomes, equivalently,

$$Y_t \geq \Omega_t + \mathbb{E}_t m_{t+1} Y_{t+1}, \tag{39}$$

where

$$\Omega_t = u_{c_t}[c_t - H_t - g_t], \qquad (40)$$

with derivative,

$$\Omega_{c_t} = \frac{u_{cc_t}}{u_{c_t}} (c_t - H_t - g_t) u_{c_t} + u_{c_t}$$

= $[1 - \gamma c_t^{-1} (c_t - H_t - g_t)] u_{c_t} = [1 - \gamma + \gamma \frac{H_t + g_t}{c_t}] u_{c_t} \ge 0,$ (41)

since, from Section 3.1, $\frac{u_{cc_t}}{u_{c_t}} = -\gamma/c_t$. Note that if the constraint (39) is nonbinding, the government is the social planner widely treated in the literature.

Solved forward,

$$Y_{t} = \mathbb{E}_{t} u_{c,t} \sum_{j=0}^{\infty} \hat{p}_{t+j} (c_{t+j} - H_{t+j} - g_{t+j}) = u_{ct} b_{t},$$
(42)

is the government's surplus valued in terms of the marginal utility of consumption.

Finally, by its very definition, a Ramsey planner heeds all equilibrium constraints imposed by competitive markets. In particular, when consumers are pessimistic with belief distortions defined in terms of their continuation values, where $m_t = m_{t+1}^c = \frac{e^{e^{c}U_{t+1}}}{\mathbb{E}_t e^{e^{c}U_{t+1}}}$, the planner faces two additional implementability constraints that come from (6) and (7):

$$M_{t+1}^{c} = \frac{e^{\sigma^{c} \mathcal{U}_{t+1}}}{\mathbb{E}_{t} e^{\sigma^{c} \mathcal{U}_{t+1}}} M_{t}^{c},$$
(43)

$$\mathcal{U}_t = u(c_t) + \frac{\beta}{\sigma^c} \log \mathbb{E}_t e^{\sigma^c \mathcal{U}_{t+1}}.$$
(44)

7. Ramsey Planning in Four Belief Regimes

Subject to the fossil resource constraint (2) and (3), the national income identity (33), and the implementability constraint (39), the taxing Ramsey authority chooses {c, E, k, Q, Y} to maximize society's expected welfare, and {N, n} to minimize discounted *relative entropy* defined in (1),

$$\max_{c,E,k,Q,Y} \min_{N,n,} \mathbb{E}_0 \sum_{t=0}^{\infty} \beta^t N_t(x^t) \left[u(c_t) - \frac{\beta}{\sigma} (\mathbb{E}_t n_{t+1} \log n_{t+1}) \right], \tag{45}$$

subject to $N_{t+1} = n_{t+1}N_t$, $N_0 = 1$, $\mathbb{E}_t n_{t+1} = 1$, and $0 > \sigma > -\infty$, where N_t is the government's martingale multiplier, equivalent to M_t defined earlier for the consumer, and $\sigma < 0$ is the planner's *parameter of ambiguity aversion*.

Exploiting a linear homogeneity property such that $V(Y_t, k_t, Q_t, N_t) = N_t \mathcal{V}_t(Y_t, k_t, Q_t)$, the problem may be cast as the recursive (Bellman) Lagrangian

$$\mathcal{V}_{t}(\mathbf{Y}_{t}, k_{t}, Q_{t}) = \max_{c_{t}, E_{t}, k_{t+1}, Q_{t+1} \mathbf{Y}_{t+1}} \min_{n_{t+1}} u(c_{t}) - \frac{\beta}{\sigma} \mathbb{E}_{t} n_{t+1} \log n_{t+1} \\
+ \beta \mathbb{E}_{t} n_{t+1} \mathcal{V}_{t+1}(\mathbf{Y}_{t+1}, k_{t+1}, Q_{t+1}) \\
+ \Phi_{t} [\Omega_{t} + \mathbb{E}_{t} m_{t+1} \mathbf{Y}_{t+1} - \mathbf{Y}_{t}] + \mathcal{M}_{t},$$
(46)

where

$$\mathcal{M}_{t} = \begin{pmatrix} \lambda_{t}[(1 - e^{x_{t}(Q_{0} - Q_{t})})F(k_{t}, H_{t}, E_{t}) + (1 - \delta)k_{t} - c_{t} - g_{t} - k_{t+1}] \\ \varphi_{t}[Q_{t} - E_{t} - Q_{t+1}] \\ v[Q_{0} - \sum_{i=0}^{\infty} E_{i}] \end{pmatrix},$$

is a group of Lagrangian constraints and λ_t , Φ_t , φ_t are non-negative Lagrangian co-state variables, and v is constant.²⁴

The preceding problem constitutes a Stackelberg game between the government as the leader and the private sector as the follower. Embedded in this game is another game with a minimizing opponent to represent worst-case outcomes. This latter game, framed as *inner minimization*, may be played by either government or consumers, as in Section 3.4, or even both, depending on respective attitudes toward extreme risk. The solution to this subgame implies a plan for *outer maximization* of an indirect value function, typically a *risk sensitivity recursion*, such as the one introduced in Section 3.4.2.

7.1. Political Planner

In this example, consumers are assumed to have arbitrarily distorted—skeptical beliefs about the approximating scientific model π , as described in Section 2.1. Importantly, the government is ignorant of these beliefs. The reason for its ignorance is fundamental: as noted by Hansen and Sargent (2012), the government's ambiguity arises because the possible space of models that are unknown to the fiscal planner but known to the consumer is so vast that it is impossible to infer the private sector's probability model from finite data, providing a motive to construct robust climate and fiscal policies by solving a socalled multiplier problem that protects against worst-case belief distortions. As Hansen and Sargent (2012) put it, the government's ignorance of private beliefs is akin to a set or cloud of probability distributions over events *x* centered on the reference or approximating density π constrained by a discounted relative entropy set of probability distributions reflecting the unknown beliefs of the private sector.²⁵ Its ignorance notwithstanding, this political planner acts under the assumption that the unknown beliefs are true.

With this in mind, the recursion (46) becomes

$$\mathcal{V}_{t}(Y_{t}, k_{t}, Q_{t}) = \max_{c_{t}, \mathcal{E}_{t}, k_{t+1}, Q_{t+1}} \min_{n_{t+1}} u(c_{t}) - \frac{\rho}{\sigma} \mathbb{E}_{t} n_{t+1} \log n_{t+1} \\
+ \beta \mathbb{E}_{t} n_{t+1} \mathcal{V}_{t+1}(Y_{t+1}, k_{t+1}, Q_{t+1}) \\
+ \Phi_{t} [\Omega_{t} + \mathbb{E}_{t} n_{t+1} Y_{t+1} - Y_{t}] + \mathcal{M}_{t}.$$
(47)

ß

The political government's acceptance of private-sector beliefs M_t as true means that m_{t+1} , which multiplies Y_{t+1} in the implementability constraint, is set equal to n_{t+1} —the authority adopts the consumer's distortion as its own.²⁶ Ambiguity is activated with the entropy constraint penalizing deviations of distorted beliefs from true beliefs (1), where $-\frac{1}{\sigma}$ is a Lagrangian multiplier, and $\sigma < 0$ measures the planner's ambiguity aversion. As σ moves toward $-\infty$, the government's preference for robustness rises. As σ approaches a break-down $\underline{\sigma} < 0$ from above, the government's concern about distortions to expectations is maximal. Conversely, as σ approaches zero, the government's preference for robustness diminishes, until, in the limit the government fully adopts the approximating reference model as true and not subject to doubt.

7.1.1. Inner Game with Nature

1

Ambiguity for this planner produces an *ex post* worst-case probability model with distorted *ex post homogeneity* in beliefs between the government and the private sector.

Minimization of (47) with respect to n_{t+1} leads to

$$n_{t+1} = \frac{e^{\sigma[\mathcal{V}_{t+1} + \Phi_t Y_{t+1}]}}{\mathbb{E}_t e^{\sigma[\mathcal{V}_{t+1} + \Phi_t Y_{t+1}]}} \equiv n_{t+1}^{PO},$$
(48)

with limits $\lim_{\sigma \downarrow -\infty} n_{t+1}^{PO} \to 0$, and $\lim_{\sigma \uparrow 0} n_{t+1}^{PO} \to 1$, indicating that the conditional likelihood ratio n^{PO} is inversely related to the intensity of the planner's doubts about private beliefs, approaching 1 as doubt ceases.

7.1.2. Outer Maximization with Implied Risk-Sensitive Recursion

The continuation value V_{t+1} in the exponent of the formula for n^{PO} comes from the planner's distrust of the reference model itself, while continuation wealth Y_{t+1} points to the planner's ignorance of private-sector beliefs. Formula (48), indicates that, given $\sigma < 0$, ambiguity leads this robust planner to put more probability weight on histories with low continuation values V_{t+1} and Y_{t+1} , and lower probabilities on histories with high continuation values.

Substituting (48) for both n_{t+1} and m_{t+1} in (47) produces a variation on Hansen and Sargent's (1995) discounted risk-sensitive recursion, one that is augmented with household wealth Y, a forward-looking entity valued at the shadow price Φ , in the exponent,²⁷

$$\mathcal{V}(\mathbf{Y}_{t}, k_{t}, Q_{t}) = \max_{c_{t}, E_{t}, k_{t+1}, Q_{t+1}, Y_{t+1}} u(c_{t}) + \Phi_{t}[\Omega_{t} - \mathbf{Y}_{t}] + \mathcal{M}_{t} + \frac{\beta}{\sigma} \log \mathbb{E}_{t} e^{\sigma(\mathcal{V}_{t+1} + \Phi_{t}Y_{t+1})}.$$
(49)

The distortion n^{PO} that attains the minimum of the right side of (49) tilts the x_{t+1} distribution exponentially toward lower continuation values via multiplication of $\pi(x_{t+1})$ by n_{t+1}^{PO} in (48).

The first-order conditions (FONCs) for $\{c_t, E_t\}$ for an interior maximum are: $\forall t \ge 1$, Flows

$$c_t: \quad \lambda_t = u_c + \Omega_{c_t} \Phi_t, \tag{50}$$

$$E_t: \quad Y_{E_t}\lambda_t = \varphi_t + v, \tag{51}$$

where $Y_{E_t} = v \frac{Y_t}{E_t}$, is the marginal product of fossil energy based on the Cobb-Douglas assumption.

Envelope conditions

$$\mathcal{L}_t: \quad \mathcal{V}_{k_t} = (1 - \delta + Y_{k_t})\lambda_t, \tag{52}$$

$$k_t: \quad \mathcal{V}_{k_t} = (1 - \delta + Y_{k_t})\lambda_t, \tag{52}$$
$$Q_t: \quad \mathcal{V}_{Q_t} = \varphi_t + \lambda_t x_t D_t F_t, \tag{53}$$
$$Y_t: \quad \mathcal{V}_{Y_t} = -\Phi_t. \tag{54}$$

$$\mathcal{U}_t: \quad \mathcal{V}_{\mathbf{Y}_t} = -\Phi_t. \tag{54}$$

Note that $\lambda \ge 0$ and $\varphi \ge 0$ imply $V_k \ge 0$ and $V_{Q_t} \ge 0$, respectively. Further, $\Phi \ge 0$ implies $V_{\rm Y} \leq 0$.

Stocks

$$k_{t+1}: = \frac{\beta}{\lambda_t} \mathbb{E}_t n_{t+1}^{PO} \mathcal{V}_{k_{t+1}} = \beta \mathbb{E}_t n_{t+1}^{PO} \frac{\lambda_{t+1}}{\lambda_t} (1 - \delta + Y_{k_{t+1}}),$$
(55)

$$Q_{t+1}: \varphi_t = \beta \mathbb{E}_t n_{t+1}^{PO} \mathcal{V}_{Q_{t+1}} = \beta \mathbb{E}_t n_{t+1}^{PO} (\varphi_{t+1} + \lambda_{t+1} x_{t+1} D_{t+1} F_{t+1}),$$
(56)
 $Y_{t+1}:$

$$\Phi_t = -\mathcal{V}_{Y_{t+1}} = \Phi_{t+1} = \bar{\Phi}, \tag{57}$$

where the second equality in each equation follows from the envelope conditions (52)-(54), and (56), giving a marginal utility valuation of the benefit of fossil energy net of climate damages. Given (57), the Lagrangian multiplier Φ_t is a constant and, by the Kuhn-Tucker condition, is zero if the wealth constraint is nonbinding, indicating that the planner is a social and not a Ramsey planner.

Note that because both the private sector's and the planner's probabilities are twisted by a martingale multiplier, the ambiguity for this planner effectively delivers *ex post* a model of endogenously distorted homogeneous beliefs. Hansen and Sargent (2012) emphasize that n^{PO} is not intended to "solve" an impossible inference problem, and being the planner's *cautious* inference about unknown private beliefs, should be viewed as merely a device to construct a robust Ramsey policy. If the planner were to solve the private sector's Euler equations using the minimizing n^{PO} in order to derive its *ex post* decision rules for consumption, labor, and energy, it would not necessarily end up reproducing their observed values, meaning that private beliefs cannot be reverse-engineered from such observations.

7.1.3. Paternalistic Planner

Woodford (2010) originally introduced a monetary authority facing a type of ambiguity described here: while trusting the reference model π , the government is ignorant about the private sector's distorted beliefs m^s . It expresses its ambiguity by setting $\sigma < 0$. Trust in its own model means that the planner sets n_{t+1} multiplying continuation value \mathcal{V}_{t+1} equal to unity.²⁸

7.1.4. Inner Game with Nature

The minimizing value of n_{t+1} is

$$n_{t+1} = \frac{e^{\sigma[\Phi_t \mathbf{Y}_{t+1}]}}{\mathbb{E}_t e^{\sigma[\Phi_t \mathbf{Y}_{t+1}]}} \equiv n_{t+1}^{PA}, \tag{58}$$

with limits $\lim_{\sigma \downarrow -\infty} n_{t+1}^{PA} \to 0$, and $\lim_{\sigma \uparrow 0} n_{t+1}^{PA} \to 1$, indicating that the conditional likelihood ratio n^{PA} is inversely related to the intensity of the planner's doubt about private beliefs, approaching 1 as doubt ceases.

An important distinction between the political planner and the paternalistic planner with ambiguity is that here, the planner's worst-case distortion of beliefs n^{PA} is solely determined by continuation values of wealth, since the only sources of ambiguity are private-sector beliefs that distort the consumer's expectation of future household wealth. Formula (58) instructs us that, via n^{PA} , a robust paternalistic planner assigns greater probability weights to histories with low continuation values of wealth, weighted with marginal utility of consumption in (39).

7.1.5. Outer Maximization with Implied Risk-Sensitive Recursion

Substitution of the formula for n_{t+1}^{PA} in (46) implies another variation on Hansen and Sargent's (1995) discounted risk-sensitive recursion,

$$\mathcal{V}(\mathbf{Y}_{t}, k_{t}, Q_{t}) = \max_{c_{t}, E_{t}, k_{t+1}, Q_{t+1}, Y_{t+1}} \{ u(c_{t}) + \Phi_{t}[\Omega_{t} - \mathbf{Y}_{t}] + \frac{\beta}{\sigma} \mathbb{E}_{t} \log e^{\sigma \Phi_{t} \mathbf{Y}_{t+1}} + \mathcal{M}_{t} \\ + \beta \mathbb{E}_{t} \mathcal{V}_{t+1}(\mathbf{Y}_{t+1}, k_{t+1}, Q_{t+1}) \},$$

which differs from the recursion (49) for a political planner in that the exponent does not include the continuation value V_{t+1} , since, here, risk sensitivity does not apply to the planner's own trusted model.

The FONCs for $\{c, E\}$ are previously given by (50) and (51). The envelope conditions are also the same as before. However, for choosing $\{k_{t+1}, Q_{t+1}, Y_{t+1}\}$, the government must solve

$$1 = \mathbb{E}_t \beta \frac{\lambda_{t+1}}{\lambda_t} (1 - \delta + Y_{k_{t+1}}), \tag{59}$$

$$\varphi_t = \beta \mathbb{E}_t \mathcal{V}_{Q_{t+1}} = \beta \mathbb{E}_t (\varphi_{t+1} + \lambda_{t+1} x_{t+1} D_{t+1} F_{t+1}), \tag{60}$$

$$\Phi_t = -\mathcal{V}_{\mathbf{Y}_{t+1}} = \Phi_{t+1} = n_{t+1}^{PA} \Phi_t.$$
(61)

From (61) follows that Φ_t is a martingale: $\mathbb{E}_t \Phi_{t+1} = \mathbb{E}_t n_{t+1}^{PA} \Phi_t = \Phi_t$, unless the wealth constraint $0 \leq [\Omega_t + \mathbb{E}_t n_{t+1}^{PA} Y_{t+1} - Y_t]$ is non-binding, in which case $\Phi_t = 0$, and the government reverts to a social planner.

7.2. Pessimistic Planner, Skeptical Consumer

This section treats a variation of the paternalistic planner in Section 7.1.3, where, instead of trusting the approximating model π , the authority has pessimistic doubts about it, meaning it now faces two kinds of ambiguity: one that derives from its ignorance of private beliefs, indexed by M, and the other stemming from its own doubts about the model, indexed by N. Accordingly, the planner minimizes with respect to both m and n, given two Lagrange penalty functions. For the sake of simplicity, I will assume that in the following recursion, a single risk sensitivity σ applies to both kinds of ambiguities:

$$\begin{aligned} \mathcal{V}_{t}(\mathbf{Y}_{t}, k_{t}, Q_{t}) &= \max_{c_{t}, E_{t}, k_{t+1}, Q_{t+1} \mathbf{Y}_{t+1}} \min_{n_{t+1}, m_{t+1}} u(c_{t}) \\ &- \frac{\beta}{\sigma} \mathbb{E}_{t}(n_{t+1} \log n_{t+1} + m_{t+1} \log m_{t+1}) \\ &+ \beta \mathbb{E}_{t} n_{t+1} \mathcal{V}_{t+1}(\mathbf{Y}_{t+1}, k_{t+1}, Q_{t+1}) \\ &+ \Phi_{t}[\Omega_{t} + \mathbb{E}_{t} m_{t+1} \mathbf{Y}_{t+1} - \mathbf{Y}_{t}] + \mathcal{M}_{t}. \end{aligned}$$

7.2.1. Inner Minimization

Minimization with respect to *n* and *m* produces the following worst-case multipliers

$$n_{t+1}^p = \frac{e^{\sigma \mathcal{V}_{t+1}}}{\mathbb{E}_t e^{\sigma \mathcal{V}_{t+1}}},\tag{62}$$

$$m_{t+1}^p = \frac{e^{\sigma \Phi_t Y_{t+1}}}{\mathbb{E}_t e^{\sigma \Phi_t Y_{t+1}}}.$$
(63)

7.2.2. Outer Maximization with Implied Risk-Sensitive Recursion

Substitution as before yields the risk-sensitive recursion

$$\mathcal{V}(\mathbf{Y}_{t}, k_{t}, Q_{t}) = \max_{c_{t}, H_{t}, \mathcal{E}_{t}, k_{t+1}, Q_{t+1}, \mathbf{Y}_{t+1}} u(c_{t}, H_{t}) + \Phi_{t}[\Omega_{t} - \mathbf{Y}_{t}] + \mathcal{M}_{t} + \frac{\beta}{\sigma} (\log \mathbb{E}_{t} e^{\sigma \mathcal{V}_{t+1}} + \log \mathbb{E}_{t} e^{\sigma \Phi_{t} \mathbf{Y}_{t+1}}).$$
(64)

The first-order conditions for $\{k_{t+1}, Q_{t+1}, Y_{t+1}\}$ are

$$1 = \frac{\beta}{\lambda_t} \mathbb{E}_t n_{t+1}^p \mathcal{V}_{k_{t+1}} = \beta \mathbb{E}_t n_{t+1}^p \frac{\lambda_{t+1}}{\lambda_t} (1 - \delta + Y_{k_{t+1}}), \tag{65}$$

$$\varphi_t = \beta \mathbb{E}_t n_{t+1}^p \mathcal{V}_{Q_{t+1}} = \beta \mathbb{E}_t n_{t+1}^p (\varphi_{t+1} + \lambda_{t+1} x_{t+1} D_{t+1} F_{t+1})$$
(66)

$$\Phi_t = -\mathcal{V}_{Y_{t+1}} = \Phi_{t+1} = \frac{m_{t+1}^r}{n_{t+1}^p} \Phi_t.$$
(67)

Those for $\{c, H, E\}$ remain the same as before. A special case:

1. The consumer has rational expectations ($m^p = 1$). As noted earlier, this case has been widely treated in papers on robust climate policy.

7.3. Pessimistic Planner, Pessimistic Consumer

This section treats a variation on the preceding belief regime by replacing its skeptical consumers with the pessimistic consumers from Section 3.4. Doing so will require adding two more implementability constraints: (1) the law of motion for the households' worst-case beliefs M_t^c in (43), because the authority needs to keep track of its evolution, and (2) the consumer's risk-sensitive utility recursion (44), because increments to the worst-case

likelihood ratio M_t^c are determined by that household's utility U_t .²⁹ For this policy maker, the Bellman recursion is,

$$\begin{split} \mathcal{V}_{t}(\mathbf{Y}_{t}, k_{t}, Q_{t}) &= \max_{c_{t}, E_{t}, k_{t+1}, Q_{t+1} \mathbf{Y}_{t+1}, \mathcal{M}_{t}^{c}, \mathcal{U}_{t}} \min_{n_{t+1}, u}(c_{t}) - \frac{\beta}{\sigma} \mathbb{E}_{t}(n_{t+1}\log n_{t+1}) \\ &+ \beta \mathbb{E}_{t} n_{t+1} \mathcal{V}_{t+1}(\mathbf{Y}_{t+1}, k_{t+1}, Q_{t+1}) \\ &+ \Phi_{t}[\Omega_{t} + \beta \mathbb{E}_{t} \frac{\mathcal{M}_{t+1}^{c}}{\mathcal{M}_{t}^{c}} \mathbf{Y}_{t+1} - \mathbf{Y}_{t}] + \mathcal{M}_{t} \\ &+ \beta \mathbb{E}_{t} \mu_{t+1} \left[\frac{e^{\sigma^{c} \mathcal{U}_{t+1}}}{\mathbb{E}_{t} e^{\sigma^{c} \mathcal{U}_{t+1}}} \mathcal{M}_{t}^{c} - \mathcal{M}_{t+1}^{c} \right] \\ &+ \varepsilon_{t} \left[u(c_{t}, H_{t}) + \frac{\beta}{\sigma^{c}} \log \mathbb{E}_{t} e^{\sigma^{c} \mathcal{U}_{t+1}} - \mathcal{U}_{t} \right] + \mathcal{M}_{t}, \end{split}$$

where μ_{t+1} and ε_t are the Lagrangian shadow prices for the law of motion for M_{t+1}^c and the consumer's worst-case utility U_t , respectively.

Outer Maximization with Implied Risk-Sensitive Recursion

With m^c computed by the consumer in Section 3.4,

$$m_{t+1}^c = \frac{M_{t+1}^c}{M_t^c} = \frac{e^{\sigma_n \mathcal{U}_{t+1}}}{\mathbb{E}_t e^{\sigma_n \mathcal{U}_{t+1}}},$$

and n^p the worst-case multiplier chosen by this planner,

$$n_{t+1}^p = rac{e^{\sigma_n \mathcal{V}_{t+1}}}{\mathbb{E}_t e^{\sigma_n \mathcal{V}_{t+1}}},$$

the risk-sensitive recursion to be solved under dual ambiguities is,

$$\begin{aligned} \mathcal{V}(\mathbf{Y}_{t}, k_{t}, Q_{t}) &= \max_{c_{t}, E_{t}, M_{t}^{c}, \mathcal{U}_{t} k_{t+1}, Q_{t+1}, Y_{t+1}} u(c_{t}) + \Phi_{t} [\Omega_{t} + \beta \mathbb{E}_{t} \frac{M_{t+1}^{c}}{M_{t}^{c}} \mathbf{Y}_{t+1} - \mathbf{Y}_{t}] \\ &+ \frac{\beta}{\sigma} \log \mathbb{E}_{t} e^{\sigma \mathcal{V}_{t+1}} + \beta \mathbb{E}_{t} \mu_{t+1} \left[\frac{e^{\sigma^{c} \mathcal{U}_{t+1}}}{\mathbb{E}_{t} e^{\sigma^{c} \mathcal{U}_{t+1}}} M_{t}^{c} - M_{t+1}^{c} \right] \\ &+ \varepsilon_{t} \left[u(c_{t}, H_{t}) + \frac{\beta}{\sigma^{c}} \log \mathbb{E}_{t} e^{\sigma^{c} \mathcal{U}_{t+1}} - \mathcal{U}_{t} \right] + \mathcal{M}_{t}. \end{aligned}$$

The first-order condition for *c* changes a little from before and becomes

$$c_t: \quad \lambda_t = [1 + \varepsilon_t] u_c + \Omega_{c_t} \Phi_t, \tag{68}$$

while the condition for *E* remains (51). The first-order conditions for $\{k_{t+1}, Q_{t+1}, Y_{t+1}\}$ are

$$1 = \beta \mathbb{E}_{t} n_{t+1}^{p} \frac{\lambda_{t+1}}{\lambda_{t}} (1 - \delta + Y_{k_{t+1}}),$$
(69)

$$\varphi_{t} = \beta \mathbb{E}_{t} n_{t+1}^{p} \mathcal{V}_{Q_{t+1}} = \beta \mathbb{E}_{t} n_{t+1}^{p} (\varphi_{t+1} + \lambda_{t+1} x_{t+1} D_{t+1} F_{t+1}),$$
(70)

$$\Phi_{t+1} = \frac{m_{t+1}}{n_{t+1}^p} \Phi_t, \tag{71}$$

$$0 \leq \Phi_t[\Omega_t + \mathbb{E}_t m_{t+1}^c Y_{t+1} - Y_t].$$
(72)

From (71), Φ_t is the submartingale:

$$\mathbb{E}_t \Phi_{t+1} \ge \frac{\mathbb{E}_t m_{t+1}^c}{\mathbb{E}_t n_{t+1}^p} \Phi_t = \Phi_t, \tag{73}$$

unless the wealth constraint is nonbinding, when, by the Kuhn-Tucker condition (72), $\Phi_t = 0$. Finally, given endogenous belief distortions M^c in the private sector, the first-order conditions with respect to M_t^c and U_t are

$$M_t^c: \quad \mu_t - \frac{\Phi_{t-1}}{M_{t-1}^c} Y_t = \beta \mathbb{E}_t m_{t+1}^c \left(\mu_{t+1} - \frac{\Phi_t}{M_t^c} Y_{t+1} \right), \tag{74}$$

$$\mathcal{U}_t: \quad \varepsilon_t = \sigma^c m_t^c M_{t-1}^c (\mu_t - \mathbb{E}_{t-1} m_t^c \mu_t) + m_t^c \varepsilon_{t-1}. \tag{75}$$

The forward solution of (74) implies that the shadow value of increasing m_t^c is proportional to the value of debt (to the consumer) in units of the marginal utility of consumption,

$$\mu_t = \frac{\Phi_{t-1}}{M_{t-1}^c} Y_t = \frac{\Phi_{t-1}}{M_{t-1}^c} u_{c,t} b_t,$$
(76)

obtained by using (42). Substituting this in (75) yields

$$\varepsilon_t = \sigma^c \Phi_{t-1} m_t^c u_{ct} (b_t - \mathbb{E}_{t-1} m_t^c b_t) + m_t^c \varepsilon_{t-1}, \tag{77}$$

where the term in parentheses is the innovation in government debt, with positive surprises producing a negative shock to the pessimistic likelihood ratio.

8. The Equilibrium Price of Capital

The consumption Euler condition (68) implies the discount factor,

$$\varrho_{t+j,t}^{*} \equiv \beta \frac{\lambda_{t+j}}{\lambda_{t}} \\
= \beta \frac{(1+\varepsilon_{t+j})u_{c_{t+j}} + \Omega_{c_{t+j}}\Phi_{t+j}}{(1+\varepsilon_{t})u_{c_{t}} + \Omega_{c_{t}}\Phi_{t}} \\
= \beta \frac{u_{c_{t+j}}}{u_{c_{t}}} \frac{1+\varepsilon_{t+j} + [1-\gamma+\gamma\vartheta_{t+1}]\Phi_{t+j}}{1+\varepsilon_{t+j}(1-\gamma+\gamma\vartheta_{t}]\Phi_{t}} \\
\equiv \Psi(\varepsilon_{t+j}, n_{t+j}^{*})\varrho_{t+j,t},$$
(78)

where

$$\Psi(\varepsilon_{t+j}, n_{t+j}^*) = \frac{1 + \varepsilon_{t+j} + [1 - \gamma + \gamma \vartheta_{t+j}] n_{t+j}^* \Phi_{t+j-1}}{1 + \varepsilon_t + [1 - \gamma + \gamma \vartheta_t] \Phi_t},$$
(79)

 $\vartheta_t = \frac{H_t + g_t}{c_t}$ is the inverse of the average propensity to consume, namely the ratio of wage income plus the lump-sum rebate to consumption, and n_{t+j}^* , associated with the planner's implementability constraint on household wealth, varies according to policy regime as shown in Table 1. The shadow prices μ and ε are zero, except when the private sector has pessimistic beliefs $m = m^c$.

For the Ramsey plans derived previously, the conditions for capital k imply the distorted discount factor,

$$\varrho_{t+j,t}^{**} \equiv n_{t+j}^{**} \beta \frac{\lambda_{t+j}}{\lambda_t} = n_{t+j}^{**} \Psi(\varepsilon_{t+j}, n_{t+j}^*) \varrho_{t+j,t},$$
(80)

where, like n_{t+j}^* , n_{t+j}^{**} varies by belief regime as shown in Table 1.³⁰ Note that from (78) and (80), the discount factor $\varrho_{t+j,t}^*$ is an n^{**} -distorted version of the previous consumption discount factor $\varrho_{t+j,t}^*$. The corresponding t + j distorted equilibrium price of capital is

$$\hat{p}_{t+j,t}^{**} = n_{t+j}^{**} \hat{p}_{t+j,t}^{*}.$$
(81)

	т	<i>n</i> *	n**	Ŷ	\hat{p}^*	\hat{p}^{**}
Skeptical consumers						
Political planner	m^s	1	n^{PO}	$m^{s}p$	$\Psi(0,1)p$	$n^{PO}\Psi(0,1)p$
Paternalistic planner	m^s	n^{PA}	1	$m^{s}p$	$\Psi(0, n^{PA})p$	$\Psi(0, n^{PA})p$
Pessimistic planner	m^s	$\frac{m^p}{n^p}$	n^p	$m^{s}p$	$\Psi(0, \frac{m^p}{n^p})p$	$n^p \Psi(0, \frac{m^p}{n^p}) p$
Pessimistic consumers		11.				
Pessimistic planner	m^{c}	$\frac{m^c}{n^p}$	n^p	$m^{c}p$	$\Psi(\varepsilon, \frac{m^c}{n^p})p$	$n^p \Psi(\varepsilon, \frac{m^c}{n^p}) p$

Table 1. Equilibrium Arrow–Debreu prices in alternative belief regimes.

9. A Comment about Methodology

In the extant literature, the typical approach to finding analytically tractable solutions to the kind of maximin dynamic programming problems posed in Section 7 is to form their Isaacs-Bellman-Flemming equations that involve guessing and verifying functional forms as well as specifying detailed assumptions regarding preferences and probability distributions. See for example Hennlock (2009) and Li et al. (2016). As a bit of an exception, Anderson et al. (2013) do provide some analytic insights based on their problem's first-order conditions, especially regarding the importance of the role of deep uncertainty in measuring the TCRE parameter defined in Section 4. However, they leave more detailed conclusions to an evaluation of numerical solutions of their stochastic finite-horizon robust optimization problems.

An innovation of this paper is to derive specific and detailed formulas for the social cost of carbon and both carbon and capital taxes through evaluations of the Euler conditions derived from optimization. As will become apparent in the next three sections, each case will reveal important roles for certain second-order moments in the economy, specifically for a number of covariances between measures of the martingale belief distortions and a variety of economic variables, including net returns to fossil energy and capital that arise as a consequence of evaluating the expectations of products of random variables.

Given some key assumptions and a number of results stated as lemmas in Section 13.1, it becomes possible to sign these covariances, allowing us to determine with fair accuracy the likely signs of ambiguity premiums that must be added to the social cost of carbon, the carbon tax, and any capital subsidy in the various belief regimes analyzed in this paper.

10. The Social Cost of Carbon

The generic form of the first-order condition for carbon stores Q_t , based (alternatively) on (56), (61), (66) and (70), is

$$\varphi_t = \beta \mathbb{E}_t n_{t+1}^{**} (\varphi_{t+1} + \lambda_{t+1} x_{t+1} D_{t+1} F_{t+1}).$$
(82)

Defining the marginal-utility scaled shadow price $\omega_t = \varphi_t / \lambda_t$, this becomes

Formula (83) gives a recursion for worst-case climate-caused damages from the point of view of a planner who may or may not be facing ambiguity, depending on n^{**} . The *social cost of carbon* is its forward solution,

$$\omega_t = \mathbb{E}_t \lim_{T \to \infty} \sum_{j=0}^T (\prod_{i=0}^t p_{t+i,t}^{**}) x_{t+j} D_{t+j} F_{t+j}.$$
(84)

Formula (84) reveals that damages are priced at $p_{t+j,t}^{**} = q_{t+j,t}^{**} \pi_{t+j}$, i.e., the current *t*-period robust Arrow–Debreu price of capital in Formula (81), demonstrating an equivalence

 $A = a \left[r_2 + r_2 D E \right]$ (95)

$$\Lambda_{t+1} \equiv \varrho_{t+1}[\omega_{t+1} + x_{t+1}D_{t+1}F_{t+1}], \tag{85}$$

and also

$$f(\Phi_t, \varepsilon_t) \equiv \frac{1}{1 + \varepsilon_t + \Phi_t [1 - \gamma + \gamma \vartheta_t]} \le f(\Phi_t, 0) < f(0, 0) = 1,$$
(86)

where $\vartheta_t = \frac{H_t + g_t}{c_t} \forall t$. Then use (78) and (79) to write the worst-case social cost of carbon (83) as a distortion of Λ_{t+1}

Notably, the SCC is the sum of two parts. The first part is

 $I: f(\Phi_t, \varepsilon_t)(1 + \varepsilon_{t+1}) \mathbb{E}_t [n_{t+1}^{**} \Lambda_{t+1}],$

which discounts future damages valued at the planner's n^{**} -distorted price of damages. Based on Table 1, n^{**} is n^{PO} for the political planner, 1 for the paternalistic planner, and n^p for the pessimistic planner facing either climate-skeptic or climate-pessimistic consumers.

The second part is

II:
$$f(\Phi_t, \varepsilon_t) \mathbb{E}_t [n_{t+1}^{**} n_{t+1}^* \Lambda_{t+1} (1 - \gamma + \gamma \vartheta_{t+1}) \Phi_t],$$

which augments damages in part I with the value of the planner's implementability constraint (if binding) at the n^* -distorted price of the net benefits of fossil energy. Based on Table 1, the combined ambiguity distortion $n^* \times n^{**}$ is n^{PO} for the political planner, n^{PA} for the paternalistic planner, m^p for the pessimistic planner facing climate-skeptical consumers, and m^c for the pessimistic planner facing climate-pessimistic consumers.

In the special instance when beliefs are rational and homogeneous, and the Ramsey planner's implementability constraint (39) is not binding ($\Phi_t = 0$), the social cost of carbon is the expected value of Λ_{t+1} (see for example Golosov et al. (2014)),

$$\omega_t^{SP} = \mathbb{E}_t \varrho_{t+1} [\omega_{t+1} + x_{t+1} D_{t+1} F_{t+1}] = \mathbb{E}_t \Lambda_{t+1} \equiv \Lambda_t^s.$$
(88)

11. The Carbon Tax

The conventional formula for the optimal carbon tax, as derived in Golosov et al. (2014), is

Social Planner's CarbonTax :
$$\tau_t^{e-SP} = \omega_t^{SP} = \Lambda_t^s$$
. (89)

This section shows that when beliefs are heterogeneous, the preceding formula for the carbon tax is inadequate.

The Ramsey planner's first-order condition (51) with respect to energy E_t is

$$Y_{E_t}\lambda_t - \varphi_t = v_t$$

where *Y* is defined in (20). Advancing one period and taking discounted expectations, the preceding expression implies,

$$\mathbb{E}_t \beta (Y_{E_{t+1}} \lambda_{t+1} - \varphi_{t+1}) = v.$$

Together, these two expressions imply

$$\mathbb{E}_{t} \frac{\beta \frac{\lambda_{t+1}}{\lambda_{t}} (Y_{E_{t+1}} - \varphi_{t+1} / \lambda_{t+1})}{Y_{E_{t}} - \varphi_{t} / \lambda_{t}} = \mathbb{E}_{t} \frac{\varrho_{t+1}^{*} (Y_{E_{t+1}} - \omega_{t+1})}{Y_{E_{t}} - \omega_{t}} = 1.$$
(90)

In Formula (90), $q_{t+1,t}^*$ is the discount factor a Ramsey planner applies to future expected excess returns to energy over the social cost of carbon—which I shall call the *Net Social Benefit of Carbon*. The corresponding t + j equilibrium price of fossil energy is

$$\hat{p}_{t+i,t}^* = \varrho_{t+i,t}^* \pi_{t+j}(x_{t+j}|x^t).$$
(91)

Market equilibrium requires that returns to all assets and activities be equal:

$$\mathbb{E}_{t}\left[\frac{\varrho_{t+1}^{*}(Y_{E_{t+1}} - \omega_{t+1})}{Y_{E_{t}} - \omega_{t}}\right] = \mathbb{E}_{t}\varrho_{t+1}^{**}(1 - \delta + Y_{k_{t+1}}) = 1,$$
(92)

where the right-hand side comes from Euler Equation (55) (or (60), (65), (69)).

Formula (92) is a version of Hotelling's welfare-optimal rule and posits that, in equilibrium, the expected socially optimal pre-tax gross return on the extant stock of fossil fuel remaining in the ground, which society discounts with stochastic discount factor q^* , is equal to the expected pre-tax gross return to capital in place, discounted using the government's SDF q^{**} . By comparison, Hotelling's rule, when derived for laissez-faire in (31), posits an equivalence between the private after-tax return to fossil fuel in the ground and the private after-tax gross return to the extant stock of capital, discounted using the private sector's distorted SDF \hat{q}_{t+1} . Since both q^*_{t+1} and \hat{q}_{t+1} may reflect ambiguous beliefs, formula (92) is potentially a robust version of Hotelling's rule. Note that the expected sign of the numerator is positive (or negative) in period t + 1 if its sign is positive (or negative) in period t.

Market equilibrium for the price of fossil energy $p^e = Y_E$ was previously derived as obeying the difference Equation (28),

$$Y_{E_t} - \tau_t^e = \mathbb{E}_t \hat{\varrho}_{t+1} (Y_{E_{t+1}} - \tau_{t+1}^e), \tag{93}$$

while, from (90), a socially optimal energy price must obey the rule

$$Y_{E_t} - \mathcal{O}_t = \mathbb{E}_t \varrho_{t+1}^* (Y_{E_{t+1}} - \mathcal{O}_{t+1}).$$

Since, in equilibrium, both expressions must hold, subtract the first from the second equation to eliminate Y_{E_i} :

$$\begin{aligned} \tau_t^e - \varpi_t &= \mathbb{E}_t \varrho_{t+1}^* [Y_{E_{t+1}} - \varpi_{t+1}] - \mathbb{E}_t [\hat{\varrho}_{t+1} (Y_{E_{t+1}} - \tau_{t+1}^e)] \\ &+ \mathbb{E}_t \hat{\varrho}_{t+1} \varpi_{t+1} - \mathbb{E}_t \hat{\varrho}_{t+1} \varpi_{t+1} \\ &= \mathbb{E}_t \hat{\varrho}_{t+1} [\tau_{t+1}^e - \varpi_{t+1}] + \mathbb{E}_t \varrho_{t+1}^* [Y_{E_{t+1}} - \varpi_{t+1}] - \mathbb{E}_t \hat{\varrho}_{t+1} (Y_{E_{t+1}} - \varpi_{t+1}). \end{aligned}$$

Adding and subtracting $E_t \hat{\varrho}_{t+1} \omega_{t+1}$ yields

$$\tau_t^e - \omega_t = \mathbb{E}_t \hat{\varrho}_{t+1} [\tau_{t+1}^e - \omega_{t+1}] + \mathbb{E}_t Z_{t+1}, \qquad (94)$$

where

$$Z_{t+1} = (\varrho_{t+1}^* - \hat{\varrho}_{t+1})(Y_{E_{t+1}} - \omega_{t+1}),$$

is the difference between the government's discounted *Net Social Benefit* of fossil fuel and the private sector's discounted *Net Social Benefit*, where, from (90), $Y_{E_{t+j}} \ge \omega_{t+j}$. Intuitively, the carbon tax exceeds the social cost of carbon ω in every period if $\forall j > 0$, the social discount factor is higher than the private discount factor, $\varrho_{t+j}^* \ge \hat{\varrho}_{t+j}$. So if, in every period,

the private sector is more myopic, i.e., less patient with respect to returns to fossil fuels than the planner, the government will add a premium to the carbon tax above the expected social cost of carbon ω . Exactly how and by how much, is determined as follows.

Lemma 1 (The carbon tax). *The carbon tax is the sum of two terms: (1) the expected social cost of carbon* ω_t *and (2) a premium* χ_t *:*

$$\hat{\tau}_t^e = \omega_t + \chi_t, \tag{95}$$

where, utilizing (17),

$$\chi_{t} \equiv \mathbb{E}_{t} \sum_{j=0}^{\infty} \prod_{i=0}^{j} \hat{\varrho}_{t+i} Z_{t+j} = \mathbb{E}_{t} \sum_{j=0}^{\infty} \beta^{j} \left(\prod_{i=1}^{j} m_{t+i} \frac{u_{c}(x^{t+i})}{u_{c}(x^{t})} \right) Z_{t+j}$$

$$= \mathbb{E}_{t} \sum_{j=0}^{\infty} \frac{M_{t+j}}{M_{t}} \varrho_{t+j,t} Z_{t+j} = \mathbb{E}_{t} \sum_{j=0}^{\infty} \hat{p}_{t+j,t} (\varrho_{t+j}^{*} - \hat{\varrho}_{t+j}) (Y_{E_{t+j}} - \omega_{t+j})$$

$$= \mathbb{E}_{t} \sum_{j=0}^{\infty} \hat{p}_{t+j,t} Z_{t+j}.$$
(96)

Proof. Solve the difference Equation (94). \Box

The first component of $\hat{\tau}^e$ is the Social Cost of Carbon previously derived that may or may not already contain an ambiguity-related premium. The second component adds a further ambiguity premium depending on belief regime. In essence, a government uses this formula to impose a premium on (or grant a discount toward) the carbon tax if cumulative expected differentially discounted net private benefits of fossil fuel are positive (or negative), where the sign and size of the premium (or discount) χ_t depends on the signs and sizes of all future Z_{t+j} priced at $\hat{p}_{t+j,t}$ that need to be determined. Note that in Formula (96), $\hat{p}_{t+j,t}(Y_{E_{t+j}} - \omega_{t+j})$ is the (possibly belief distorted) market value of the excess of private returns of fossil fuels over their social cost in period t + j. The premium χ_t is the expected sum of all such terms, each multiplied by the difference between the government's and the private sector's discount factor. In general, the premium is positive if the public is myopic compared with the planner: $\varrho_{t+j}^* - \hat{\varrho}_{t+j} > 0$ in all periods. Intuitively, a skeptical consumer who disbelieves the seriousness of climate change, will tend to use more energy than warranted from society's point of view because it myopically undervalues future climate costs. By contrast, a pessimistic private sector would opt to use less. As will become apparent later, this calculus is modified, if the authority itself has ambiguity about private beliefs.

An asset-pricing interpretation of (95) is that the optimal carbon tax is the sum of two possibly robust asset prices: (1) expected cumulative fossil fuel damages per unit of carbon valued at prices p_{t+j}^{**} in (84), and (2) cumulative net private benefits per unit of carbon over the social cost of carbon, valued at \hat{p}_{t+j} and weighted by the difference in discount factors $\varrho_{t+j}^* - \hat{\varrho}_{t+j}$. The interpretation of $\hat{\tau}_t^e$ as the potentially robust price of an underlying asset—the government-imposed cap on emissions—extends a result by Belfiori (2017), who also derived an equivalence between the optimal carbon tax and the optimal price of traded carbon permits in an economy with a cap and trade.

The case of a social planner under rational expectations, in which $\Phi = 0$ and beliefs are undistorted and homogeneous, is a suitable benchmark for comparison. However, it is exceptional in that in all other policy/belief regimes, the government may either impose a premium or give a concession. The premium (or concession) in Formula (96) is the expected value of the sum of products of random variables over all t + j, so the sign is not obvious from inspection, although it can be deciphered via decomposition into covariance components derived in Appendix D, which distinguishes between two cases for each of the possible policy regimes when either (1) the wealth constraint in (39) is binding ($\Phi_t > 0$), implying Ramsey planning, or (2) it is not ($\Phi_t = 0$), implying social planning.

To pursue this, define a new variable Ξ_{t+j}^e as the weighted proportional difference between the two stochastic discount factors ϱ_{t+j}^e and $\hat{\varrho}_{t+j}$:

$$\Xi_{t+j}^{e} \equiv \varsigma_{t+j}^{e} \frac{\varrho_{t+j}^{*} - \hat{\varrho}_{t+j}}{\hat{\varrho}_{t+j}} = \varsigma_{t+j}^{e} \left(\frac{\varrho_{t+j}^{*}}{\hat{\varrho}_{t+j}} - 1 \right), \tag{97}$$

where

$$\varsigma_{t+j}^{e} \equiv \frac{\hat{\varrho}_{t+j}(Y_{E_{t+j}} - \omega_{t+j}),}{\mathbb{E}_t \hat{\varrho}_{t+j}(Y_{E_{t+j}} - \omega_{t+j}),} \Longrightarrow \mathbb{E}_t \varsigma_{t+j}^{e} = 1,$$
(98)

is the *j*-th period's normalized discounted excess return to fossil energy over the social cost of carbon. Utilizing (78), Ξ_{t+j}^e is

$$\Xi_{t+j}^{e} = \varsigma_{t+j}^{e} (\frac{\Psi(\varepsilon_{t+j}, n_{t+j}^{*})}{m_{t+j}} - 1),$$
(99)

where $\Psi(\varepsilon_{t+j}, n_{t+j}^*)$ is defined in (79). Notice that Ξ^e depends on the government's belief multiplier n^* which varies according to policy regime, as shown in Table 1, and on the private sector's belief distortion $m_{t+1} = m_{t+1}^c$ when consumers are pessimistic and $m_{t+1} = m_{t+1}^s$ when they are skeptical. Importantly, whereas m^s is an exogenous martingale process, m^c is the pessimistic consumer's worst-case multiplier that depends on continuation utility as derived in (6). From the preceding, using (78) and (79), the expected value of Ξ_{t+j}^e is

$$\mathbb{E}_{t} \mathbb{E}_{t+j}^{e} = \mathbb{E}_{t} \frac{\varrho_{t+j}^{*} - \hat{\varrho}_{t+j}}{\hat{\varrho}_{t+j}} \varsigma_{t+j}^{e} \\
= \mathbb{E}_{t} \left[\left(\frac{1}{m_{t+j}} \frac{1 + \varepsilon_{t+j} + [1 - \gamma + \gamma \vartheta_{t+j}] n_{t+j}^{*} \Phi_{t+j-1}}{1 + \varepsilon_{t} + [1 - \gamma + \gamma \vartheta_{t}] \Phi_{t}} - 1 \right) \varsigma_{t+j}^{e} \right]. \quad (100)$$

Note that (100) gives the difference between two non-centered covariances with the normalized discounted excess return to fossil energy, one involving the government's and the other the private sector's stochastic discount factor.

Lemma 2 is key to determining the sign of the carbon tax premium. Based on (95),

Lemma 2.

$$sign \ \chi_t \ = \ sign[\hat{\tau}^e_t - \omega_t] \ = sign \ \mathbb{E}_t \Xi^e_{t+j} \ \forall j.$$
(101)

Proof. First,

 \implies

$$\mathbb{E}_t Z_{t+j} \stackrel{\geq}{\equiv} 0, \quad if \quad \mathbb{E}_t \Xi^e_{t+j} \stackrel{\geq}{\equiv} \quad 0, \ j = 1, \cdots \infty$$

To show this, multiply and divide Z_{t+j} by $\hat{\varrho}_{t+j} \mathbb{E}_t \hat{\varrho}_{t+j} (Y_{E_{t+j}} - \omega_{t+j})$:

$$Z_{t+j} = Z_{t+j} \frac{\hat{\varrho}_{t+j}}{\hat{\varrho}_{t+j}} \frac{\mathbb{E}_{t} \hat{\varrho}_{t+j} (Y_{E_{t+j}} - \omega_{t+j})}{\mathbb{E}_{t} \hat{\varrho}_{t+j} (Y_{E_{t+j}} - \omega_{t+j})}$$

$$= \left[\frac{\hat{\varrho}_{t+j} (Y_{E_{t+j}} - \omega_{t+j})}{\mathbb{E}_{t} \hat{\varrho}_{t+j} (Y_{E_{t+j}} - \omega_{t+j})} \frac{\hat{\varrho}_{t+j}^{*} - \hat{\varrho}_{t+j}}{\hat{\varrho}_{t+j}} \right] \mathbb{E}_{t} \hat{\varrho}_{t+j} - \omega_{t+j})$$

$$= \left[\varsigma_{t+j}^{e} \frac{\hat{\varrho}_{t+j}^{*} - \hat{\varrho}_{t+j}}{\hat{\varrho}_{t+j}} \right] \mathbb{E}_{t} \hat{\varrho}_{t+j} (Y_{E_{t+j}} - \omega_{t+j})$$

$$= \Xi_{t+j}^{e} \mathbb{E}_{t} \hat{\varrho}_{t+j} (Y_{E_{t+j}} - \omega_{t+j}), \qquad (102)$$

$$\mathbb{E}_{t} Z_{t+j} = \mathbb{E}_{t} \Xi_{t+j}^{e} \mathbb{E}_{t} \hat{\varrho}_{t+j} (Y_{E_{t+j}} - \omega_{t+j}),$$

In the absence of any known time-dependent anomalies, there is no reason to believe that the sign of $\mathbb{E}_t \Xi_{t+i}^e$ is different at different times, justifying

Assumption 1. sign $\mathbb{E}_t \sum_{j=0}^{\infty} \Xi_{t+j}^e = sign \mathbb{E}_t \Xi_{t+j}^e \ \forall j$

Since in (96), the sign of χ_t depends on the sign of all its future expected elements, $\mathbb{E}_t \hat{p}_{t,t+j} Z_{t+j}$, we require

Lemma 3. If $\mathbb{E}_t \Xi^e_{t+j} \gtrless 0$ for any $j \ge 0$, then $\chi_t \gtrless 0$ for any $j \ge 0$.

Proof. First, in (96),

$$\mathbb{E}_t \hat{p}_{t+j,t} Z_{t+j} \stackrel{\geq}{=} 0, \quad if \quad \forall j \ge 1, \quad \mathbb{E}_t \hat{p}_{t+j,t} \Xi_{t+j}^e \stackrel{\geq}{=} 0.$$

The lemma follows by applying Assumption 1, and (96) and (102). \Box

A suitable formula for the carbon tax in terms of the Ξ_{t+j}^{e} is found by substituting (96) and (102) into (95):

$$\hat{\tau}_{t}^{e} = \omega_{t} + \mathbb{E}_{t} \sum_{j=0}^{\infty} \hat{p}_{t+j,t} \Xi_{t+j}^{e} \hat{\varrho}_{t+j} (Y_{E_{t+j}} - \omega_{t+j}).$$
(103)

The essence of the preceding is that if we know the sign of Ξ_{t+j}^e for any *j*, then we shall know the sign of any premium χ_t over the standard carbon tax. The implications of this for the different belief regimes are stated in several propositions in Section 13.

12. An Ex Ante Tax on Capital

While $p_t^e(x^t)$ and $\tau_t^e(x^t)$ are uniquely determined in (24) and (95), respectively, it is easy to demonstrate, following Chamley (1986), Zhu (1992), and Chari et al. (1994), that a state-by-state capital income tax is not uniquely determined because an implementable allocation $\{b, k, Q\}$ that uniquely determines household wealth W in (36) can be obtained by a multiplicity of capital tax and bond policies $\{\tau^k, b\}$ at prices $\{\hat{p}, r\}$.³² However, across states of nature, the history-dependent value of tax payments is fully determined. So with this in mind, define the *effective* or *ex ante* tax rate on capital as the ratio of the prices of two "assets", one yielding a stream of tax revenues and the other yielding a stream of gross before-tax capital returns, conditional on history x^t , where the latter is defined by

$$\bar{\tau}_{t+1}^{k}(x^{t}) \equiv \frac{\mathbb{E}_{t}\hat{p}_{t+1,t}(x_{t+1},x^{t})\tau_{t+1}^{k}(x^{t+1})(Y_{k_{t+1}}(x^{t+1})-\delta)}{\mathbb{E}_{t}\hat{p}_{t+1,t}(x_{t+1},x^{t})(Y_{k_{t+1}}(x^{t+1})-\delta))}.$$
(104)

In this formula, the *ex ante* tax rate $\bar{\tau}_{t+1}^k(x^t)$ is the ratio of expected tax revenues to expected gross returns from capital, conditional on history x^t and valued at possibly distorted Arrow–Debreu prices \hat{p} .

With the use of (14), (104) is re-written as

$$\overline{t}_{t+1}^{k}(x^{t}) \equiv \frac{\mathbb{E}_{t}\hat{p}_{t+1,t}(x_{t+1},x^{t})[1-\delta+Y_{k_{t+1}}(x^{t+1})]-1}{\mathbb{E}_{t}\hat{p}_{t+1,t}(x_{t+1},x^{t})(Y_{k_{t+1}}(x^{t+1})-\delta)}, \\
= \frac{\mathbb{E}_{t}\hat{\varrho}_{t+1}[1-\delta+Y_{k_{t+1}}]-1}{\mathbb{E}_{t}\hat{\varrho}_{t+1}(Y_{k_{t+1}}-\delta)}.$$
(105)

Based on consumption-Euler Equations (50) or (68),

$$1 = \mathbb{E}_{t} n_{t+1}^{**} \beta \frac{\lambda_{t+1}}{\lambda_{t}} (1 + Y_{k_{t+1}} - \delta) = \mathbb{E}_{t} n_{t+1}^{**} \beta \frac{[1 + \varepsilon_{t+1}] u_{c_{t+1}} + \Omega_{c_{t+1}} \Phi_{t+1}}{[1 + \varepsilon_{t}] u_{c_{t}} + \Omega_{c_{t}} \Phi_{t}} (1 + Y_{k_{t+1}} - \delta) = \mathbb{E}_{t} n_{t+1}^{**} \beta \frac{1 + \varepsilon_{t+1} + [1 - \gamma + \gamma \vartheta_{t+1}] n_{t+1}^{*} \Phi_{t+j-1}}{1 + \varepsilon_{t} + [1 - \gamma + \gamma \vartheta_{t}] \Phi_{t}} (1 + Y_{k_{t+1}} - \delta),$$
(106)

the government discounts *gross* returns to capital in period t + 1 using e^{**} , derived in (79), which is also used to discount future damages resulting from fossil use in (84), as shown earlier, where the distorting factor n_{t+1}^{**} varies according to belief regime as detailed in Table 1.

Substituting (107) into (105), an alternative expression for $\bar{\tau}_t^k$ is

$$\bar{\tau}_{t+1}^{k} = \frac{\mathbb{E}_{t}[(\hat{\varrho}_{t+1} - \varrho_{t+1}^{**})(1 - \delta + Y_{k_{t+1}}]}{\mathbb{E}_{t}[\hat{\varrho}_{t+1}(Y_{k_{t+1}} - \delta)]}.$$
(107)

Noteworthy in this expression is that it echoes the expression in (96) determining the sign of the carbon tax premium, likewise depending on the difference between the private sector's discount factor and that of the government, $\hat{\varrho}_{t+1} - \varrho_{t+1}^{**}$. Simply put, the numerator in (107) is the difference between two non-centered covariances between the stochastic pre-tax return $1 - \delta + Y_{k_{t+1}}$ and the private sector's and the government's SDF, respectively. In general terms, the ex ante tax on capital is negative if the private sector is more myopic, i.e., less patient with respect to returns to capital than the government, specifically, if the covariance of private returns with the private stochastic discount factor is less than the corresponding covariance with the planner's stochastic discount factor. Conversely, if private returns are better correlated with the government's stochastic discount factor, or are equal to it, then the average tax is positive or zero. Earlier literature by Chamley (1986), Judd (1985), and Atkeson et al. (1999), concludes that in a deterministic economy with time-additive preferences, the optimal *ex ante* tax on capital is zero, except possibly at time 0. Subsequently, Chari et al. (1994) and Zhu (1992) found that in a stochastic economy with homogeneous beliefs, the tax is positive (or negative) if a weighted average of the change in period elasticities of the utility function is positive (or negative). Below, I shall show how a divergence in beliefs between the private sector and government produces a gap between the government's and the private sector's discount factors that contribute additional motives to subsidize or penalize capital, depending on the source of ambiguity.

Now mimic expression (100) and define the weighted proportional difference between the private and the government's stochastic discount factors,

$$\Xi_{t+1}^{k} \equiv \varsigma_{t+1}^{k} \frac{\hat{\varrho}_{t+1} - \varrho_{t+1}^{**}}{\hat{\varrho}_{t+1}} = \varsigma_{t+1}^{k} (1 - \frac{\varrho_{t+1}^{**}}{\hat{\varrho}_{t+1}}) = \varsigma_{t+1}^{k} (1 - \frac{n_{t+1}^{**}}{m_{t+1}} \Psi(\varepsilon_{t+1}, n_{t+1}^{*})), \quad (108)$$

where

$$\varsigma_{t+1}^{k} \equiv \frac{\hat{\varrho}_{t+1}[1-\delta+Y_{k_{t+1}}]}{\mathbb{E}_{t}[\hat{\varrho}_{t+1}(1-\delta+Y_{k_{t+1}})]},\tag{109}$$

is the normalized discounted gross return to capital, such that $\mathbb{E}_{tS_{t+1}^k} = 1$, and where $\Psi(\varepsilon_{t+1}, n_{t+1}^*)$ is defined in (79). The next lemma proves that the sign of the capital tax equals the sign of $E_t \Xi_{t+1}^k$:

Lemma 4.

$$\bar{\tau}_{t+1}^k \stackrel{\leq}{\equiv} 0, \quad if \quad \mathbb{E}_t \Xi_{t+1}^k \stackrel{\leq}{\equiv} 0. \tag{110}$$

Proof. Use (109) in (108) and write,

$$\mathbb{E}_{t} \Xi_{t+1}^{k} \equiv \mathbb{E}_{t} [\varsigma_{t+1}^{k} \frac{\hat{\varrho}_{t+1} - \varrho_{t+1}^{**}}{\hat{\varrho}_{t+1}}] \\
= \mathbb{E}_{t} [\frac{\hat{\varrho}_{t+1} [1 - \delta + Y_{k_{t+1}}]}{\mathbb{E}_{t} [\hat{\varrho}_{t+1} (1 - \delta + Y_{k_{t+1}})]} \frac{\hat{\varrho}_{t+1} - \varrho_{t+1}^{**}}{\hat{\varrho}_{t+1}}] \\
= \frac{\mathbb{E}_{t} [\hat{\varrho}_{t+1} (Y_{k_{t+1}} - \delta)]}{\mathbb{E}_{t} [\hat{\varrho}_{t+1} (1 - \delta + Y_{k_{t+1}})]} \bar{\tau}_{t+1}^{k},$$
(111)

where the last term is obtained by multiplying and dividing by $\mathbb{E}_t[\hat{\varrho}_{t+1}(Y_{k_{t+1}} - \delta)]$ and using (107). The proof follows by observing that the term multiplying $\bar{\tau}_{t+1}^k$ is positive. \Box

For later use, re-write (111) as

$$\mathbb{E}_{t} \Xi_{t+1}^{k} = \mathbb{E}_{t} \varsigma_{t+1}^{k} \left(1 - \frac{n_{t+1}^{**}}{m_{t+1}} \frac{1 + \varepsilon_{t+1} + [1 - \gamma + \gamma \vartheta_{t+1}] n_{t+1}^{*} \Phi_{t}}{1 + \varepsilon_{t} + [1 - \gamma + \gamma \vartheta_{t}] \Phi_{t}} \right), \\
= \mathbb{E}_{t} \varsigma_{t+1}^{k} \left(1 - f(\Phi_{t}, \varepsilon_{t}) \frac{n_{t+1}^{**}}{m_{t+1}} (1 + \varepsilon_{t+1} + [1 - \gamma + \gamma \vartheta_{t+1}] n_{t+1}^{*} \Phi_{t} \right). \quad (112)$$

13. Policy Implications of Belief Heterogeneity and Ambiguity

13.1. Preliminary Results

Having derived formulas for the expected social cost of carbon, and carbon as well as capital taxation, we are now prepared to consider the policy implications for the various belief regimes, with particular focus on any premia or surcharges that may arise as a consequence of belief distortions and ambiguities. The results are summarized in six propositions. As will become apparent, the sign and size of surcharges on the carbon tax and those of the *ex ante* tax on capital will depend on the signs of several key second-order moments, namely covariances involving measures related to the state of the economy, ζ^k and ζ^{e} , the average propensity to consume out of disposable income ϑ , and measures of the intensity of robustness concerns reflected in the endogenous worst-case belief distortions m^c, m^p, n^p, n^{POL}, n^{PAT} that arise from evaluating Formulas (87), (100), and (112) containing the expectations of products and ratios of random variables. The resulting expressions involve sums of expectations of their individual components and their covariances, known to be non-trivial because, as shown later, the assumption of endogeneity means that all involved variables are functions of the fundamental climate shock process x_t and are therefore related to each other. The manner of deconstructing optimal tax policy applied in this paper is novel in the literature, which has mostly had to rely on simulations to quantify policy effects, and should be considered a unique contribution of this paper.

The signs of the covariances needed to evaluate expectations and to prove later propositions are not always clear from intuition and need to be formally derived and stated in the form of several lemmas.

Assumption 2. $\frac{dc_{t+1}}{dx_{t+1}} \leq 0.$

It is generally assumed that consumption declines with global warming (see Frankhouser and Tol 2005; Weitzman 2009).

Lemma 5.

$$\frac{d\hat{\varrho}_{t+j}}{dx_{t+j}} > 0$$

Proof.

$$\begin{aligned} \frac{d\hat{\varrho}_{t+j}}{dx_{t+j}} &= \beta M_{t+j} \frac{d\frac{u_{c_{t+j}}}{u_{c_t}}}{dx_{t+j}} + \beta \frac{u_{c_{t+j}}}{u_{c_t}} \frac{dM_{t+j}}{dx_{t+j}} \\ &= -\beta M_{t+j} \left(\frac{c_t}{c_{t+j}}\right)^{\gamma} \left(\gamma c_{t+j}^{-1} \frac{dc_{t+j}}{dx_{t+j}}\right) \\ &+ \beta \left(\frac{c_t}{c_{t+j}}\right)^{\gamma} \frac{dM_{t+j}}{dx_{t+j}} > 0, \end{aligned}$$

where first terms in each line are positive by Assumption 2, and the second lines are also positive since $\frac{dM_{t+j}}{dx_{t+j}} > 0$ if $M = M^c$, and zero otherwise. \Box

Assumption 3.
$$\frac{dk_{t+1}}{dx_{t+1}} < 0.$$

Li et al. (2016) assume that capital utilization declines upon a climate cost shock. Frankhouser and Tol (2005, p. 5) observe that "the overall effect of climate change on the accumulation of capital is in principle ambiguous", but that "it seems safe to speculate that the capital accumulation effect will probably be negative". More recent evidence that a significant portion (50%) of total GDP losses can be attributed to disincentives to invest capital is less ambiguous (see Willner et al. 2021).

Assumption 4. $\frac{dH_{t+1}}{dx_{t+1}} \leq 0.$

Labor input likely declines as a result of a climate cost shock since, as documented in Dasgupta et al. (2021); Kjellstrom (2014); Kjellstrom et al. (2009); Somanathan et al. (2018), productive labor is lost as a consequence of warming, justifying

Assumption 5. $\frac{dE_{t+1}}{dx_{t+1}} \simeq 0.$

Wilbanks et al. (2008) find that climate warming reduces energy use and production, as verified by the BEA (BEA 2019), although, as an Environmental Protection Agency web page, deleted by the Trump Administration in 2017 but saved and still available on (EPA 2017), reports, climate warming may lead to partially offsetting heating and cooling demands.

Lemma 6.

$$\frac{dY_{z_{i,t}}}{dx_t} = -(Q_0 - Q_t)D_t F_{z_{i,t}} + \sum_{j=1}^n \frac{\partial Y_{z_{i,t}}}{\partial z_{j,t}} \frac{dz_{j,t}}{dx_t},
\approx -(Q_0 - Q_t)D_t F_{z_{i,t}} < 0,$$
(113)

for all factors of production, k, E, and H.

Proof. The signs of $\frac{dz_{j,t}}{dx_t}$ are given in Assumptions 3–5, and $\frac{\partial Y_{z_{i,t}}}{\partial z_{j,t}} > 0$. Recent literature (Dasgupta et al. 2021; Njuki et al. 2020; Zhang et al. 2017) suggests that the indicated sum is likely negative but small, leaving as main driver of climate cost shocks their effects on total factor productivity, here basically represented by the damage function.

Assumption 6. $\frac{db_{t+1}}{dx_{t+1}} > 0.$

The effect of a climate shock on debt b_t can plausibly be said to be positive. In their study of Columbia and Peru, Maldonado and Gallagher (2022) provide some evidence that climate shocks significantly affect public debt trajectories towards significantly higher levels and, in some cases, raise probabilities of increasing debt during climate stress. Nor do developed economies in Europe seem to be immune from this effect. For example, Zenios (2022) combined projections from the IMF World Economic Outlook with simulations of versions of an IAM model obtained from Emmerling et al. (2016) and Gazzotti et al. (2021) to show that climate shocks will raise the sovereign debt-to-GDP ratio in Italy and Cyprus over time.

Lemma 7.

$$\frac{dm_{t+1}^c}{dx_{t+1}} \ge 0.$$

Proof. From (6),

$$\frac{dm_{t+1}^c}{dx_{t+1}} = \frac{\partial m_{t+1}^c}{\partial \mathcal{U}_{t+1}} \frac{d\mathcal{U}_{t+1}}{dx_{t+1}} = \sigma^c m_{t+1}^c (1 - m_{t+1}^c) \frac{d\mathcal{U}_{t+1}}{dx_{t+1}}.$$

The term multiplying $\frac{d\mathcal{U}_{t+1}}{dx_{t+1}}$ is negative, indicating that decreases in \mathcal{U}_{t+1} raise m_{t+1}^p toward 1. The envelope conditions (10)–(12) imply

$$\begin{aligned} \frac{d\mathcal{U}_{t+1}}{dx_{t+1}} &= \mathcal{U}_{k_{t+1}} \frac{dk_{t+1}}{dx_{t+1}} + \mathcal{U}_{b_{t+1}} \frac{db_{t+1}}{dx_{t+1}} + \mathcal{U}_{Q_{t+1}} \frac{dQ_{t+1}}{dx_{t+1}} \\ &= u_{c_{t+1}} \left[R_{t+1}^k \frac{dk_{t+1}}{dx_{t+1}} + \frac{db_{t+1}}{dx_{t+1}} \right] \le 0, \end{aligned}$$

if the negative effect on capital outweighs the presumably positive effect on debt. Otherwise, the result follows from $\sigma^c < 0$ and $m^c \le 1.^{33}$

Lemma 8.

$$rac{dn^{PO}_{t+1}}{dx_{t+1}} > 0$$

Proof. Evaluate

$$\begin{split} \frac{dn_{t+1}^{PO}}{dx_{t+1}} &= \frac{\partial n_{t+1}^{PO}}{\partial \mathcal{V}_{t+1}} \frac{d\mathcal{V}_{t+1}}{dx_{t+1}} + \frac{\partial n_{t+1}^{PO}}{\partial \mathcal{V}_{t+1}} \frac{dY_{t+1}}{dx_{t+1}} \\ &= \sigma n_{t+1}^{PO} (1 - n_{t+1}^{PO}) \left(\mathcal{V}_{k_{t+1}} \frac{dk_{t+1}}{dx_{t+1}} + \mathcal{V}_{Y_{t+1}} \frac{dY_{t+1}}{dx_{t+1}} + \bar{\Phi} \frac{dY_{t+1}}{dx_{t+1}} \right) \\ &= \sigma n_{t+1}^{PO} (1 - n_{t+1}^{PO}) \left((1 - \delta + Y_{k_{t+1}}) (u_{c_{t+1}} + \Omega_{c_{t+1}} \bar{\Phi}) \frac{dk_{t+1}}{dx_{t+1}} \right) \\ &+ \sigma n_{t+1}^{PO} (1 - n_{t+1}^{PO}) \left(\Phi_t \frac{dY_{t+1}}{dx_{t+1}} - \Phi_{t+1} \frac{dY_{t+1}}{dx_{t+1}} \right) \\ &= \sigma n_{t+1}^{PO} (1 - n_{t+1}^{PO}) \left((1 - \delta + Y_{k_{t+1}}) (u_{c_{t+1}} + \Omega_{c_{t+1}} \bar{\Phi}) \frac{dk_{t+1}}{dx_{t+1}} \right) > 0, \end{split}$$

which follows from $\sigma < 0$, $n^{PO} \le 1$, envelope conditions (52) and (54), Euler condition (50) with $\xi = 0$, from Assumption 3.³⁴, and from the previous result that $\Omega_{c_{t+1}} > 0.^{35}$

The preceding result accords with the properties $0 < n_{t+1}^{PO} \leq 1$, $\lim_{\sigma \uparrow 0} n_{t+1}^{PO} \to 1$, $\lim_{\sigma \downarrow -\infty} n_{t+1}^{PO} \to 0$, and the partial derivative

$$\frac{\partial n_{t+1}^{PO}}{\partial \sigma} = (\mathcal{V}_{t+1} + \Phi_t Y_{t+1})(1 - n_{t+1}^{PO})n_{t+1}^{PO} \ge 0.$$
(114)

As one might expect intuitively, increased ambiguity aversion ($\sigma \downarrow$) reduces n^{PO} and drives it towards zero, a result that echoes those in Gilboa and Schmeidler (1989) and Millner et al. (2012).

Lemma 9.

$$\frac{dn_{t+1}^{PA}}{dx_{t+1}} > 0.$$

Proof. From (39) and results (41),

$$\begin{aligned} \frac{dn_{t+1}^{PA}}{dx_{t+1}} &= \frac{\partial n_{t+1}^{PA}}{\partial Y_{t+1}} \frac{dY_{t+1}}{dx_{t+1}} = \sigma \Phi_t n_{t+1}^{PA} (1 - n_{t+1}^{PA}) \frac{dY_{t+1}}{dx_{t+1}} \\ &= \sigma \Phi_t n_{t+1}^{PA} (1 - n_{t+1}^{PA}) \left(\Omega_{c_{t+1}} \frac{dc_{t+1}}{dx_{t+1}} \right) \ge 0. \end{aligned}$$

Given $\sigma < 0$ and Assumption 2, this implies a positive correlation between n^{PA} and x. \Box

By the preceding arguments,

Lemma 10.

$$\frac{\partial n_{t+1}^p}{\partial x_{t+1}} \ge 0.$$

Proof. The proof follows argument similar to the proof of Lemma 10. \Box

The preceding discussion leads up to the following two lemmas:

Lemma 11. Given definition (109),

$$\frac{d\varsigma_{t+1}^k}{dx_{t+1}} \gtrless 0.$$

Proof. Based on Lemma 5 and the result in (113),

$$\begin{aligned} \frac{d\zeta_{t+1}^k}{dx_{t+1}} &= \frac{d\hat{\varrho}_{t+1}(1-\delta+Y_{k_{t+1}})}{dx_{t+1}} \\ &= (1-\delta+Y_{k_{t+1}})\frac{d\hat{\varrho}_{t+1}}{dx_{t+1}} + \hat{\varrho}_{t+1}\frac{dY_{k_{t+1}}}{dx_{t+1}} > 0. \end{aligned}$$

Lemma 12.

$$\frac{d\varsigma_{t+j}^e}{dx_{t+j}} > 0 \ \forall j > 0.$$

Proof. Use Lemma 5 and the result in (113) and note that, from Formula (98), the sign of $\frac{d\varsigma_{t+j}^e}{dx_{t+j}}$ is the same as the sign of

> $\cdot(\gamma)$ dâ

$$\frac{d\hat{\varrho}_{t+j}(Y_{E_{t+j}} - \varpi_{t+j})}{dx_{t+j}} = (Y_{E_{t+j}} - \varpi_{t+j})\frac{d\hat{\varrho}_{t+j}}{dx_{t+j}} + \hat{\varrho}_{t+j}\frac{dY_{E_{t+j}}}{dx_{t+j}} > 0.$$

Lemma 13. $\frac{d\Lambda_{t+j}}{dx_{t+j}} \ge 0.$

Proof. This follows from its definition in (85). \Box

Assumption 7.

$$\frac{d\vartheta_{t+j}}{dx_{t+j}} = \frac{d\frac{H_t + g_t}{c_t}}{dx_{t+j}} \approx 0.$$

Since climate lowers both income and consumption, the net effect on ϑ is likely negligible, justifying an assumption that ϑ is not strongly negatively correlated with Λ , ζ^k , ζ^{e} , m^{c} , n^{PO} , m^{c} , n^{PA} , and n^{p} , all of which are positively correlated with x.

Under the assumption that climate shocks are the sole stochastic process x driving the economy, the next proposition reflects the fact that variables that are functions of x must be correlated with each other. The signs of the covariances listed therein follow from the preceding lemmas.

Lemma 14. Signs of key covariances

(i) The signs of $cov(\Lambda, \vartheta)$, $cov(\varsigma^e, \vartheta)$, $cov(\varsigma^k, \vartheta)$, $cov(n^{PO}, \Lambda)$, $cov(n^{PO}, \Lambda\vartheta)$, $cov(n^{PA}, \Lambda)$, $cov(n^{PA}, \Lambda \vartheta), cov(n^{PA}, \vartheta \varsigma^e), cov(n^{PA}, \varsigma^k), cov(n^{PA}, \vartheta \varsigma^k), cov(m^{PA}, \Lambda), cov(m^{PA}, \vartheta \Lambda),$ $cov(n^p, \Lambda), cov(n^p, \zeta^k), cov(n^p, m^c), cov(m^p, \zeta^k), cov(m^p, \Lambda), cov(m^p, \vartheta\Lambda), cov(m^p \zeta^k, \vartheta),$ $cov(m^cm^p, \Lambda)$, $cov(m^c, \Lambda)$, and $cov(m^c, \vartheta\Lambda)$ are positive.

(*ii*) The signs of $cov(\varsigma^e, \frac{1}{m^e})$, $cov(n^p, \frac{1}{m^e})$, $cov(\varsigma^e, \frac{1}{n^p})$, and $cov(\vartheta \varsigma^e, \frac{1}{n^p})$ are negative. (*iii*) The signs of $cov(\varsigma^k, \frac{n^p}{m^e})$, $cov(\varsigma^e, \frac{n^p}{m^e})$, $cov(\varsigma^e, \frac{m^p}{n^p})$, and $cov(\varsigma^e\vartheta, \frac{m^p}{n^p})$ are indeterminate.

The following propositions distinguish two situations: (i) when the implementability (the marginal-utility-of-consumption value of household wealth) constraint is binding, $(\Phi > 0)$, and (ii) when it is not, $(\Phi = 0)$. In the latter case, all policies revert to those of an unconstrained social planner.³⁶ To set a baseline for comparison, the first proposition establishes results that obtain with homogeneous beliefs under rational expectations RE.

13.2. Homogeneous Beliefs

Proposition 1 (Benchmark rational expectations). If (i) beliefs are homogeneous and rational, and (ii) labor income plus tax rebates as a proportion of consumption do not decrease over time in all periods, then

1 The social cost of carbon exceeds the standard formulation Λ_t^s derived in (88) for a social planner, unless $\bar{\Phi} = 0$,

$$\begin{split} \omega_t^{RE} &= \Lambda_t^s + \gamma \bar{\Phi} f(\bar{\Phi}, 0) [(\mathbb{E}_t \vartheta_{t+1} - \vartheta_t) \Lambda_t^s + cov(\Lambda, \vartheta)] \geq \Lambda_t^s, \ \bar{\Phi} > 0, \ \forall t, \\ &= \Lambda_t^s, \ \Phi_t = 0 \ \forall t. \end{split}$$

2. The carbon tax premium is positive, unless $\Phi = 0$,

$$\mathbb{E}_{t} \Xi_{t+j}^{e-RE} = \gamma \bar{\Phi} f(\bar{\Phi}, 0) \left[\mathbb{E}_{t} \vartheta_{t+j} - \vartheta_{t} + cov(\varsigma^{e}, \vartheta) \right] > 0, \quad \bar{\Phi} > 0, \quad \forall t$$

= 0, $\bar{\Phi} = 0 \quad \forall t.$

3. The ex ante rate on capital is negative, unless $\Phi = 0$,

$$\mathbb{E}_{t} \Xi_{t+1}^{k-RE} = -\gamma \bar{\Phi} f(\bar{\Phi}, 0) \Big[\mathbb{E}_{t} \vartheta_{t+1} - \vartheta_{t} + cov(\varsigma^{k}, \vartheta) \Big] < 0, \quad \bar{\Phi} > 0, \forall t$$

= 0, $\bar{\Phi} = 0 \forall t.$

This proposition sets up a background and baseline for comparison with the conclusions for the remaining belief regimes. It also establishes an important distinction between social and Ramsey planning. For a social planner, defined as one for whom the implementability constraint (39) is not binding ($\Phi = 0$)—nor invoked in most of the literature—the policy settings coincide with the formulas derived in Golosov et al. (2014) for the optimal carbon tax and the social cost of carbon, and with the optimal tax on capital derived by Zhu (1992), Chari et al. (1994), and Atkeson et al. (1999). By contrast, a true Ramsey planner, being mindful of preferences and household budgets, adjusts all calculations by factors involving the inter-temporal rate of substitution and the average propensity to consume.

13.3. Heterogeneous Beliefs: Skeptical Consumers

The next proposition establishes that even without ambiguity, skepticism alters optimal policy.

Proposition 2 (No ambiguity). By producing incentives to spend more on fossil energy and less on capital than is socially optimal, the mere presence of skepticism is an inducement to the Ramsey planner but not a social planner to raise the carbon tax and to subsidize capital, where

1. Social cost of carbon is higher than under RE, unless $\Phi = 0$:

$$egin{array}{rcl} arpi_t &=& arpi_t^{RE} > \Lambda_t^s \ \Phi_t > 0, \ orall t, \ &=& \Lambda_t^s \ \Phi_t = 0; \end{array}$$

2. The carbon tax is higher than under RE, unless $\Phi = 0$:

$$\begin{split} \mathbb{E}_t \Xi^e_{t+j} &\geq \mathbb{E}_t \Xi^{e-RE}_{t+j} \geq 0, \ \Phi_t > 0 \ \forall t \\ &= 0, \ \Phi_t = 0; \end{split}$$

3. Ex ante capital tax rate is less than under RE, *unless* $\Phi = 0$ *:*

$$\begin{split} \mathbb{E}_t \Xi_{t+1}^k &\leq \mathbb{E}_t \Xi_{t+1}^{k-RE} \leq 0, \ \Phi_t > 0 \ \forall t \\ &= 0, \ \Phi_t = 0. \end{split}$$

This proposition highlights an intriguing point: having little faith in climate science, climate skeptics might naturally want to pay a lower carbon tax. Yet, with manifestly poetic justice, the very consequence of skepticism by itself is an increase in both the social cost of carbon and the carbon tax, and a decrease in the tax on capital.

The next two propositions provide results for two regimes in which the Ramsey planner faces a climate-skeptical public whose beliefs are not known and are, indeed unknowable. The first regime is political in that the planner believes the unknown beliefs of the private sector to be true. In the second regime, the planner is paternalistic in that it believes the science model to be true.

Proposition 3 (Political planner). *Ignorance of private beliefs believed to be true leads to the following policy alterations:*

1. The social cost of carbon contains an ambiguity premium for both Ramsey and social plans:

$$\begin{split} \varpi_t &= \varpi_t^{RE} + \bar{\Phi}f(\bar{\Phi}, 0) \Big[(1 - \gamma) cov(n^{PO}, \Lambda) + \gamma cov(n^{PO}, \Lambda \vartheta) \Big] \\ &+ f(\bar{\Phi}, 0) cov(n^{PO}, \Lambda) \ge \varpi_t^{RE}, \ \forall t, \\ &= \Lambda_t^s + cov(n^{PO}, \Lambda) \ge \Lambda_t^s, \ \Phi = 0; \end{split}$$

2. The carbon tax is higher than under RE, unless $\Phi = 0$:

$$\begin{split} \mathbb{E}_t \Xi_{t+j}^{e-PO} &\geq \mathbb{E}_t \Xi_{t+j}^{e-RE} \geq 0, \ \Phi_t > 0 \ \forall t, \\ &= 0, \ \Phi_t = 0; \end{split}$$

3. Ex ante capital tax is lower than under RE, *unless* $\Phi = 0$ *:*

$$\begin{split} \mathbb{E}_t \Xi_{t+1}^{k-PO} &\leq \mathbb{E}_t \Xi_{t+1}^{k-RE} \leq 0, \ \bar{\Phi} > 0 \ \forall t \\ &= 0, \ \bar{\Phi} = 0. \end{split}$$

The preceding proposition establishes that a political planner's ignorance about private beliefs, even if held to be correct, justifies a positive ambiguity premium for the social cost of carbon, activated by correlations between the planner's worst-case belief multiplier n^{PO} and the social cost of carbon that would apply if rational expectations prevailed. In addition, unless the government is a social planner, ambiguity raises the carbon tax and lowers the capital tax.

Proposition 4 (**Paternalistic planner**). *Ignorance of private beliefs that the planner also believes to be false leads to the following policy alterations:*

1. The social cost of carbon contains an ambiguity premium, unless $\Phi = 0$:

$$\begin{split} \varpi_t &= \varpi_t^{RE} + \Phi_t f(\Phi_t, 0) \left[(1 - \gamma) cov(n^{PA}, \Lambda) + \gamma cov(n^{PA}, \Lambda \vartheta) \right] \\ &\geq \varpi_t^{RE} \quad \Phi_t > 0, ; \forall t, \\ &= \Lambda_t^s, \quad \Phi_t = 0; \end{split}$$

2. The carbon tax contains an ambiguity premium, unless $\Phi = 0$:

$$\begin{split} \mathbb{E}_{t}\Xi_{t+j}^{e-PA} &\geq \mathbb{E}_{t}\Xi_{t+j}^{e-RE} + \gamma \Phi_{t}f(\Phi_{t},0)cov(n^{PA},\varsigma^{e}\vartheta) \geq \mathbb{E}_{t}\Xi_{t+1}^{e-RE} \\ &= \geq 0, \ \Phi_{t} > 0 \ \forall t, \\ &= 0, \ \Phi_{t} = 0; \end{split}$$

3. The ex ante capital tax contains an ambiguity subsidy, unless $\Phi = 0$:

$$\begin{split} \mathbb{E}_{t}\Xi_{t+1}^{k-PA} &\leq \mathbb{E}_{t}\Xi_{t+1}^{k-RE} - \Phi_{t}f(\Phi_{t},0)\Big[1 + \gamma cov(n^{PA},\zeta^{k}) + \gamma cov(n^{PA},\zeta^{k}\vartheta)\Big] \\ &\leq \mathbb{E}_{t}\Xi_{t+1}^{k-RE} \leq 0, \ \Phi_{t} > 0 \ \forall t, \\ &= 0, \ \Phi_{t} = 0. \end{split}$$

The main distinction between a political and a paternalistic planner, as defined in this paper, is that for the latter, an ambiguity premium for the social cost of carbon applies only if the government is a Ramsey planner and not a social planner. For both types of Ramsey planner, a positive ambiguity premium on the social cost of carbon is optimal because the planner's worst-case martingale belief distortion correlates with the certainty-equivalent version of the social cost of carbon.

The next proposition introduces the possibility of the government itself having doubts about the model.

Proposition 5 (**Pessimistic planner—skeptical consumer**). *Ignorance of private beliefs that the government believes to be false, in combination with a planner's doubts about the model, leads to the following policy alterations:*

1. The social cost of carbon contains an ambiguity premium in both Ramsey and social plans:

$$\begin{split} & \varphi_t \geq \varphi_t^{RE} + f(\Phi_t, 0) cov(n^p, \Lambda) \\ & + f(\Phi_t, 0) \Phi_t[(1 - \gamma) cov(m^p, \Lambda) + \gamma cov(m^p, \vartheta \Lambda)] \geq \varphi_t^{RE}, \ \Phi_t > 0 \\ & \geq \varphi_t^{RE} + cov(n^p, \Lambda) \geq \varphi_t^{RE}, \ \Phi_t = 0 \ \forall t. \end{split}$$

2. The carbon tax contains a positive or negative ambiguity premium, unless $\Phi = 0$:

$$\begin{split} \mathbb{E}_{t} \Xi_{t+j}^{e} &\geq \mathbb{E}_{t} \Xi_{t+j}^{e-RE} + (1-\gamma) \Phi_{t} f(\Phi_{t}, 0) \left[cov(m^{p}, \frac{1}{n^{p}}) + cov(\varsigma^{e}, \frac{m^{p}}{n^{p}}) \right] \\ &+ \gamma \Phi_{t} f(\Phi_{t}, 0) \left[cov(\vartheta, \varsigma^{e}) + cov(m^{p}, \frac{1}{n^{p}}) + cov(\varsigma^{e}\vartheta, \frac{m^{p}}{n^{p}}) \right] \\ &\stackrel{\geq}{\equiv} \Xi_{t+j}^{e-RE}, \quad \Phi_{t} > 0 \quad \forall t \\ &= 0, \quad \Phi_{t} = 0. \end{split}$$

3. The ex ante capital tax is a subsidy in both Ramsey and social plans:

$$\begin{split} \mathbb{E}_{t} \Xi_{t+1}^{k-R-s} &\leq \mathbb{E}_{t} \Xi_{t+1}^{k-RE} - f(\Phi_{t}, 0) [cov(\varsigma^{k}, n^{p}) + (1-\gamma)cov(\varsigma^{k}, m^{p})] \\ &- \gamma \Phi_{t} f(\Phi_{t}, 0) cov(\vartheta, \varsigma^{k} m^{p})] \leq \mathbb{E}_{t} \Xi_{t+1}^{k-RE} \leq 0, \ \Phi_{t} \geq 0, \\ &= -cov(\varsigma^{k}, n^{p}) - (1-\gamma)cov(\varsigma^{k}, m^{p}) \leq 0, \ \Phi_{t} = 0. \end{split}$$

Consistent with Propositions 3 and 4, the marginal contribution of an increased correlation between the government's ignorance of skeptical private beliefs $m^p \pi$ with the normalized discounted excess return to fossil energy over the social cost of carbon ζ^e or the discounted gross return to capital ζ^k , is to raise both the social cost of carbon and the carbon tax, and to lower the *ex ante tax* on capital. The marginal contribution of correlations involving the planner's own model doubts is to raise the social cost of carbon and to lower both the carbon tax and the *ex ante* tax on capital.

In the next and final proposition, the government remains pessimistic, but consumers are pessimistic rather than skeptical.

13.4. Heterogeneous Beliefs: Pessimistic Consumers

Proposition 6 (**Pessimistic planner—pessimistic consumer**). The combination of pessimism in the private sector and the government's pessimism has the following implications:

1. The social cost of carbon contains an ambiguity premium in both Ramsey and social plans:

$$\begin{split} \varpi_t &\geq [1 - f(\Phi_t, \varepsilon_t)\varepsilon_t] \varpi_t^{RE} + f(\Phi_t, \varepsilon_t)[\varepsilon_t + cov(n^p, \Lambda)] \\ &+ f(\Phi_t, \varepsilon_t) \Phi_t[(1 - \gamma)cov(m^c, \Lambda) + \gamma cov(m^c, \vartheta \Lambda)] \\ &\geq \varpi_t^{RE}, \ \Phi_t > 0, \\ &\geq \frac{1}{1 + \varepsilon_t}[\varpi_t^{RE} + \varepsilon_t + cov(n^p, \Lambda)] > \Lambda_t^s \ \Phi_t = 0, \ \forall t. \end{split}$$

2. If $\Phi > 0$, the premium on the carbon tax may be positive or negative, and is negative otherwise:

$$\begin{split} \mathbb{E}_{t} \Xi_{t+j}^{e} &\geq \mathbb{E}_{t} \Xi_{t+j}^{e-RE} \\ &- f(\Phi_{t}, \varepsilon_{t}) f(\Phi_{t}, 0) [1 + \Phi_{t} (1 - \gamma + \gamma \mathbb{E}_{t} \vartheta_{t+1} + \gamma cov(\varsigma^{e}, \vartheta))] \varepsilon_{t} \\ &+ f(\Phi_{t}, \varepsilon_{t}) \left[\varepsilon_{t} + cov(\varsigma^{e}, \frac{1}{m^{c}}) + \Phi_{t} [(1 - \gamma) cov(\varsigma^{e}, \frac{1}{n^{p}}) + \gamma cov(\vartheta \varsigma^{e}, \frac{1}{n^{p}})] \right] \\ & \lessapprox \mathbb{E}_{t} \Xi_{t+j}^{e-RE}, \ \Phi_{t} > 0, \ \forall t \\ &= \frac{1}{1 + \varepsilon_{t}} cov(\varsigma^{e}, \frac{1}{m^{c}}) < 0, \ \Phi_{t} = 0. \ \forall t \end{split}$$

3. The ex ante capital tax rate may be positive or negative:

$$\begin{split} \mathbb{E}_{t} \Xi_{t+1}^{k-R-p} &\leq \mathbb{E}_{t} \Xi_{t+1}^{k-RE} \\ &- f(\Phi_{t}, \varepsilon_{t}) f(\Phi_{t}, 0) \left[1 + \Phi_{t} \left(1 - \gamma + \gamma [\mathbb{E}_{t} \vartheta_{t+1} + cov(\varsigma^{k}, \vartheta)] \right) \right] \varepsilon_{t} \\ &- f(\Phi_{t}, \varepsilon_{t}) \left[cov(\varsigma^{k}, \frac{n^{p}}{m^{c}}) + cov(\frac{1}{m^{c}}, n^{p}) + (1 + \varepsilon_{t}) cov(\varsigma^{k}, n^{p}) \right] \\ &\lessapprox \mathbb{E}_{t} \Xi_{t+1}^{k-RE}, \ \Phi_{t} > 0 \\ &= -\frac{1}{1 + \varepsilon_{t}} \left[cov(\varsigma^{k}, \frac{n^{p}}{m^{c}}) + cov(\frac{1}{m^{c}}, n^{p}) \right] \\ &- cov(\varsigma^{k}, n^{p}) \lessapprox 0, \ \Phi_{t} = 0. \end{split}$$

As in Proposition 5, which concerned skeptical beliefs, in this belief regime, the marginal effect of private-sector pessimism m^c is to raise the social cost of carbon. However, the effect on carbon and capital taxation is the opposite, producing a reduction in the carbon tax and an increase in the tax on capital. The intuition is that increased consumer doubts about the climate model tend to motivate carbon consumption below the socially optimal level and to increase capital spending above its socially optimal level. The net effect of an increase in ϖ_t and a decrease in $\mathbb{E}_t \Xi_{t+j}^e$ may or may not in the end produce a lower carbon tax itself, because from (95) is the sum of two terms: $\varpi_t + \chi_t$, where χ is the sum of terms that contain expected future values of Ξ_{t+j}^e .

Proposition 6 echoes Proposition 5 in that the marginal contribution of any correlation between the planner's own model doubts n^p and asset returns represented by ζ^k and ζ^e is to likewise raise the social cost of carbon and to lower both the carbon tax and the *ex ante* tax on capital. This becomes apparent if the role of private beliefs is de-activated by setting $m^p \equiv 1$ or $m^c \equiv 1$, respectively, leading to

+
$$f(\Phi_t, 0)\Phi_t\left[(1-\gamma)cov(\varsigma^e, \frac{1}{n^p}) + \gamma cov(\vartheta\varsigma^e, \frac{1}{n^p})\right],$$
 (116)

$$\mathbb{E}_t \Xi_{t+1}^{k-R-s} \leq \mathbb{E}_t \Xi_{t+1}^{k-RE} - f(\Phi_t, 0) cov(\varsigma^k, n^p).$$

$$(117)$$

The belief regimes in the three preceding formulas, representing policies stripped of any effects due to belief distortions in the private sector—skeptical or pessimistic—are most closely related to the extant literature on robust climate policy and so serve best for comparisons, to which I now turn.

In a policy regime most similar to that treated in Proposition 5, with m^p set equal to 1 but differing in some details, Hennlock (2009) attributes doubts about the model not to a policy authority *per se* but to a utility-maximizing representative consumer representing society who computes robust feedback rules that, as in (115), generate an ambiguity

premium on the expected social cost of carbon, so that with nonlinear damage, policy becomes more responsive to changes in climate. Li et al. (2016) study a dynamic optimization problem that is nearly identical to the planning models in Section 7, but under the assumption that the government is a social planner ($\Phi = 0$ in the present paper) and not a Ramsey planner. They find that even a relatively small increase in the concern about model uncertainty can cause a significant drop in optimal energy extraction and a rise in the socially optimal carbon tax, which in this paper would also correspond to the result in (115). Likewise, Cai and Lontzek (2019) find that with empirically plausible parameterizations of Epstein-Zin preferences to represent attitudes towards risk, the uncertainty associated with anthropogenic climate change implies carbon taxes much higher than associated with deterministic models, while Lemoine and Traeger (2016) conclude that a government's aversion to Knightian uncertainty in the face of an ambiguous tipping point increases the optimal tax on carbon dioxide emissions, but only by a small amount. As with Li et al. (2016), since the derived tax effects come via changes in the SCC, those two conclusions also best correspond to formula (115). Heal and Millner (2013) do not consider taxation per se, but find that the value of abatement (that would presumably include carbon taxation) rises as ambiguity aversion increases. Rezai and van der Ploeg (2017) take a somewhat different approach to modeling ambiguity and consider a so-called agnostic policy authority—essentially a government having ambiguity about the approximating climate model—as facing potential models ranging from denialist to scientific. If such a government pursues max-max policies, it imposes higher carbon taxes as a precaution. Later, with the help of Nordhaus's (1993) DICE model to simulate carbon taxation, Rezai and van der Ploeg (2019) broaden their earlier results and consider an agnostic planner who adopts Pascal's Wager (Pascal's (1670))³⁷ with the question: what would such an agnostic but rational planner—one who does not know or care which model is correct but who wishes to avoid the worst-do when faced with some probability that the approximating model, adhered to by so-called deniers, is false? Their conclusion is that the *hedge-your-bet* optimal carbon tax is quite close to the optimal tax derived in a non-denialist scientific setting, even if the probability of the model being false is a mere ten percent. Further, when ambiguity about whether scientists or deniers are correct rises, as represented by a parameter of constant relative ambiguity aversion, the optimal carbon pricing policy moves ever closer to the science-based policy.

Finally, Anderson et al. (2013), who study a model similar in spirit to the models in this paper, except for an additional robustness channel capable of affecting growth, find that a planner's increased deep uncertainty about the model can result in either a decrease or an increase in the optimal carbon tax, depending on other factors, such as market features and social preferences. In the present paper, the conclusions are driven by similar forces involving preferences and market features, but in the form of second-order moments represented by the covariances between worst-case martingale belief multipliers *m* and relevant market features, such as net asset-returns to fossil energy and capital.

Table 2 summarizes the preceding propositions. A principal leitmotif of this paper is belief heterogeneity, a main driver of ambiguities in all regimes studied here. Even if (as a mental, albeit unrealistic exercise in Proposition 2) one were to assume away ambiguity, the very presence of heterogeneity in beliefs between the private sector and the government, leading to a positive spread between their respective discount factors, is sufficient to alter the policies of a Ramsey planner, though not those of a social planner. As a consequence, the government increases its estimate of the social cost of carbon and adds a premium to the carbon tax, while raising the capital subsidy rate.

A government with its own doubts about the model is compelled to raise the social cost of carbon and acquires further motives to either raise or lower taxes, depending on the case. However, in all instances, the planner will raise the social cost of carbon. The political and paternalistic planners of Propositions 3 and 4 face ambiguity because of their ignorance of arbitrary private beliefs held by consumers who regard the scientific model with skepticism and discount the future at relatively higher rates. They therefore use

relatively more fossil energy and invest in relatively less capital than is socially optimal. To encourage socially optimal choices, the government taxes carbon and subsidizes capital.

The pessimistic planners in Propositions 5 and 6 are identical except with respect to the kind of beliefs they face. The former operates under ambiguities that result from doubts about the model itself and from its ignorance of private beliefs. The latter government confronts a single ambiguity caused by its own doubts about the model, but being constrained by consumers who have known pessimistic beliefs, this government effectively manages two kinds of ambiguities, its own and those of the public. In both cases, the government's doubts about the model, indexed by n^p , provide motives to raise the social cost of carbon and the capital subsidy but to lower the carbon tax. A pessimistic government's ignorance of the beliefs of skeptical consumers in Proposition 5 motivates policies that mimic those of the political and paternalistic planners in Propositions 3 and 4.

Pessimistic consumers discount the future less than do skeptical or rational consumers and therefore use less carbon energy and invest more in capital relative to socially optimal rates and quantities. So the presence of private-sector pessimism is an inducement for the government to lower carbon taxes and to raise the capital tax. Given that the pessimistic government's own doubts produce motives in the opposite direction, the net effect of private and government doubts can be ambiguous.

	Φ	ω_t	χ_t	$ar{ au}_t^k$
I. Homogeneous Beliefs (RE)				
Ramsey	>0	$\geq \Lambda^s$	$\chi^{RE} \ge 0$	$ au^{k-RE} < 0$
Social	0	Λ^s	0	0
II. Heterogeneous Beliefs				
1. Skeptical consumers				
Ambiguity absent				
Ramsey	>0	$\geq \Lambda^s$	$\geq \chi^{RE}$	$\leq \tau^{k-RE}$
Social	0	Λ^s	≥ 0	≤ 0
Ambiguity present				
Political planner				
Ramsey	>0	$\geq \Lambda^s$	$\geq \chi^{RE}$	$\leq \tau^{k-RE}$
Social	0	$\geq \Lambda^s$	≥ 0	≤ 0
Paternalistic planner				
Ramsey	>0	$\geq \Lambda^s$	$\geq \chi^{RE}_t$	$\leq \tau^{k-RE}$
Social	0	$\geq \Lambda^s$	≥ 0	≤ 0
Pessimistic planner				
Ramsey	>0	$\geq \Lambda^s$	$\gtrless \chi^{RE}$	$\leq \tau^{k-RE}$
Social	0	$\geq \Lambda^s$	$\geq \chi^{RE}$	$\leq \tau^{k-RE}$
Effect of planner's ambiguity				
Ramsey	>0	\uparrow	\uparrow	\downarrow
Social	0	1	\uparrow	\downarrow
Effect of consumer's ambiguity				
Ramsey	>0	\uparrow	\downarrow	\downarrow
Social	0	\uparrow	\downarrow	\downarrow
2. Pessimistic consumers				
Pessimistic planner				
Ramsey	>0	$\geq \Lambda^s$	$\gtrless \chi^{RE}$	$\leq \tau^{k-RE}$
Social	0	$\geq \Lambda^s$	$\geq \chi_t^{RE}$	$\underset{\tau^{k-RE}}{\overset{\leqslant}{\underset{\tau^{k-RE}}{\overset{\ast}{\underset{\tau^{k-RE}}{\underset{\tau^{k-RE}}{\overset{\ast}{\underset{\tau^{k-RE}}{\overset{\ast}{\underset{\tau^{k-RE}}{\underset{\tau^{k-RE}}{\overset{\ast}{\underset{\tau^{k-RE}}{\underset{\tau^{k-RE}}{\underset{\tau^{k-RE}}{\underset{\tau^{k-RE}}{\underset{\tau^{k-RE}}{\underset{\tau^{k-RE}}{\underset{\tau^{k-RE}}{\underset{\tau^{k-RE}}{\underset{\tau^{k-RE}}{\underset{\tau^{k-RE}}{\underset{\tau^{k-RE}}{\underset{\tau^{k-RE}}{\underset{\tau^{k-RE}}{\underset{\tau^{k-RE}}{\underset{\tau^{k-RE}}{\underset{\tau^{k-RE}}{\underset{t^{k-RE}}}{\underset{t^{k-RE}}{\underset{t^{k-RE}}{\underset{t^{k-RE}}}{\underset{t^{k-RE}}}{\underset{t^{k-RE}}{\underset{t^{k-RE}}{\underset{t^{k-RE}}}{\underset{t^{k-RE}}}{\underset{t^{k-RE}}}{\underset{t^{k-RE}}}{\underset{t^{k-RE}}}{\underset{t^{k-RE}}}{\underset{t^{k-RE}}}{\underset{t^{k-RE}}}{\underset{t^{k-RE}}}{\underset{t^{k-RE}}}{\underset{t^{k-RE}}}{\underset{t^{k-RE}}}{\underset{t^{k-RE}}}}}}}}}}}}}}}}}}}}}}}}}}}}}}}}}}}}$
Effect of planner's ambiguity				
Ramsey	>0	\uparrow	$\uparrow\downarrow$	$\uparrow\downarrow$
Social	0	\uparrow	$\uparrow\downarrow$	$\uparrow\downarrow$
Effect of consumer's ambiguity				
Ramsey	>0	\uparrow	$\uparrow\downarrow$	$\uparrow\downarrow$
Social	0	NA	\downarrow	$\uparrow\downarrow$

Table 2. Propositions 1–6.

14. A Feedback from Taxes to Consumers' Pessimistic Beliefs

When faced with pessimistic consumers, the policies of a Ramsey planner ($\Phi > 0$) present an instance of a possible two-way feedback between carbon tax policy and privatesector pessimism via the debt channel.³⁸ For example, consider a positive surprise in debt $b_t > \mathbb{E}_{t-1}m_t^c b_t$. From (77), the shadow value to the taxing authority of the consumer's utility ε_t must drop, implying a drop in the consumer's worst-case utility U_t , hence an increased pessimism via (6).

The tax implications follow from formulas (79), (86), (87), (95), (100) and (112), which show that the social cost of carbon and the carbon tax are decreasing functions of ε_t , and that the *ex ante tax* on capital is an increasing function of ε_t , so that a *t*-period surprise increase in debt that causes a drop in ε_t and an increase in consumer pessimism should be associated with an increase in the SCC and the carbon tax, and a drop in the tax on capital.

The taxing authority is motivated by two considerations. On the one hand, for purely fiscal reasons, it wants to raise taxes in those states against which it is cheaper to issue debt. It therefore raises tax rates in high-debt states caused by climate shocks, and conversely lowers taxes in low-debt states when climate is calmer. The government also has a goal of setting the prices of carbon and capital in ways that are optimal for society. It turns out that these twin goals coincide: by manipulating debt and taxes to raise (or reduce) the household's utility and turning it more (or less) pessimistic and less (or more) willing to use fossil energy but more (or less) willing to spend on capital, the government manages both, an optimal allocation and a need to smooth debt and taxes over time.

15. Conclusions

This paper is about the climate policy implications of belief heterogeneity and ambiguity in a dynamic market economy governed by a benign welfare maximizing authority, here referred to as a Ramsey planner. To keep this paper focused, many important features relevant for practical policy, such as non-linearities in the mechanisms from the burning of fossil fuels to climate change, are set aside, as are some macroeconomic issues, such as the implications of either exogenous or directed endogenous technological change on growth.³⁹ Nor do I account for substitution among available types of fossil energy inputs, such as oil, gas, and coal, and green energy from sun, water, and even nuclear power. Finally, following many examples in the literature, to keep the presentation manageable, all sources of uncertainty have been combined into a single "climate-cost shock" variable representing all CO_2 -related economic climate damages, including those related to the dynamics of CO_2 , productivity shocks other than from climate, and cost shocks from alternative sources of energy, including renewable energy.

This paper, has, in the main, sought to adhere to the spirit of its antecedents, including Stern (2007), Nordhaus (2008), Acemoglu et al. (2012), von Below (2012), van der Ploeg and Withagen (2014), Golosov et al. (2014), Belfiori (2017), Rezai and van der Ploeg (2017), Barrage (2018), and Cai and Lontzek (2019). Golosov et al. (2014), whose model I consider a benchmark for comparisons, proved that the optimal carbon tax (expressed as a proportion of GDP) depends solely on the social cost of carbon, the carbon persistence parameter, and a discount factor. While this is true here as well when beliefs are homogeneous and rational, this paper's value added is an accounting of how belief distortions about the underlying model will alter that discount factor, depending on type of ambiguity.

The principal insight in this paper is that dissonance in beliefs, expressed as skepticism or doubt about forces governing economic outcomes arising from climate cost shocks, produce ambiguity and create wedges between the private sector's discount factor and the government's discount factor and therefore between the way in which consumers and society differentially price two assets with uncertain pay-offs: one from physical capital, whose returns are positive, and the other from atmospheric carbon accumulation, whose returns are negative.

Ambiguity that is due to the government's ignorance of optimistically distorted private beliefs raises the expected social cost of carbon and the carbon tax, which is offset by a

lowering of capital taxes. This conclusion carries a certain political irony: a public that confidently denies or minimizes the fact of climate change may actually see its carbon taxes increased.

Ambiguity that arises as a consequence of neither government nor private sector trusting the scientific model has mixed effects depending on the relative degree of pessimism in either sector. So, while any doubts held by either the planner or the private sector produce an ambiguity premium to the social cost of carbon, their differential tax effects are mixed. A combination of government doubt about the model and consumer skepticism motivates lowering the tax on capital, but has ambiguous effects on any carbon tax premium. The combination of fear of model misspecification on the part of both the government and the private sector justifies an ambiguity premium for the social cost of carbon, but opposing effects of some key correlations in the economy leave the net effect on carbon tax premiums ambiguous. The source of this disparity of outcomes derives from two channels: (i) an indirect positive effect through the social cost of carbon and (ii) a direct positive *or* negative effect on the tax itself, depending on the source of ambiguity.

In all cases discussed in this paper, the perspective of Arrow–Debreu asset pricing theory illuminates an equivalence between Pigouvian carbon taxation and optimal pricing of carbon permits in a cap-and-trade economy. This correspondence extends to conditions of Knightian uncertainty, when derived asset prices must meet the test of robustness.

Being theoretic, this paper begs the question of just how applicable the findings herein might be to the real world. In the absence of counter-factual history, the best approach is stochastic simulations of a properly calibrated DSGIE climate-economy model governed by alternative Ramsey regimes discussed here. Alternatively, solving Isaacs-Bellman-Flemming equations associated with the belief regimes in Section 7 as Hennlock (2009) and Li et al. (2016) have done, also holds promise.

Finally, a word of caution. This paper is normative in that it posits not what is but what should be, based on the ideal of welfare maximization implemented by a benign authority heedful of consumers' preferences and budgets. In the reality, such a Ramsey planner may be mere fiction, even in nominal democracies, such as the United States, where an overriding authority, the Supreme Court, has recently ruled that the Executive has no authority to implement carbon policies without explicit and detailed instructions by a legislature that has shown little inclination to address the approaching climate catastrophe. In the real world, policy regimes may even turn rogue: in 2017, the Envronmental Protection Agency under the previous US Administration scrubbed its website of all references to climate. The EPA's website EPA (2017), cited earlier, is available only because it was copied and preserved on another website.

Funding: This research did not receive any specific grant from funding agencies in the public, commercial, or not-for-profit sectors.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

Acknowledgments: A previous version of the paper, entitled "Confronting Climate Skepticism or Fear: Ramsey Carbon Taxation and Fiscal Policy under Ambiguity", was presented at the Troxler Symposium, "Economic Analysis, Environmental Uncertainties, and Policy Implications" at East Carolina University, 8 November 2019. I would like to thank all participants for comments and insights.

Conflicts of Interest: The author declares no conflict of interest.

40 of 56

Appendix A. The Household's Intertemporal Budget Constraint

Solve (4) forward for b_t , starting with t = 0:

$$b_{t} \geq \sum_{t=0}^{T-1} \left(\prod_{j=0}^{t-1} \sum_{x^{j+1}} \hat{p}_{j+1}(x_{j+1}) \right) [c_{t}(x^{t}) - H_{t} - g_{t}] \\ + \left(\prod_{j=0}^{T-1} \sum_{x^{j+1}} \hat{p}_{j+1}x_{j+1,0} \right) \left[\sum_{x^{T-1}} k_{T} + \sum_{x^{T}} \hat{p}_{T+1}x_{T+1}b_{T+1} \right] \\ + \left(\prod_{j=0}^{T-1} \sum_{x^{j+1}} \hat{p}_{j+1}x_{j+1} \right) \sum_{x^{T}} [p_{T}^{e} - \tau_{T}^{e}] Q_{T-1}(x^{T}) \\ + \sum_{t=1}^{T-1} \left(\prod_{j=0}^{t-1} \sum_{x^{j+1}} \hat{p}_{j+1}x_{j+1} \right) \left[1 - \sum_{x^{t}} \hat{p}_{t+1}x_{t+1}R_{t}^{k} \right] k_{t} \\ + \sum_{t=1}^{T-1} \left(\prod_{j=0}^{t-1} \sum_{x^{j+1}} \hat{p}_{j+1}x_{j+1} \right) \left[1 - \sum_{x^{t}} \hat{p}_{t+1}x_{t+1}R_{t}^{k} \right] [p_{t}^{e} - \tau_{t}^{e}] Q_{t},$$
 (A1)

where $p_0 = 1$, and Hotelling's rule (31) $\left(\frac{p_{t+1}^{e} - \tau_{t+1}^{e}}{p_{t}^{e} - \tau_{t}^{e}} = R_{t+1}^{k}\right)$ is used in the last term. By the no-arbitrage condition (14), the last two lines are zero. The third line vanishes under the assumption of resource exhaustibility, where $\lim_{T\to\infty} Q_{T-1} = 0$. The second line disappears as a consequence of the household's no-Ponzi game condition

$$\lim_{T \to \infty} \sum_{x^T} \hat{q}_T(x^T) [k_{T+1}(x^T) + \sum_{x_T} \hat{p}_{T+1}(x_{T+1}|x^T) b_{T+1}(x^T)] = 0.$$
(A2)

This leaves the first line to be evaluated. Note, from (17), the t-step ahead pricing kernel,

$$\begin{aligned} \hat{q}_{t}(x^{t}, x_{0}) &\equiv \Pi_{j=0}^{t-1} \hat{p}_{j+1,0}(x^{j}) \\ &= \Pi_{j=0}^{t-1} \beta m_{j+1}(x^{j+1}) \pi_{j}(x^{j}) \frac{u_{c_{j+1}}(x^{j+1})}{u_{c_{j}}(x^{j})} \\ &= \beta^{t} M_{t}(x^{t}) \pi_{t}(x^{t}) \frac{u_{c_{t}}(x^{t})}{u_{c_{0}}(x^{0})}, \ q_{0} = 1, M_{0} = 1, \end{aligned}$$
(A3)

is the price of an Arrow–Debreu contract written at t = 0. Substituting the last line in (A3) into the term in parentheses in the first line of (A1) then yields an expression for the household's intertemporal budget constraint shown in (35).

Appendix B. Derivation of Implementability Constraint (37)

From (36), the expectation of W_{t+1} with respect to the distorted measure $m_{t+1}\pi_{t+1}$, valued at the price of an Arrow–Debreu security *b*, is

$$\begin{split} &\sum_{x_{t+1}} \hat{p}_{t+1}(x_{t+1}|x^t) \mathcal{W}_{t+1}(x^{t+1}) = \\ &\sum_{x_{t+1}} \hat{p}_{t+1}(x_{t+1}|x^t) [b_{t+1}(x^{t+1}) + R_{t+1}^k(x^{t+1})k_{t+1}(x^t)] \\ &+ \sum_{x_{t+1}} \hat{p}_{t+1}(x_{t+1}|x^t) (p_{t+1}^e(x^{t+1}) - \tau_{t+1}^e(x^{t+1})) Q_{t+1}(x^t) \\ &= \sum_{x_{t+1}} \hat{p}_{t+1}(x_{t+1}|x^t) [b_{t+1}(x^{t+1}) + R_{t+1}^k(x^{t+1})k_{t+1}(x^t)] \\ &+ [p_t^e(x^t) - \tau_t^e(x^t)] \sum_{x_{t+1}} \hat{p}_{t+1}(x_{t+1}|x^t) R_{t+1}^k(x^{t+1}) Q_{t+1}(x^t), \end{split}$$

where (25) and Hotelling's rule (31) are used to get the last line. Application of no-arbitrage condition (14) further simplifies the expression for household wealth:

$$\sum_{x_{t+1}} \hat{p}_{t+1}(x_{t+1}|x^t) \mathcal{W}_{t+1}(x^{t+1}) = \sum_{x_{t+1}} \left[\hat{p}_{t+1}(x_{t+1}|x^t) b_{t+1}(x^{t+1}) + k_{t+1}(x^t) \right] + \left[p_t^e(x^t) - \tau_t^e(x^t) \right] Q_{t+1}(x^t),$$
(A4)

indicating that the consumer's expected wealth in the following period includes the unused store of fossil fuel valued at its current after-tax price in addition to the capital stock and bonds carried over into the next period. Using the household's period budget constraint (4) and (A4), household wealth defined in (36) is, equivalently,

$$\mathcal{W}_{t}(x^{t}) = b_{t}(x^{t}) + R_{t}^{k}k_{t}(x^{t-1}) + (p_{t}^{e}(x^{t}) - \tau_{t}^{e}(x^{t}))Q_{t}(x^{t-1})
= c_{t}(x^{t}) - H_{t}(x^{t}) - g_{t}(x^{t}) + k_{t+1}(x^{t})
+ \sum_{x^{t+1}} \left[\hat{p}_{t+1}(x_{t+1}|x^{t})b_{t+1}(x^{t+1}) + (p_{t}^{e}(x^{t}) - \tau_{t}^{e}(x^{t}))Q_{t+1}(x^{t}) \right]
\geq c_{t}(x^{t}) - H_{t}(x^{t}) - g_{t}(x^{t}) + \sum_{x_{t+1}} \hat{p}_{t+1}(x_{t+1}|x^{t})\mathcal{W}_{t+1}(x^{t+1})
= c_{t}(x^{t}) - H_{t}(x^{t}) - g_{t}(x^{t}) + \sum_{x_{t+1}} \hat{\varrho}_{t+1}\pi_{t+1}\mathcal{W}_{t+1}(x^{t+1})
= c_{t}(x^{t}) - H_{t}(x^{t}) - g_{t}(x^{t}) + \beta \sum_{x_{t+1}} m_{t+1}\pi_{t+1}\frac{u_{c_{t+1}}}{u_{c_{t}}}\mathcal{W}_{t+1}(x^{t+1}). \quad (A5)$$

Appendix C. The Social Cost of Carbon

This appendix evaluates the expectation (83) for the four belief regimes:

$$\omega_t = f(\Phi_t, \varepsilon_t) \mathbb{E}_t \left[n_{t+1}^{**} \Lambda_{t+1} \left(1 + \varepsilon_{t+1} + [1 - \gamma + \gamma \vartheta_{t+1}] n_{t+1}^* \Phi_t \right) \right],$$

where $\Lambda_{t+1} = \varrho_{t+1}[(1-\rho)\varpi_{t+1} + x_{t+1}Y_{t+1}], \mathbb{E}_t\Lambda_{t+1} = \Lambda_t^s, f(\Phi_t, \varepsilon_t) = \frac{1}{1+\varepsilon_t+[1-\gamma+\gamma\vartheta_t]\Phi_t}$ $\leq f(\Phi_t, 0) < f(0, 0) = 1,$ $\vartheta_t = \frac{H_t + g_t}{c_t} \forall t, \text{ and } \varepsilon_{t+1} > 0 \text{ only if consumers are pessimistic.}^{40}$ A. Homogeneous beliefs

RE solution $(n^* = 1; n^{**} = 1, \epsilon = 0, \Phi_t = \bar{\Phi})$

$$\begin{split} \varpi_{t}^{RE} &= f(\bar{\Phi}, 0) \mathbb{E}_{t} [\Lambda_{t+1} (1 + [1 - \gamma + \gamma \vartheta_{t+1}] \bar{\Phi})] \\ &= f(\bar{\Phi}, 0) \mathbb{E}_{t} [1 + (1 - \gamma) \bar{\Phi}] \mathbb{E}_{t} \Lambda_{t+1} + \gamma \bar{\Phi} f(\bar{\Phi}, 0) \mathbb{E}_{t} \Lambda_{t+1} \vartheta_{t+1} \\ &= f(\bar{\Phi}, 0) [(1 + \bar{\Phi}[1 - \gamma + \gamma \mathbb{E}_{t} \vartheta_{t+1}]) \Lambda_{t}^{s} + \bar{\Phi} \gamma cov(\Lambda, \vartheta)], \\ &= \Lambda_{t}^{s} + \gamma \bar{\Phi} f(\bar{\Phi}, 0) [(\mathbb{E}_{t} \vartheta_{t+1} - \vartheta_{t}) \Lambda_{t}^{s} + cov(\Lambda, \vartheta)], \quad \bar{\Phi} > 0, \; \forall t, \\ &= \Lambda_{t}^{s}, \; \bar{\Phi} = 0 \; \forall t. \end{split}$$
(A6)

B. Heterogeneous beliefs

1. **Skeptical private sector**

пг

Planner has no ambiguity about distorted private beliefs m^s ($n^* = m^s$, $n^{**} =$ (a) 1, $\epsilon = 0$)

$$\begin{split} \omega_t^o &= f(\Phi_t, 0) \mathbb{E}_t \left[\Lambda_{t+1} \left(1 + \left[(1 - \gamma) + \gamma \vartheta_{t+1} \right] m_{t+1}^o \Phi_t \right) \right] \\ &= f(\Phi_t, 0) \mathbb{E}_t \Lambda_{t+1} + (1 - \gamma) \Phi_t f(\Phi_t, 0) \mathbb{E}_t \Lambda_{t+1} + \gamma \Phi_t f(\Phi_t, 0) \mathbb{E}_t \Lambda_{t+1} \vartheta_{t+1} \\ &= f(\Phi_t, 0) \left[(1 + \Phi_t (1 - \gamma + \gamma \mathbb{E}_t \vartheta_{t+1}) \Lambda_t^s + \gamma \Phi_t cov(\Lambda, \vartheta) \right] \\ &= \omega_t^{RE}, \ \Phi_t > 0 \\ &= \Lambda_t^s, \ \Phi_t = 0 \ \forall t. \end{split}$$
(A7)

Political planner $(n^* = 1; n^{**} = n^{PO}, \varepsilon = 0)$ (b)

$$\begin{split} \varpi_t &= f(\bar{\Phi}, 0) \mathbb{E}_t \left[n_{t+1}^{PO} \Lambda_{t+1} (1 + [1 - \gamma + \gamma \vartheta_{t+1}] \bar{\Phi}) \right] \\ &= f(\bar{\Phi}, 0) (1 + (1 - \gamma) \bar{\Phi}) \mathbb{E}_t n_{t+1}^{PO} \Lambda_{t+1} + \gamma \bar{\Phi} f(\bar{\Phi}, 0) \mathbb{E}_t n_{t+1}^{PO} \Lambda_{t+1} \vartheta_{t+1} \\ &= f(\Phi, 0) [(1 + \Phi [1 - \gamma + \gamma \mathbb{E}_t \vartheta_{t+1}]) \Lambda_t^s + \gamma \Phi cov(\Lambda, \vartheta)] + f(\Phi, 0) cov(n^{PO}, \Lambda) \\ &+ \bar{\Phi} f(\bar{\Phi}, 0) \left[(1 - \gamma) cov(n^{PO}, \Lambda) + \gamma cov(n^{PO}, \Lambda \vartheta) \right] \\ &= \varpi_t^{RE} + \Phi f(\Phi, 0) \left[(1 - \gamma) cov(n^{PO}, \Lambda) + \gamma cov(n^{PO}, \Lambda \vartheta) \right] \\ &+ f(\bar{\Phi}, 0) cov(n^{PO}, \Lambda) \ge \varpi_t^{RE}, \bar{\Phi} > 0, \ \forall t \\ &= \Lambda_t^s + cov(n^{PO}, \Lambda) \ge \Lambda_t^s, \ \bar{\Phi} = 0. \end{split}$$
(A8)
$$(c) \quad \text{Paternalistic planner} \ (n^* = n^{PA}, n^{**} = 1, \varepsilon = 0) \\ \\ \varpi_t &= f(\Phi_t, 0) \mathbb{E}_t \left[\Lambda_{t+1} \left(1 + [1 - \gamma + \gamma \vartheta_{t+1}] n_{t+1}^{PA} \Phi_t \right) \right] \\ &= f(\Phi_t, 0) (\Lambda_t^s + (1 - \gamma) \Phi_t \mathbb{E}_t n_{t+1}^{PA} \Lambda_{t+1} + \gamma \Phi_t f(\Phi_t, 0) \mathbb{E}_t n_{t+1}^{PA} \Lambda_{t+1} \vartheta_{t+1} \end{split}$$

$$= f(\Phi_{t}, 0)(\Lambda_{t}^{*} + (1 - \gamma)\Phi_{t}\mathbb{E}_{t}n_{t+1}^{*}\Lambda_{t+1} + \gamma\Phi_{t}f(\Phi_{t}, 0)\mathbb{E}_{t}n_{t+1}^{*}\Lambda_{t+1}\vartheta_{t+1}$$

$$= f(\Phi_{t}, 0)[(1 + \Phi_{t}[1 - \gamma + \gamma\mathbb{E}_{t}\vartheta_{t+1}])\Lambda_{t}^{s} + \gamma\Phi_{t}cov(\Lambda, \vartheta)] + f(\Phi_{t}, 0)cov(n^{PA}, \Lambda)$$

$$+ \Phi_{t}f(\Phi_{t}, 0)\Big[(1 - \gamma)cov(n^{PA}, \Lambda) + \gamma cov(n^{PA}, \Lambda\vartheta)\Big]$$

$$= \omega_{t}^{RE} + \Phi_{t}f(\Phi_{t}, 0)\Big[(1 - \gamma)cov(n^{PA}, \Lambda) + \gamma cov(n^{PA}, \Lambda\vartheta)\Big] \ge \omega_{t}^{RE}, \ \Phi_{t} > 0, \ \forall t$$

$$= \Lambda_{t}^{s}, \ \Phi_{t} = 0 \ \forall t.$$
(A9)

(d) **Pessimistic planner**
$$(n^* = \frac{m^p}{n^p}, n^{**} = n^p, \varepsilon = 0.$$

$$\begin{split} \omega_t &= f(\Phi_t, 0) \mathbb{E}_t \left[n_{t+1}^p \Lambda_{t+1} \left(1 + [1 - \gamma + \gamma \vartheta_{t+1}] \frac{m_{t+1}^p}{n_{t+1}^p} \Phi_t \right) \right] \\ &= f(\Phi_t, 0) [\Lambda_t^s + cov(n^p, \Lambda)] + (1 - \gamma) \Phi_t f(\Phi_t, 0) [\Lambda_t^s + cov(m^p, \Lambda)] \\ &+ \gamma \Phi_t f(\Phi_t, 0) [\Lambda_t^s \mathbb{E}_t \vartheta_{t+1} + cov(\Lambda, \vartheta) + cov(m^p, \vartheta\Lambda)] \\ &= f(\Phi_t, 0) [(1 + \Phi_t [1 - \gamma + \gamma \mathbb{E}_t \vartheta_{t+1})]) \Lambda_t^s + \gamma \Phi_t cov(\Lambda, \vartheta)) + f(\Phi_t, 0) cov(n^p, \Lambda) \\ &+ f(\Phi_t, 0) \Phi_t [(1 - \gamma) cov(m^p, \Lambda) + \gamma cov(m^p, \vartheta\Lambda)]. \\ &= \omega_t^{RE} + f(\Phi_t, 0) [cov(n^p, \Lambda) + \Phi_t [(1 - \gamma) cov(m^p, \Lambda) + \gamma cov(m^p, \vartheta\Lambda)], \Phi_t > 0, \forall t, \\ &= \Lambda_t^s + cov(n^p, \Lambda) \ge \Lambda_t^s, \Phi_t = 0, \forall t. \end{split}$$
(A10)

2. **Pessimistic consumers**

(a) **Pessimistic planner** $(n^* = \frac{m^c}{n^p}, n^{**} = n^p, \varepsilon \neq 0)^{41}$

$$\begin{split} \omega_t &= f(\Phi_t, \varepsilon_t) \mathbb{E}_t \left[n_{t+1}^p \Lambda_{t+1} \left(1 + \varepsilon_{t+1} + [1 - \gamma + \gamma \vartheta_{t+1}] \frac{m_{t+1}^c}{n_{t+1}^p} \Phi_t \right) \right] \\ &= f(\Phi_t, \varepsilon_t) [\mathbb{E}_t n_{t+1}^p \Lambda_{t+1} + \mathbb{E}_t n_{t+1}^p \Lambda_{t+1} \varepsilon_{t+1}] \\ &+ f(\Phi_t, \varepsilon_t) [(1 - \gamma) \Phi_t \mathbb{E}_t m_{t+1}^c \Lambda_{t+1} + \gamma \Phi_t \mathbb{E}_t m_{t+1}^c \Lambda_{t+1} \vartheta_{t+1}] \\ &\geq f(\Phi_t, \varepsilon_t) [\Lambda_t^s + \cos(n^p, \Lambda) + \varepsilon_t] \\ &+ (1 - \gamma) \Phi_t f(\Phi_t, \varepsilon_t) [\Lambda_t^s + \cos(m^c, \Lambda)] \\ &+ \gamma \Phi_t f(\Phi_t, \varepsilon_t) [\Lambda_t^s \mathbb{E}_t \vartheta_{t+1} + \cos(\Lambda, \vartheta) + \cos(m^c, \vartheta \Lambda)] \\ &= f(\Phi_t, \varepsilon_t) [(1 + \Phi_t [1 - \gamma + \gamma \mathbb{E}_t \vartheta_{t+1}]) \Lambda_t^s + \gamma \Phi_t \cos(\Lambda, \vartheta)] \\ &+ f(\Phi_t, \varepsilon) [\varepsilon_t + \cos(n^p, \Lambda)] \\ &+ f(\Phi_t, \varepsilon_t) \Phi_t [(1 - \gamma) \cos(m^c, \Lambda) + \gamma \cos(m^c, \vartheta \Lambda)]. \end{split}$$

Adding and subtracting

$$\omega_t^{RE} = f(\Phi_t, 0)[(1 + \Phi_t[1 - \gamma + \gamma \mathbb{E}_t \vartheta_{t+1}])\Lambda_t^s + \gamma \Phi_t cov(\Lambda, \vartheta)],$$

leads to,

$$\begin{split} \varpi_t &\geq [1 - f(\Phi_t, \varepsilon_t)\varepsilon_t] \varpi_t^{RE} + f(\Phi_t, \varepsilon_t)[\varepsilon_t + cov(n^p, \Lambda)] \\ &+ f(\Phi_t, \varepsilon_t)\Phi_t[(1 - \gamma)cov(m^c, \Lambda) + \gamma cov(m^c, \vartheta\Lambda)] \\ &\geq \varpi_t^{RE}, \ \Phi_t > 0, \\ &\geq \Lambda_t^* + cov(n^p, \Lambda) > \Lambda_t^s \ \Phi_t = 0, \varepsilon_t = 0, \ \forall t. \end{split}$$
(A11)

Appendix D. Sign of Carbon Tax Premium χ

For the decompositions below, recall that the skeptic's belief multiplier m^s is assumed to be an arbitrary, independent random variable, uncorrelated with anything, where $\mathbb{E}_t m_{t+j}^s = 1$. This last property is shared by all martingale multipliers in this paper, so $\mathbb{E}_t n_{t+j}^{PO} = 1$ and $\mathbb{E}_t n_{t+j}^{PA} = 1$, as well. Where applicable, the derivations below apply Jensen's inequality to a convex function, such that, for all versions of *m* displayed in Table 1, $\mathbb{E}_t \frac{1}{m_{t+1}} \ge \frac{1}{\mathbb{E}_t m_{t+1}} = 1$. Also used, where necessary, is the law of iterated expectations.⁴² From (100),

$$\mathbb{E}_{t}\Xi^{e}_{t+j} = \mathbb{E}_{t}\left[\left(f(\Phi_{t},\varepsilon_{t})\frac{1}{m_{t+j}}(1+\varepsilon_{t+j}+[1-\gamma+\gamma\vartheta_{t+j}]n^{*}_{t+j}\Phi_{t+j-1})-1\right)\varsigma^{e}_{t+j}\right],$$

where $\vartheta_t = \frac{H_t + g_t}{c_t}$, $\Phi_{t+j} = n_{t+j}^* \Phi_{t+j-1}$, n^* is defined in Table 1, $f(\Phi_t, \varepsilon_t)$ is defined as before, and $m_{t+j} = m_{t+j}^s$ or $m_{t+j} = m_{t+j}^c$.⁴³

A. Homogeneous beliefs

RE solution $(m = 1, n^* = 1, \varepsilon = 0, \Phi_t = \bar{\Phi}, \varepsilon = 0)^{44}$

$$\mathbb{E}_{t} \mathbb{E}_{t+j}^{e-RE} = \mathbb{E}_{t} \varsigma_{t+j}^{e} \left[f(\bar{\Phi}, 0) \left(1 + [1 - \gamma + \gamma \vartheta_{t+j}] \bar{\Phi} \right) - 1 \right] \\ = f(\bar{\Phi}, 0) \mathbb{E}_{t} \varsigma_{t+j}^{e} [1 + (1 - \gamma) \bar{\Phi}] + \gamma \bar{\Phi} f(\bar{\Phi}, 0) \mathbb{E}_{t} \varsigma_{t+j}^{e} \vartheta_{t+j} - \mathbb{E}_{t} \varsigma_{t+j}^{e} \\ = f(\bar{\Phi}, 0) [1 + \bar{\Phi} (1 - \gamma + \gamma \mathbb{E}_{t} \vartheta_{t+1} + \gamma cov(\varsigma^{e}, \vartheta))] - 1 \\ = \bar{\Phi} f(\bar{\Phi}, 0) \gamma \left[\mathbb{E}_{t} \vartheta_{t+j} - \vartheta_{t} + cov(\varsigma^{e}, \vartheta) \right] > 0, \quad \bar{\Phi} > 0, \\ = 0, \quad \bar{\Phi} = 0 \; \forall t.$$
(A12)

The preceding utilizes $\mathbb{E}_t \varsigma_{t+1}^e = 1$, and

$$f(\Phi_t, 0)[1 + (1 - \gamma + \gamma \mathbb{E}_t \vartheta_{t+j})\Phi_t] - 1 = \frac{1 + (1 - \gamma + \gamma \mathbb{E}_t \vartheta_{t+j})\Phi_t}{1 + (1 - \gamma + \gamma \vartheta_t]\Phi_t} - 1$$
$$= \gamma \Phi_t f(\Phi_t, 0)(\mathbb{E}_t \vartheta_{t+j} - \vartheta_t).$$

B. Heterogeneous beliefs

1. Skeptical private sector

(a) **No ambiguity** $(m = m^s; n^* = 1 \rightarrow \Phi_t = \bar{\Phi}_t, \varepsilon = 0)^{45}$

$$\begin{split} \mathbb{E}_{t} \mathbb{E}_{t+j}^{e} &= \mathbb{E}_{t} \varsigma_{t+j}^{e} \left[\frac{f(\bar{\Phi}, 0)}{m_{t+j}^{s}} \left(1 + [1 - \gamma + \gamma \vartheta_{t+j}] \bar{\Phi} \right) - 1 \right] \\ &= f(\bar{\Phi}, 0) \mathbb{E}_{t} \varsigma_{t+j}^{e} \frac{1}{m_{t+j}^{s}} [1 + (1 - \gamma) \bar{\Phi}] \\ &+ \gamma f(\bar{\Phi}, 0) \mathbb{E}_{t} \varsigma_{t+j}^{e} \frac{1}{m_{t+j}^{s}} \vartheta_{t+j} \bar{\Phi} - \mathbb{E}_{t} \varsigma_{t+j}^{e} \\ &\geq f(\bar{\Phi}, 0) \frac{1}{\mathbb{E}_{t} m_{t+j}^{s}} \mathbb{E}_{t} [\varsigma_{t+j}^{e} (1 + (1 - \gamma) \bar{\Phi})] \\ &+ \gamma \bar{\Phi} f(\bar{\Phi}, 0) \frac{1}{\mathbb{E}_{t} m_{t+j}^{s}} \left(\mathbb{E}_{t} \varsigma_{t+j}^{e} \vartheta_{t+j} + cov(\varsigma^{e}, \vartheta) \right) - \mathbb{E}_{t} \varsigma^{e} \\ &= f(\bar{\Phi}, 0) [1 + \bar{\Phi} (1 - \gamma + \gamma [\mathbb{E}_{t} \vartheta_{t+j} + cov(\varsigma^{e}, \vartheta)])] - 1 \\ &= \gamma \bar{\Phi} f(\bar{\Phi}, 0) [\mathbb{E}_{t} \vartheta_{t+j} - \vartheta_{t} + cov(\varsigma^{e}, \vartheta)] > 0 \\ &= \mathbb{E}_{t} \mathbb{E}_{t+j}^{e-RE} > 0, \ \bar{\Phi} > 0, \\ &= 0, \ \bar{\Phi} = 0, \ \forall t. \end{split}$$

(b) **Political planner** $(m = m^s, n^* = 1, \varepsilon = 0)$.

$$\mathbb{E}_{t} \mathbb{E}_{t+j}^{e-PO} = \mathbb{E}_{t} \varsigma_{t+j}^{e} \left[\frac{f(\bar{\Phi}, 0)}{m_{t+j}^{s}} \left(1 + [1 - \gamma + \gamma \vartheta_{t+j}] \bar{\Phi} \right) - 1 \right] \\
= f(\bar{\Phi}, 0) \mathbb{E}_{t} \frac{1}{m_{t+j}^{s}} \mathbb{E}_{t} \varsigma_{t+j}^{e} [1 + (1 - \gamma) \bar{\Phi}] \\
+ \gamma f(\bar{\Phi}, 0) \mathbb{E}_{t} \frac{1}{m_{t+j}^{s}} \varsigma_{t+j}^{e} \vartheta_{t+j} \bar{\Phi} - \mathbb{E}_{t} \varsigma_{t+j}^{e} \\
\geq f(\bar{\Phi}, 0) \left[1 + \bar{\Phi} (1 - \gamma + \gamma [\mathbb{E}_{t} \vartheta_{t+j} + cov(\varsigma^{e}, \vartheta)] \right) \right] - 1 \\
= \bar{\Phi} f(\bar{\Phi}, 0) \gamma [\mathbb{E}_{t} \vartheta_{t+j} - \vartheta_{t} + cov(\varsigma^{e}, \vartheta)] > 0, \\
= \mathbb{E}_{t} \mathbb{E}_{t+j}^{e-RE}, \quad \bar{\Phi} > 0, \\
= 0, \quad \bar{\Phi} = 0, \quad \forall t.$$
(A14)

(c) **Paternalistic planner** $(m = m^s, n^* = n^{PA}; \Phi_{t+j} = n_{t+j}^{PA} \Phi_{t+j-1}, \varepsilon = 0)$

$$\mathbb{E}_{t} \mathbb{E}_{t+j}^{e-PA} = \mathbb{E}_{t} \varsigma_{t+j}^{e} \left[\frac{f(\Phi_{t}, 0)}{m_{t+j}^{s}} \left(1 + [1 - \gamma + \gamma \vartheta_{t+j}] n_{t+j}^{PA} \Phi_{t+j-1} \right) - 1 \right] \\
= f(\Phi_{t}, 0) \mathbb{E}_{t} \varsigma_{t+j}^{e} \frac{1}{m_{t+j}^{s}} [1 + (1 - \gamma) n_{t+j}^{PA} \Phi_{t+j-1}] \\
+ \gamma f(\Phi_{t}, 0) \mathbb{E}_{t} \frac{1}{m_{t+j}^{s}} \varsigma_{t+j}^{e} \vartheta_{t+j} n_{t+j}^{PA} \Phi_{t+j-1} - \mathbb{E}_{t} \varsigma_{t+j}^{e} \\
\geq f(\Phi_{t}, 0) \frac{\mathbb{E}_{t} \varsigma_{t+j}^{e}}{\mathbb{E}_{t} m_{t+j}^{s}} + (1 - \gamma) \Phi_{t} f(\Phi_{t}, 0) \frac{\mathbb{E}_{t} \varsigma_{t+j}^{e}}{\mathbb{E}_{t} m_{t+j}^{s}} \\
+ \gamma \Phi_{t} f(\Phi_{t}, 0) \left(\frac{\mathbb{E}_{t} \varsigma_{t+j}^{e}}{\mathbb{E}_{t} m_{t+j}^{s}} \mathbb{E}_{t} \vartheta_{t+j} + cov(n^{PA}, \varsigma^{e} \vartheta) + cov(\varsigma^{e}, \vartheta) \right) - \mathbb{E}_{t} \varsigma_{t+j}^{e} \\
= f(\Phi_{t}, 0) \left[1 + \Phi_{t} (1 - \gamma + \gamma [\mathbb{E}_{t} \vartheta_{t+j} + cov(\varsigma^{e}, \vartheta)] \right) - 1 \\
+ \gamma \Phi_{t} f(\Phi_{t}, 0) cov(n^{PA}, \varsigma^{e} \vartheta) \\
= \mathbb{E}_{t} \mathbb{E}_{t+j}^{e-RE} + \gamma \Phi_{t} f(\Phi_{t}, 0) cov(n^{PA}, \varsigma^{e} \vartheta) > 0, \quad \Phi_{t} > 0, \\
\geq 0, \quad \Phi_{t} = 0, \quad \forall t.$$
(A15)

(d) **Pessimistic planner**
$$(m = m^s, n^* = \frac{m^p}{n^p}; \Phi_{t+j} = \frac{m^{p+j}_{t+j}}{n^p_{t+j}} \Phi_{t+j-1}, \varepsilon = 0)^{46}$$

$$\begin{split} \mathbb{E}_{t} \mathbb{E}_{t+j}^{e} &= \mathbb{E}_{t} \varsigma_{t+j}^{e} \left[f(\Phi_{t}, 0) \frac{1}{m_{t+j}^{s}} \left(1 + [1 - \gamma + \gamma \vartheta_{t+j}] \frac{m_{t+1}^{p}}{n_{t+1}^{p}} \Phi_{t+j-1} \right) - 1 \right] \\ &= f(\Phi_{t}, 0) \mathbb{E}_{t} \left[\frac{\varsigma_{t+j}^{e}}{m_{t+j}^{s}} \left(1 + [1 - \gamma + \gamma \vartheta_{t+j}] \frac{m_{t+j}^{p}}{n_{t+j}^{p}} \Phi_{t+j-1} \right) \right] - 1 \\ &= f(\Phi_{t}, 0) \left[\mathbb{E}_{t} \left(\frac{\varsigma_{t+j}^{e}}{m_{t+j}^{s}} \right) + (1 - \gamma) \Phi_{t} \left(\mathbb{E}_{t} \frac{\varsigma_{t+j}^{e} m_{t+j}^{p}}{m_{t+j}^{s} n_{t+j}^{p}} \right) + \gamma \Phi_{t} \left(\mathbb{E}_{t} \frac{\varsigma_{t+j}^{e} \vartheta_{t+j} m_{t+j}^{p}}{m_{t+j}^{s} n_{t+j}^{p}} \right) \right] - 1 \\ &\geq f(\Phi_{t}, 0) [1 + \Phi_{t} (1 - \gamma + \gamma \mathbb{E}_{t} \vartheta_{t+1} + \gamma cov(\varsigma^{e}, \vartheta))] - 1 \\ &+ \Phi_{t} f(\Phi_{t}, 0) (1 - \gamma) \left[cov(m^{p}, \frac{1}{n^{p}}) + cov(\varsigma^{e}, \frac{m^{p}}{n^{p}}) \right] \\ &+ \Phi_{t} f(\Phi_{t}, 0) \gamma \left[cov(\vartheta, \varsigma^{e}) + cov(m^{p}, \frac{1}{n^{p}}) + cov(\varsigma^{e}, \frac{m^{p}}{n^{p}}) \right] \\ &= \mathbb{E}_{t} \Xi_{t+j}^{e-RE} + (1 - \gamma) \Phi_{t} f(\Phi_{t}, 0) \left[cov(m^{p}, \frac{1}{n^{p}}) + cov(\varsigma^{e}\vartheta, \frac{m^{p}}{n^{p}}) \right] \\ &+ \gamma \Phi_{t} f(\Phi_{t}, 0) \left[cov(\vartheta, \varsigma^{e}) + cov(m^{p}, \frac{1}{n^{p}}) + cov(\varsigma^{e}\vartheta, \frac{m^{p}}{n^{p}}) \right] \\ &= \mathbb{E}_{t} \Xi_{t+j}^{e-RE} > 0, \quad \Phi_{t} = 0. \end{split}$$

$$\tag{A16}$$

2. Pessimistic consumers

(a) **Pessimistic planner**
$$(m = m^c, n^* = \frac{m^c}{n^p}; \Phi_{t+j} = \frac{m_{t+j}^c}{n_{t+j}^p} \Phi_{t+j-1}, \varepsilon \neq 0)^{47}$$
,

$$\begin{split} \mathbb{E}_{t} \mathbb{E}_{t+j}^{e} &= \mathbb{E}_{t} \xi_{t+j}^{e} \left[f(\Phi_{t}, \varepsilon_{t}) \frac{1}{m_{t+j}^{e}} \left(1 + \varepsilon_{t+j} + [1 - \gamma + \gamma \vartheta_{t+j}] \frac{m_{t+1}^{e}}{n_{t+j}^{p}} \Phi_{t+j-1} \right) - 1 \right] \\ &= f(\Phi_{t}, \varepsilon_{t}) \mathbb{E}_{t} \left[\frac{\xi_{t+j}^{e}}{m_{t+j}^{e}} \left(1 + \varepsilon_{t+j} + [1 - \gamma + \gamma \vartheta_{t+j}] \frac{m_{t+j}^{e}}{n_{t+j}^{p}} \Phi_{t+j-1} \right) \right] - 1 \\ &\geq f(\Phi_{t}, \varepsilon_{t}) \left[\frac{\mathbb{E}_{t} \xi_{t+j}^{e}}{\mathbb{E}_{t} m_{t+j}^{e}} + \varepsilon_{t} + \cos(\zeta^{e}, \frac{1}{m^{c}}) \right] - 1 \\ &+ (1 - \gamma) \Phi_{t} f(\Phi_{t}, \varepsilon_{t}) \left[\left(\frac{\mathbb{E}_{t} \xi_{t+j}^{e}}{\mathbb{E}_{t} n_{t+j}^{p}} \right) + \cos(\zeta^{e}, \frac{1}{n^{p}}) \right] \\ &+ \gamma \Phi_{t} f(\Phi_{t}, \varepsilon_{t}) \left[\left(\frac{\mathbb{E}_{t} \xi_{t+j}^{e} \vartheta_{t+j}}{\mathbb{E}_{t} n_{t+j}^{e}} \right) + \cos(\zeta^{e}, \frac{1}{n^{p}}) \right] \\ &= f(\Phi_{t}, \varepsilon_{t}) \left[1 + \varepsilon_{t} + \cos(\zeta^{e}, \frac{1}{m^{c}}) \right] - 1 \\ &+ (1 - \gamma) \Phi_{t} f(\Phi_{t}, \varepsilon_{t}) \left[1 + \varepsilon_{t} + \cos(\zeta^{e}, \frac{1}{n^{p}}) \right] \\ &= f(\Phi_{t}, \varepsilon_{t}) \left[1 + \varepsilon_{t} + \cos(\zeta^{e}, \frac{1}{m^{p}}) \right] \\ &+ \gamma \Phi_{t} f(\Phi_{t}, \varepsilon_{t}) \left[\mathbb{E}_{t} \vartheta_{t+j} + \cos(\frac{1}{n^{p}}, \zeta^{e} \vartheta) + \cos(\zeta^{e}, \vartheta) \right] \\ &= f(\Phi_{t}, \varepsilon_{t}) \left[1 + \Phi_{t} (1 - \gamma + \gamma (\mathbb{E}_{t} \vartheta_{t+j} + \cos(\zeta^{e}, \vartheta))) \right] - 1 \\ &+ f(\Phi_{t}, \varepsilon_{t}) \left[\varepsilon_{t} + \cos(\zeta^{e}, \frac{1}{m^{c}}) \right] \\ &+ \Phi_{t} f(\Phi_{t}, \varepsilon_{t}) \left[(1 - \gamma) \cos(\zeta^{e}, \frac{1}{n^{p}}) + \gamma \cos(\vartheta^{e}, \frac{1}{n^{p}}) \right]. \end{split}$$

Adding and subtracting

$$f(\Phi_t, 0)[1 + \Phi_t(1 - \gamma + \gamma \mathbb{E}_t \vartheta_{t+1} + \gamma cov(\varsigma^e, \vartheta))],$$

leads to

$$\mathbb{E}_{t}\Xi_{t+j}^{e} = \mathbb{E}_{t}\Xi_{t+j}^{e-RE} - f(\Phi_{t},\varepsilon_{t})f(\Phi_{t},0)[1+\Phi_{t}(1-\gamma+\gamma\mathbb{E}_{t}\vartheta_{t+1}+\gamma cov(\varsigma^{e},\vartheta))]\varepsilon_{t} + f(\Phi_{t},\varepsilon_{t})\left[\varepsilon_{t}+cov(\varsigma^{e},\frac{1}{m^{c}})\right] + \Phi_{t}f(\Phi_{t},\varepsilon_{t})\left[(1-\gamma)cov(\varsigma^{e},\frac{1}{n^{p}})+\gamma cov(\vartheta\varsigma^{e},\frac{1}{n^{p}})\right] \stackrel{<}{\leq} 0, \ \Phi_{t} > 0, \ \forall t = cov(\varsigma^{e},\frac{1}{m^{c}}) < 0, \ \Phi_{t} = 0, \ \forall t,$$
(A17)

where, from (75) and (77),

$$\begin{split} \mathbb{E}_{t} \frac{\varsigma_{t+1}^{e}}{m_{t+j}^{c}} \varepsilon_{t+1} &= \sigma^{c} M_{t}^{c} \left(\mathbb{E}_{t} \varsigma_{t+1}^{e} \mu_{t+1} - \mathbb{E}_{t} \varsigma_{t+1}^{e} \mathbb{E}_{t} m_{t+1}^{c} \mu_{t+1} \right) + \mathbb{E}_{t} \frac{\varsigma_{t+1}^{e}}{m_{t+1}^{c}} \varepsilon_{t} \\ &\geq \sigma^{c} \Phi_{t} \left[\mathbb{E}_{t} (1 - m_{t+1}^{c}) \mathbb{E}_{t} u_{c,t+1} b_{t+1} + cov(\varsigma^{e}, bu_{c}) - cov(m^{c}, bu_{c}) \right] \\ &+ \left[1 + cov(\varsigma^{e}, \frac{1}{m^{c}}) \right] \varepsilon_{t} \\ &= \sigma^{c} \Phi_{t} [cov(\varsigma^{e}, bu_{c}) - cov(m^{c}, bu_{c})] + \left[1 + cov(\varsigma^{e}, \frac{1}{m^{c}}) \right] \varepsilon_{t} \geq \varepsilon_{t}. \end{split}$$

Appendix E. Sign of Ex Ante Capital Tax

This appendix evaluates the expression (112) for each of the four belief regimes,

$$\mathbb{E}_{t}\Xi_{t+1}^{k} = \mathbb{E}_{t}\varsigma_{t+1}^{k} \bigg(1 - f(\Phi_{t},\varepsilon_{t}) \frac{n_{t+1}^{**}}{m_{t+1}} (1 + \varepsilon_{t+1} + [1 - \gamma + \gamma\vartheta_{t+1}]n_{t+1}^{*}\Phi_{t} \bigg).$$

A. Homogeneous beliefs **RE solution** (m = 1, $n^* = 1$, $n^{**} = 1$, $\varepsilon = 0$)

$$\mathbb{E}_{t} \mathbb{E}_{t+1}^{k-RE} = \mathbb{E}_{t} \Big[\varsigma_{t+1}^{k} (1 - f(\Phi_{t}, 0)[1 + \Phi_{t}[1 - \gamma + \gamma \vartheta_{t+1}]]) \Big] \\
= \mathbb{E}_{t} \varsigma_{t+1}^{k} - f(\Phi_{t}, 0)[1 + (1 - \gamma)\Phi_{t}] \mathbb{E}_{t} \varsigma_{t+1}^{k} + \gamma \Phi_{t} f(\Phi_{t}, 0) \mathbb{E}_{t} \Big[\varsigma_{t+1}^{k} \vartheta_{t+1} \Big] \\
= 1 - f(\Phi_{t}, 0) \Big[1 + \Phi_{t} \Big(1 - \gamma + \gamma [\mathbb{E}_{t} \vartheta_{t+1} + cov(\varsigma^{k}, \vartheta)] \Big) \Big] \\
= -\gamma f(\Phi_{t}, 0) \Phi_{t} \Big[\mathbb{E}_{t} \vartheta_{t+1} - \vartheta_{t} + cov(\varsigma^{k}, \vartheta) \Big] < 0, \quad \Phi_{t} > 0, \quad \forall t, \\
= 0, \quad \Phi_{t} = 0 \quad \forall t.$$
(A18)

B. Heterogeneous beliefs

1. Skeptical private sector

(a) **No ambiguity** $(m = m^s, n^* = 1, n^{**} = 1, \varepsilon = 0)$.

$$\begin{split} \mathbb{E}_{t} \mathbb{E}_{t+1}^{k} &= \mathbb{E}_{t} \left[\varsigma_{t+1}^{k} \left(1 - \frac{f(\Phi_{t}, 0)}{m_{t+1}^{s}} (1 + [(1 - \gamma) + \gamma \vartheta_{t+1}] \Phi_{t}) \right) \right] \\ &= \mathbb{E}_{t} \varsigma_{t+1}^{k} - f(\Phi_{t}, 0) \mathbb{E}_{t} \varsigma_{t+1}^{k} \left[\frac{1}{m_{t+1}^{s}} (1 + [(1 - \gamma) + \gamma \vartheta_{t+1})] \Phi_{t} \right] \\ &\leq 1 - f(\Phi_{t}, 0) \frac{\mathbb{E}_{t} \varsigma_{t+1}^{k}}{\mathbb{E}_{t} m_{t+1}^{s}} - (1 - \gamma) \Phi_{t} f(\Phi_{t}, 0) \mathbb{E}_{t} \varsigma_{t+1}^{k} \\ &- \gamma \Phi_{t} f(\Phi_{t}, 0) \mathbb{E}_{t} \varsigma_{t+1}^{k} \mathbb{E}_{t} \vartheta_{t+1} - \gamma \Phi_{t} f(\Phi_{t}, 0) cov(\varsigma^{k}, \vartheta) \\ &= 1 - f(\Phi_{t}, 0) \left[1 + \Phi_{t} \left(1 - \gamma + \gamma ([\mathbb{E}_{t} \vartheta_{t+1} + cov(\varsigma^{k}, \vartheta)] \right) \right] \\ &= -\gamma \Phi_{t} f(\Phi_{t}, 0) \left[\mathbb{E}_{t} \vartheta_{t+1} - \vartheta_{t} + cov(\varsigma^{k}, \vartheta) \right] \\ &= \mathbb{E}_{t} \mathbb{E}_{t+1}^{k-RE} < 0, \ \Phi_{t} > 0, \ \forall t \\ &= 0, \ \Phi_{t} = 0 \ \forall t. \end{split}$$
(A19)

(b) **Political planner** $(m = m^s, n^* = 1, n^{**} = n^{PO}, \varepsilon = 0).$

$$\begin{split} \mathbb{E}_{t} \mathbb{E}_{t+1}^{k-PO} &= \mathbb{E}_{t} \left[\varsigma_{t+1}^{k} \left(1 - f(\bar{\Phi}, 0) \frac{n_{t+1}^{PO}}{m_{t+1}^{s}} [1 + [1 - \gamma + \gamma \vartheta_{t+1}] \bar{\Phi}] \right) \right] \\ &= \mathbb{E}_{t} \varsigma_{t+1}^{k} - f(\bar{\Phi}, 0) \left[1 + (1 - \gamma) \bar{\Phi} \mathbb{E}_{t} \varsigma_{t+1}^{k} \frac{n_{t+1}^{PO}}{m_{t+1}^{s}} + \gamma \bar{\Phi} \mathbb{E}_{t} \varsigma_{t+1}^{k} \frac{n_{t+1}^{PO}}{m_{t+1}^{s}} \vartheta_{t+1} \right] \\ &\leq 1 - f(\bar{\Phi}, 0) \left[1 + \Phi \left(1 - \gamma + \gamma [\mathbb{E}_{t} \vartheta_{t+1} + \cos(\varsigma^{k}, \vartheta)] \right) \right) \\ &- \gamma \bar{\Phi} f(\bar{\Phi}, 0) \cos(n^{PO}, \varsigma^{k} \vartheta) - (1 - \gamma) \bar{\Phi} f(\bar{\Phi}, 0) \cos(n^{PO}, \varsigma^{k}) \\ &= 1 - f(\bar{\Phi}, 0) \left[1 + \left(1 - \gamma + \gamma \mathbb{E}_{t} \vartheta_{t+1} + \gamma \cos(\varsigma^{k}, \vartheta) \right) \bar{\Phi} \right] \\ &- \bar{\Phi} f(\bar{\Phi}, 0) \left[(1 - \gamma) \cos(n^{PO}, \varsigma^{k}) + \gamma \cos(n^{PO}, \varsigma^{k} \vartheta) \right] \\ &= \mathbb{E}_{t} \Xi_{t+1}^{k-RE} - \Phi f(\Phi, 0) \left[(1 - \gamma) \cos(n^{PO}, \varsigma^{k}) + \gamma \cos(n^{PO}, \varsigma^{k} \vartheta) \right] \\ &= 0, \ \bar{\Phi} = 0. \end{split}$$
(A20)

(c) **Paternalistic planner**
$$(m = m^s, n^* = n^{PA}, n^{**} = 1, \varepsilon = 0).$$

$$\begin{split} \mathbb{E}_{t} \mathbb{E}_{t+1}^{k-PA} &= \mathbb{E}_{t} \zeta_{t+1}^{k} - f(\Phi_{t}, 0) \mathbb{E}_{t} \frac{\zeta_{t+1}^{k}}{m_{t+1}^{s}} \Big[(1 + [(1 - \gamma) + \gamma \vartheta_{t+1}] n_{t+1}^{PA} \Phi_{t} \Big] \\ &\leq 1 - f(\Phi_{t}, 0) \left[\frac{\mathbb{E}_{t} \zeta_{t+1}^{k}}{\mathbb{E}_{t} m_{t+1}^{s}} + \Phi_{t} (1 - \gamma) \mathbb{E}_{t} \zeta_{t+1}^{k} \frac{\mathbb{E}_{t} n_{t+1}^{PA}}{\mathbb{E}_{t} m_{t+1}^{s}} \right] \\ &- \gamma \Phi_{t} f(\Phi_{t}, 0) \left[\mathbb{E}_{t} \zeta_{t+1}^{k} \frac{\mathbb{E}_{t} n_{t+1}^{PA}}{\mathbb{E}_{t} m_{t+1}^{s}} \mathbb{E}_{t} \vartheta_{t+1} + cov(n^{PA}, \zeta^{k} \vartheta) + cov(\zeta^{k}, \vartheta) \right] \\ &- (1 - \gamma) \Phi_{t} f(\Phi_{t}, 0) cov(n^{PA}, \zeta^{k}) \\ &= 1 - f(\Phi_{t}, 0) \Big[1 + \Phi_{t} \Big(1 - \gamma + \gamma [\mathbb{E}_{t} \vartheta_{t+1} + \gamma cov(\zeta^{k}, \vartheta)] \Big) \Big] \\ &- \Phi_{t} f(\Phi_{t}, 0) \Big[(1 - \gamma) cov(n^{PA}, \zeta^{k}) + \gamma cov(n^{PA}, \zeta^{k} \vartheta) \Big] \\ &= \mathbb{E}_{t} \Xi_{t+1}^{k-RE} - \Phi_{t} f(\Phi_{t}, 0) \Big[(1 - \gamma) cov(n^{PA}, \zeta^{k}) + \gamma cov(n^{PA}, \zeta^{k} \vartheta) \Big] \leq \mathbb{E}_{t} \Xi_{t+1}^{k-RE} < 0, \Phi_{t} > 0, \forall t \\ &= 0, \ \Phi_{t} = 0, \forall t. \end{split}$$

(d) **Pessimistic planner** $(m = m^s, n^* = \frac{m^p}{n^p}, n^{**} = n^p, \varepsilon = 0)$

$$\begin{split} \mathbb{E}_{t} \Xi_{t+1}^{k-R-s} &= \mathbb{E}_{t} \left[\zeta_{t+1}^{k} \left(1 - f(\Phi_{t}, 0) \frac{n_{t+1}^{k}}{m_{t+1}^{s}} \left[1 + [1 - \gamma + \gamma \vartheta_{t+1}] \frac{m_{t+1}^{p}}{n_{t+1}^{p}} \Phi_{t} \right] \right) \right] \\ &= \mathbb{E}_{t} \zeta_{t+1}^{k} - f(\Phi_{t}, 0) \left[\mathbb{E}_{t} \frac{n_{t+1}^{p}}{m_{t+1}^{s}} \zeta_{t+1}^{k} + \Phi_{t} \mathbb{E}_{t} (1 - \gamma + \gamma \vartheta_{t+1}) \frac{m_{t+1}^{p} \zeta_{t+1}^{k}}{m_{t+1}^{s}} \right] \\ &\leq 1 - f(\Phi_{t}, 0) \left[1 + \cos(\zeta^{k}, n^{p}) + (1 - \gamma) \Phi_{t} (1 + \cos(\zeta^{k}, m^{p})) \right] \\ &- \gamma \Phi_{t} f(\Phi_{t}, 0) \left[\mathbb{E}_{t} \vartheta_{t+1} + \cos(\zeta^{k}, \vartheta) + \cos(\vartheta, m^{p} \zeta^{k}) \right] \\ &= 1 - f(\Phi_{t}, 0) \left[1 + \Phi_{t} \left(1 - \gamma + \gamma [\mathbb{E}_{t} \vartheta_{t+1} + \cos(\zeta^{k}, \vartheta)] \right) \right] \\ &- f(\Phi_{t}, 0) [\cos(\zeta^{k}, n^{p}) + (1 - \gamma) \Phi_{t} \cos(\zeta^{k}, m^{p})] - \gamma \Phi_{t} f(\Phi_{t}, 0) \cos(\vartheta, \zeta^{e} m^{p}) \right] \\ &= \mathbb{E}_{t} \Xi_{t+1}^{k-RE} - f(\Phi_{t}, 0) [\cos(\zeta^{k}, n^{p}) + (1 - \gamma) \cos(\zeta^{k}, m^{p})] \\ &- \gamma \Phi_{t} f(\Phi_{t}, 0) \cos(\vartheta, \zeta^{e} m^{p}) \leq 0, \quad \Phi_{t} \geq 0 \\ &= \mathbb{E}_{t} \Xi_{t+1}^{k-RE} - \cos(\zeta^{k}, n^{p}) - (1 - \gamma) \cos(\zeta^{k}, m^{p}) \leq 0, \quad \Phi_{t} = 0. \end{split}$$
(A22)

2. Pessimistic consumers

(a) **Pessimistic planner** $(m = m^c, n^* = \frac{m^c}{n^p}, n^{**} = n^p, \varepsilon \neq 0)$

$$\begin{split} \mathbb{E}_{t} \Xi_{t+1}^{k-R-p} &= \mathbb{E}_{t} \left[\varsigma_{t+1}^{k} \left(1 - f(\Phi_{t}, \varepsilon_{t}) \frac{n_{t+1}^{p}}{m_{t+1}^{c}} \left[1 + \varepsilon_{t+1} + [1 - \gamma + \gamma \vartheta_{t+1}] \frac{m_{t+1}^{c}}{n_{t+1}^{p}} \Phi_{t} \right] \right) \right] \\ &= \mathbb{E}_{t} \varsigma_{t+1}^{k} - f(\Phi_{t}, \varepsilon_{t}) \left[\mathbb{E}_{t} \frac{n_{t+1}^{p}}{m_{t+1}^{c}} \varsigma_{t+1}^{k} (1 + \varepsilon_{t+1}) + \Phi_{t} \mathbb{E}_{t} (1 - \gamma + \gamma \vartheta_{t+1}) \varsigma_{t+1}^{k} \right] \\ &\leq 1 - f(\Phi_{t}, \varepsilon_{t}) \left[\frac{\mathbb{E}_{t} n_{t+1}^{p}}{\mathbb{E}_{t} m_{t+1}^{c}} \mathbb{E}_{t} \varsigma_{t+1}^{k} + \Phi_{t} (1 - \gamma + \gamma \mathbb{E}_{t} \vartheta_{t+1}) \mathbb{E}_{t} \varsigma_{t+1}^{k} \right] \\ &- f(\Phi_{t}, \varepsilon_{t}) \left[\gamma \Phi_{t} cov(\varsigma^{k}, \vartheta) + cov(\varsigma^{k}, \frac{n^{p}}{m^{c}}) + cov(\frac{1}{m^{c}}, n^{p}) - (1 + \varepsilon_{t}) cov(\varsigma^{k}, n^{p}) \right] \\ &= 1 - f(\Phi_{t}, \varepsilon_{t}) \left[1 + \Phi_{t} (1 - \gamma + \gamma (\mathbb{E}_{t} \vartheta_{t+1} + cov(\varsigma^{k}, \vartheta)) \right] \\ &- f(\Phi_{t}, \varepsilon_{t}) \left[cov(\varsigma^{k}, \frac{n^{p}}{m^{c}}) + cov(\frac{1}{m^{c}}, n^{p}) - (1 + \varepsilon_{t}) cov(\varsigma^{k}, n^{p}) \right]. \end{split}$$

Adding and subtracting

$$f(\Phi_t, 0) \Big[1 + \Phi_t \Big(1 - \gamma + \gamma [\mathbb{E}_t \vartheta_{t+1} + cov(\varsigma^k, \vartheta)] \Big) \Big],$$

leads to

$$\mathbb{E}_{t}\Xi_{t+1}^{k-R-p} \leq \mathbb{E}_{t}\Xi_{t+1}^{k-RE} - f(\Phi_{t},\varepsilon_{t})f(\Phi_{t},0)\left[1 + \Phi_{t}\left(1 - \gamma + \gamma[\mathbb{E}_{t}\vartheta_{t+1} + cov(\varsigma^{k},\vartheta)]\right)\right]\varepsilon_{t} \\ - f(\Phi_{t},\varepsilon_{t})\left[cov(\varsigma^{k},\frac{n^{p}}{m^{c}}) + cov(\frac{1}{m^{c}},n^{p}) + (1 + \varepsilon_{t})cov(\varsigma^{k},n^{p})\right] \stackrel{\leq}{\leq} 0, \quad \Phi_{t} > 0 \\ = -\left[cov(\varsigma^{k},\frac{n^{p}}{m^{c}}) + cov(\frac{1}{m^{c}},n^{p}) + (1 + \varepsilon_{t})cov(\varsigma^{k},n^{p})\right] \stackrel{\leq}{\leq} 0, \quad \Phi_{t} = 0,$$
(A23)

where, from (75) and (77),

$$\begin{split} \mathbb{E}_{t} \frac{\zeta_{t+1}^{k} n_{t+1}^{p}}{m_{t+1}^{c}} \varepsilon_{t+1} &= \sigma^{c} M_{t}^{c} \Big(\mathbb{E}_{t} \zeta_{t+1}^{k} n_{t+1}^{p} \mu_{t+1} - \mathbb{E}_{t} \zeta_{t+1}^{k} n_{t+1}^{p} \mathbb{E}_{t} m_{t+1}^{c} \mu_{t+1} \Big) \\ &+ \mathbb{E}_{t} \frac{\zeta_{t+1}^{k} n_{t+1}^{p}}{m_{t+1}^{c}} \varepsilon_{t} \\ &= \sigma^{c} \Phi_{t} \mathbb{E}_{t} n_{t+1}^{p} \zeta_{t+1}^{k} \mathbb{E}_{t} [1 - m_{t+1}^{c}] u_{c,t+1} b_{t+1} \\ &+ \sigma^{c} \Phi_{t} [cov(\zeta^{k} n^{p}, bu_{c}) - cov(m^{c}, bu_{c})] \\ &+ [1 + cov(\zeta^{k}, \frac{n^{p}}{m^{c}})] \varepsilon_{t} \\ &= \sigma^{c} \Phi_{t} \Big[cov(\zeta^{k}, bu_{c}) - cov(m^{c}, bu_{c}) \Big] \\ &+ (1 + \varepsilon_{t}) cov(\zeta^{k}, \frac{n^{p}}{m^{c}}) \ge \varepsilon_{t}. \end{split}$$

Notes

- ¹ The Yale Climate Opinion Maps 2020 (Marlon et al. 2020) is encouraging in that 72 percent of respondents said global warming is real and a threat to humanity and the planet, with 57 percent believing it to be human caused.
- ² Uncertainty in climate modeling has been intensively treated in the literature; see Pindyck (2007, 2013b), Roe and Baker (2007), Weitzman (2007, 2009, 2013), Heal and Millner (2013), and lately Barnett et al. (2020).

- ³ Tipping points include shutdown of the Atlantic Meridional Overturning Circulation, West Antarctic ice sheet disintegration, Amazon rainforest dieback, West African monsoon shift, permafrost and methane hydrates, coral reef die-off, Indian monsoon shift, Greenland ice sheet disintegration, boreal forest shift, and permafrost and methane hydrates. Dietz et al. (2021) note that such "climate tipping points are subject to considerable scientific uncertainty in relation to their size, probability, and how they interact with each other. Their economic impacts are even more uncertain, and consequently, these are often ignored or given a highly stylized treatment that fails to accurately represent geophysical dynamics and is nearly impossible to calibrate. As a result, tipping points are only weakly reflected in the policy advice economists give on climate change, typically by way of caveats and contextualization, rather than an integral part of the modeling that gives rise to estimates of the social cost of carbon (SCC) and other economic metrics of interest.
- ⁴ Barrage (2020) is perhaps the first to study optimal capital and carbon taxation in Ramsey planning framework.
- ⁵ This is not the first paper to dwell on a symmetry between carbon and capital taxation arising from a difference between social and private discount factors. Barrage's (2020) shows that the very logic leading to an optimal zero *ex ante* tax on capital derived in (Atkeson et al. 1999; Chari et al. 1994; Zhu 1992) implies a *positive* Pigou tax on carbon.
- ⁶ In a recent paper, Dietz and Niehörster (2019), estimate ambiguity loads—i.e., the extra insurance premiums due to ambiguity and show how these depend on the insurer's attitude to ambiguity.
- ⁷ A treatment of the proper social discount rate for far-away and potentially catastrophic consequences of today's actions is beyond the scope of this paper. For discussions, see Weitzman (2009, 2013), Pindyck (2013a), and Traeger (2014).
- ⁸ In Belfiori's (2015) inter-generational model of altruism, the difference in discount factors is endogenous because each generation of households living in an infinite sequence of generations assigns a positive weight to the welfare of future generations, causing the government's discount factor to rise above that of households and leading to a carbon tax that exceeds the standard Pigou tax.
- ⁹ An axiomatization of multiplier preferences can be found in Strzalecki (2011).
- ¹⁰ Throughout, the expectations operator $\mathbb{E}_t y(x_{t+j})$ denotes the mathematical expectations of some function $y(x_{t+j}), j \ge 1$, with respect to the probability density $\pi(x_{t+j}|x^t)$: $\mathbb{E}_t y(x_{t+j}) = \sum_{x^{t+j}|x^t} y(x_{t+j}) \pi(x_{t+j}|x^t)$.
- ¹¹ This means one cannot assign positive probability to events as functions of x_t that have probability measure zero under the distribution of the approximating model or alternatively, the distorted and the approximating distributions are, at the very least, in agreement about which events have zero probability and which events are certain.
- ¹² Previous systematic treatments of decision making under ambiguity include Klibanoff et al. (2005, 2009), and Traeger (2014).
- ¹³ Following examples by Anderson et al. (2013), Golosov et al. (2014), Nordhaus (2008), Li et al. (2016), and others, I omit leisure (or hours worked) to keep things simple. I also leave out any utility effects of environmental quality caused by climate change, since no conclusions relevant for this paper would be affected by their inclusion. See however, Barrage (2020).
- ¹⁴ See Hennlock (2009). While $\gamma < 1$ gets at early resolution of uncertainty, as noted by Bansal and Yaron (2004), the assumption of CRRA misses certain aspects of risk aversion that may create pricing issues.
- ¹⁵ For analytical convenience that will not affect conclusions, I will assume that energy production is costless, as is approximately true for oil (see Golosov et al. 2014). Accordingly, it is reasonable to fold the fossil energy extraction/production sector into the household, by making the consumer be the owner of the resource Q and the seller of energy E_t . This assumption differs from the literature (see Golosov et al. 2014; Nordhaus 1993), where, instead, the consumer owns the energy producing firm and receives its profits. However, ownership of the firm is merely a financial veil for direct ownership of the resource itself, hence a distinction without a difference.
- ¹⁶ The implied relative risk aversion is $CRR = 1 \frac{\sigma^c}{1-\beta}$ that rises as $\sigma^c \to -\infty$. and $0 > \sigma^c > -\infty$. See Epstein and Zin (1991).
- ¹⁷ The formula for *m^c* can be derived as a special case of Epstein and Zin (1989) and Weil (1990) preferences

$$\mathcal{U}_t = [(1-\beta)u_t^{1-\rho} + \beta(\mathbb{E}_t \mathcal{U}_{t+1}^{1-\gamma})^{\frac{1-\rho}{1-\gamma}}]^{\frac{1}{1-\rho}}$$

where $\frac{1}{\rho}$ is the intertemporal elasticity of substitution. For $\gamma > 0$ and $\rho = 1$, this has the interpretation of a *risk-sensitive* recursion, such that in (6), $\sigma^c \equiv (1 - \beta)(1 - \gamma)$. With $\gamma = \rho$, preferences reduce to standard time-additive expected utility.

- Readers will recognize Hansen and Sargent's (1995) discounted risk-sensitive recursion related to recursive preferences introduced earlier by Epstein and Zin (1989) and Weil (1990). Risk-sensitivity as a concept was introduced by Jacobson (1973) and later familiarized by Whittle (1981).
- ¹⁹ Excellent expositions of the physics of climate change for economists are Hsiang and Kopp (2018) and Traeger's (2018) description of the complex sets of channels involved in the transmission from carbon to temperature change.
- ²⁰ Barnett et al. (2020) also adopt this measure of climate response.
- ²¹ Modeling damages resulting from climate change as negative total productivity shocks to the economy follows common practice, as for example in Golosov et al. (2014). The damaging effects of increased temperature on productivity in the world economy have been documented by Burke et al. (2015). Donadelli et al. (2017) used vector autoregressions to show that with a 50-year horizon, a one-standard deviation temperature shock lowers both cumulative output and labor productivity growth by 1.4

percentage points. Based on their model, they further show that temperature risk is associated with non-negligible welfare costs of 18.4% of the agent's lifetime utility that grow exponentially with the size of the impact of temperature on TFP.

- 22 Belfiori (2017) models climate damage as a reduction in household utility, while Barrage (2020) considers reductions in both production and utility.
- ²³ For the derivation see Appendix A.
- ²⁴ Henceforth, the *n* multiplier refers to the government's belief, and *m* to the consumer's belief distortion.
- ²⁵ In this paper, I do not consider learning by either planner or private agents. An approach to doing so is available in Tetlow and von zur Muehlen (2009) who study robust monetary policy using *structured singular value* analysis when agents have misspecified models but are engaged in learning under the handicap that their learning of the reduced form of the economy is subject to potentially destabilizing parameter perturbations.
- ²⁶ Hansen and Sargent's (2012) refer to this as *Type I ambiguity*.
- 27 The first application of risk-sensitive decision making to economic policy is van der Ploeg (1984). An early treatment of Knightian uncertainty in economics is von zur Muehlen (1982). In his analytical climate economy model, Traeger (2018) posits risk-sensitive preferences attributable to the planner representing the consumer.
- ²⁸ Hansen and Sargent's (2012) refer to this as *Type III ambiguity*.
- ²⁹ This belief regime is related to the robust fiscal policy model in Karantounias (2020). A fifth possible belief regime, wherein the planner has doubts but the private sector trusts the extant model, is not treated here having been widely discussed in the referenced literature.
- ³⁰ In Table 1, p is defined by (19).
- ³¹ Current estimates of the social cost of carbon emissions are around USD75 per ton of carbon. Some consider this a gross underestimate and suggest the number is closer to USD220 per ton. (See Moore and Diaz (2015) and Than (2015)).
- ³² Initial wealth W_0 is a function of k_0 and S_0 and the initial tax rates τ_0^k and τ_0^e . It is well known that since initial capital is supplied inelastically, the government has an incentive to raise the initial capital tax as high as possible. Likewise, with S_0 given, there is nothing (i.e., no welfare criterion) to prevent the planner from expropriating the energy sector by setting $\tau_0^e = 1$. As is conventional, I fix both τ_0^k and τ_0^e at 0.
- ³³ By definition, next period's stock $Q_{t+1} = Q_t E_t$ is given, i.e., determined by previous E_t , hence not influenced by x_{t+1} .
- ³⁴ A positive warming shock reduces both consumption and productivity and therefore the value of future welfare V_{t+1} .
- ³⁵ Wilbanks et al. (2008) present empirical evidence that the net effect of declining consumption on wealth is negative.
- ³⁶ Proofs are provided in Appendixes C–E, and utilize the preceding assumptions and lemmas, particularly the results that n^p , n^{POL} , n^{PAT} , m^p , ς^k , ς^e , and Λ , being positively correlated with x, are positively correlated with with each other.
- ³⁷ Pascal argued that a rational person should live as though God exists and seek to believe in God. If God does not exist, such a person will have only a finite loss (some pleasures, luxury, etc.), whereas if God does exist, he stands to receive infinite gains (as represented by eternity in Heaven) and avoid infinite losses (an eternity in Hell).
- ³⁸ This section is inspired by Karantounias (2013) who studied a fiscal authority's ability to manage pessimistic expectations.
- ³⁹ See for example, Traeger (2018). Other features not addressed here include (i) *Arrhenius' Greenhouse Law* describing radiative forcing that connects carbon with temperature change, as described in Hassler et al. (2016), (ii) *tipping points* analyzed by Lemoine and Traeger (2016) and (Cai et al. 2013), describing abrupt nonlinear climate changes that pose a potentially existential threat to humanity in ways that may override concerns with belief and skepticism, and (iii) *polar amplification* analyzed by Brock and Xepapadeas (2017).
- ⁴⁰ The derivations in this and the next two Appendices utilize these facts: Let $\{x, y, z\}$ represent four random, possibly related variables, and *a* a non-random variable or constant. Then

$$\mathbb{E}[ax \times y] = a\mathbb{E}x\mathbb{E}y + acov(x, y)$$

$$\mathbb{E}[ax \times y \times z] = a\mathbb{E}x\mathbb{E}y\mathbb{E}z + acov(x, yz) + a\mathbb{E}xcov(y, z).$$

⁴¹ The derivation uses (77)

42

$$\begin{split} \mathbb{E}_{t}n_{t+1}^{p}\Lambda_{t+1}\varepsilon_{t+1} &= \sigma^{c}\Phi_{t}\mathbb{E}_{t}[n_{t+1}^{p}\Lambda_{t+1}u_{c_{t+1}}][\mathbb{E}_{t}m_{t+1}^{c}b_{t+1} - \mathbb{E}_{t}m_{t+1}^{c}\mathbb{E}_{t}m_{t+1}^{c}b_{t+1}] + E_{t}n_{t+1}^{p}\Lambda_{t+1}m_{t+1}^{c}\varepsilon_{t} \\ &+ \sigma^{c}\Phi_{t}[cov(n^{p}u_{c}\Lambda,m^{c}b) - cov(n^{p}u_{c}\Lambda,m^{c})] \\ &= \sigma^{c}\Phi_{t}[cov(n^{p}u_{c}\Lambda,m^{c}b) - cov(n^{p}u_{c}\Lambda,m^{c})] + [\Lambda_{t}^{*} + cov(n^{p}m^{c},\Lambda]\varepsilon_{t} \\ &> \varepsilon_{t}. \end{split}$$

since, with the exception of marginal utility u_c , m^c , n^p , Λ , and b are positively related to x and therefore with each other. By the law of iterated expectations $\mathbb{E}_t \mathbb{E}_{t+1} \cdots \mathbb{E}_{t+j-1} [\Xi_{t+j}^e] = \mathbb{E}_t [\Xi_{t+j}^e]$. ⁴³ Where called for, the proofs use the result that for any martingale process m, $\mathbb{E}m = 1$, if

$$\Phi_{t+j} = m_{t+j} \Phi_{t+j-1}$$

then,

$$\mathbb{E}_t \Phi_{t+j} = \mathbb{E}_t m_{t+j} \Phi_{t+j-1} = \mathbb{E}_t \prod_{i=1}^j \frac{M_{t+i}}{M_{t+i-1}} \Phi_t = \mathbb{E}_t \frac{M_{t+j}}{M_t} \Phi_t = \frac{M_t}{M_t} \Phi_t = \Phi_t.$$

⁴⁴ Constancy of Φ follows from (71).

- In the following, m^s is purely random and independent of any other variable in the economy. Likewise, Φ_{t+j} is independent, where from footnote 42, $\mathbb{E}_t m^s_{t+j} \Phi_{t+j-1} = \Phi_t$, and $\mathbb{E}_t \varsigma^e_{t+j} = 1$.
- ⁴⁶ In the following, I use the facts that $\mathbb{E}m = 1$ for each *m*, and also that $\mathbb{E}\varsigma^e = 1$. In addition, note that

$$\begin{split} \mathbb{E}_{t} \Phi_{t+j} &= \mathbb{E}_{t} \frac{m_{t+j}^{p}}{n_{t+j}^{p}} \Phi_{t+j-1} = \mathbb{E}_{t} \left(\frac{m_{t+j}^{p}}{m_{t+j-1}^{p}} / \frac{N_{t+j}^{p}}{N_{t+j-1}^{p}} \right) \Phi_{t+j-1} = \mathbb{E}_{t} \prod_{i=1}^{j} \left(\frac{m_{t+i}^{p}}{m_{t+i-1}^{c}} / \frac{N_{t+i}^{p}}{N_{t+i-1}^{p}} \right) \Phi_{t} \\ &\geq \left(\frac{\mathbb{E}_{t} m_{t+j}^{p}}{\mathbb{E}_{t} N_{t+j}^{p}} / \frac{m_{t}^{p}}{N_{t}^{p}} \right) \Phi_{t} = \left(\frac{m_{t}^{p}}{m_{t}^{p}} / \frac{N_{t}^{p}}{N_{t}^{p}} \right) \Phi_{t} = \Phi_{t}, \end{split}$$

since, by the law of iterated expectations, $\mathbb{E}_t m_{t+i}^p = m_t^p$, and likewise for N_t^p .

⁴⁷ In the following, I use

$$\begin{split} \mathbb{E}_{t} \Phi_{t+j} &= \mathbb{E}_{t} \frac{m_{t+j}^{c}}{n_{t+j}^{p}} \Phi_{t+j-1} = \mathbb{E}_{t} \left(\frac{m_{t+j}^{c}}{m_{t+j-1}^{c}} / \frac{N_{t+j}^{p}}{N_{t+j-1}^{p}} \right) \Phi_{t+j-1} = \mathbb{E}_{t} \prod_{i=1}^{j} \left(\frac{m_{t+i}^{c}}{m_{t+i-1}^{c}} / \frac{N_{t+i}^{p}}{N_{t+i-1}^{p}} \right) \Phi_{t} \\ &\geq \left(\frac{\mathbb{E}_{t} m_{t+j}^{c}}{\mathbb{E}_{t} N_{t+j}^{p}} / \frac{m_{t}^{c}}{N_{t}^{p}} \right) \Phi_{t} = \left(\frac{m_{t}^{c}}{m_{t}^{c}} / \frac{N_{t}^{p}}{N_{t}^{p}} \right) \Phi_{t} = \Phi_{t}, \end{split}$$

since, by the law of iterated expectations, $\mathbb{E}_t m_{t+i}^c = m_t^c$, and likewise for N_t^p .

References

- Acemoglu, Daron, Philippe Aghion, Leonardo Bursztyn, and David Hemous. 2012. The environment and directed technical change. *American Economic Review* 102: 131–66. [CrossRef] [PubMed]
- Anderson, Evan W., William Brock, Lars Peter Hansen, and Alan H. Sanstad. 2013. Robust analytical and computational explorations of coupled economic-climate models with carbon-climate response. *SSRN Electronic Journal* 1–43. [CrossRef]
- Atkeson, Andrew, Varadarajan V. Chari, and Patrick J. Kehoe. 1999. Taxing capital income: A bad idea. Federal Reserve Bank of Minneapolis Quarterly Review 23: 3–17. Available online: https://ideas.repec.org/a/fip/fedmqr/y1999isump3-17nv.23no.3.html (accessed on 28 September 2022).
- Bansal, Ravi, and Amir Yaron. 2004. Risks for the long run: A potential resolution of asset pricing puzzles. *Journal of Finance* 59: 1481–509. [CrossRef]
- Barnett, Michael, William Brock, and Lars Peter Hansen. 2020. Pricing uncertainty induced by climate change. *Review of Financial Studies* 33: 1024–66. [CrossRef]
- Barrage, Lint. 2018. Be careful what you calibrate for: Social discounting in general equilibrium. *Journal of Public Economics* 160: 33–49. [CrossRef]
- Barrage, Lint. 2020. Optimal dynamic carbon taxes in a climate-economy model with distortionary fiscal policy. *Review of Economic Studies* 87: 1–39. [CrossRef]
- BEA. 2019. Table 1.1.6. Real Gross Domestic Product, Chained Dollars; Suitland: Bureau of Economic Analysis, U.S. Department of Commerce. Available online: https://apps.bea.gov/iTable/iTable.cfm? (accessed on 28 September 2022).
- Belfiori, Maria Elisa. 2015. Time consistent climate policies. In *Political Economy And Instruments of Environmental Politics*. Edited by Friedrich Schneider, Andrea Kollmann and Johannes Reichl. Cambridge: MIT Press, Chapter 12.
- Belfiori, Maria Elisa. 2017. Carbon trading, carbon taxes and social discounting. European Economic Review 96: 1–17. [CrossRef]
- Belfiori, Maria Elisa. 2018. Climate change and intergenerational equity: Revisiting the uniform taxation principle on carbon energy inputs. *Energy Policy* 21: 292–99. [CrossRef]
- Berger, Loïc, Johannes Emmerling, and Massimo Tavoni. 2016. Managing catastrophic climate risks under model uncertainty aversion. *Managemant Science* 57: 1–17. [CrossRef]
- Brock, William A., and Anastasios Xepapadeas. 2017. Climate change policy under polar amplification. *European Economic Review* 94: 263–82. [CrossRef]

- Brock, William A., and Steven N. Durlauf. 2015. On sturdy policy evaluation. The Journal of Legal Studies 44: 263–82. [CrossRef]
- Burke, Marshall, Solomon M. Hsiang, and Edward Miguel. 2015. Global non-linear effect of temperature on economic production. *Nature* 52: 235–39. [CrossRef] [PubMed]
- Cai, Yongyang, Kenneth L. Judd, and Thomas S. Lontzek. 2013. *The Social Cost of Stochastic and Irreversible Climate Change*. Technical report, NBER Working Paper 18704. Cambridge: NBER. [CrossRef]
- Cai, Yongyang, and Thomas S. Lontzek. 2019. The social cost of carbon with economic and climate risks. *Journal of Political Economy* 127: 2684–734. [CrossRef]
- Chamley, Christophe. 1986. Optimal taxation of capital income in general equilibrium with infinite lives. *Econometrica* 54: 607–22. [CrossRef]
- Chari, V. V., Larence J. Christiano, and Patrick J. Kehoe. 1994. Optimal fiscal policy in a business cycle model. *The Journal of Political Economy* 102: 617–52. Available online: https://jstor.org/stable/2138759 (accessed on 28 September 2022). [CrossRef]
- Collins, Matthew, Reto Knutti, Julie Arblaster, Jean-Louis Dufresne, Thierry Fichefet, Pierre Friedlingstein, Xuejie Gao, William J. Gutowski, Tim Johns, Gerhard Krinner, and et al. 2013. Long-term climate change: Projections, commitments and irreversibility. In *Climate Change 2013: The Physical Science Basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change*. Edited by Thomas F. Stocker, Dahe Qin, Gian-Kasper Plattner, Melinda M. B. Tignor, Simon K. Allen, Judith Boschung, Alexander Nauels, Yu Xia, Vincent Bex and Pauline M. Midgley. Chapter Long-term Climate Change: Projections, Commitments and Irreversibility. Cambridge and New York: Cambridge University Press. Available online: https://www.ipcc.ch/site/assets/uploads/2018/02/WG1AR5_Chapter12_FINAL.pdf (accessed on 28 September 2022).
- Dasgupta, Shouro, Nicole van Maanen, Franziska Piontek, Christian Otto, and Carl-Friedrich Schleussner. 2021. Effects of climate change on combined labour productivity and supply: An empirical, multi-model study. *The Lancet* 5: E455–65. [CrossRef]
- Dietz, Simon, and Falk Niehörster. 2019. *Pricing Ambiguity in Catastrophe Risk Insurance*. Technical Report 354. Washington, DC: Resources for the Future.
- Dietz, Simon, and Frank Venmans. 2019. Cumulative carbon emissions and economic policy: In search of general principles. *Journal of Environmnetal Economics and Management* 96: 108–29. [CrossRef]
- Dietz, Simon, James Rising, Thomas Stoerk, and Gernot Wagner. 2021. Economic impacts of tipping points in the climate system. *Proceedings of the National Academy of Sciences of the United States of America* 118: e2103081118. [CrossRef] [PubMed]
- Donadelli, Michael, Marcus Jüppner, Max Riedel, and Christian Schlag. 2017. Temperature shocks and welfare costs. *Journal of Economic Dynamics and Control* 82: 331–55. [CrossRef]
- Emmerling, Johannes, Laurent Drouet, Lara Aleluia Reis, Michela Bevione, Loic Berger, Valentina Bosetti, Samuel Carrara, Enrica De Cian, Gauthier De MaereD'Aertrycke, Tom Longden, and et al. 2016. *Witch 2016 Model—Documentation and Implementation of the Shared Socioeconomic Pathways*. Technical Report 2016.42. Milan: Fondazione Eni Enrico Mattei. Available online: https://papers.ssrn.com/sol3/papers.cfm?abstract_id=2800970 (accessed on 28 September 2022).
- EPA, US Government. 2017. Climate Impacts on Energy. Available online: https://19january2017snapshot.epa.gov/climate-impacts/ climate-impacts-energy_.html (accessed on 28 September 2022)
- Epstein, Larry G., and Stanley E. Zin. 1989. Substitution, risk aversion and the temporal behavior of consumption and asset returns: A theoretical framework. *Econometrica* 57: 937–69. [CrossRef]
- Epstein, Larry G., and Stanley E. Zin. 1991. Substitution, risk aversion and the temporal behavior of consumption and asset returns: An empirical analysis. *Journal of Political Ecnomy* 99: 263–86. [CrossRef]
- Frankhouser, Samuael, and Richard S. J. Tol. 2005. On climate change and economic growth. *Resource and Energy Economics* 77: 1–17. [CrossRef]
- Gazzotti, Paolo, Johannes Emmerling, Andrea Castelletti, Kaj-Ivar van der Wijst, Andries Hof, and Massimo Tavoni. 2021. Persistent inequality in economically optimal climate policies. *Nature Communications* 12: 3421. Available online: https://www.nature.com/articles/s41467-021-23613-y (accessed on 28 September 2022). [CrossRef]
- Gilboa, Itzhak, and David Schmeidler. 1989. Maxmin expected utility with non-unique prior. *Journal of Mathematical Economics* 18: 141–53. [CrossRef]
- Golosov, Mikhail, John Hassler, Per Krusell, and Aleh Tsyvinski. 2014. Optimal taxes on fossil fuel in general equilibrium. *Econometrica* 82: 41–88. [CrossRef]
- Hansen, Lars Peter, and Thomas J. Sargent. 1995. Discounted linear exponential quadratic gaussian control. *IEEE Transactions on Automatic Control* 40: 60–66. [CrossRef]
- Hansen, Lars Peter, and Thomas J. Sargent. 2001. Robust control and model uncertainty. *American Economic Review* 91: 60–68. Available online: https://www.jstor.org/stable/2677734 (accessed on 28 September 2022). [CrossRef]
- Hansen, Lars Peter, and Thomas J. Sargent. 2005. Robust estimation and control under commitment. *Journal of Economic Theory* 124: 258–301. [CrossRef]
- Hansen, Lars Peter, and Thomas J. Sargent. 2007. Recursive robust estimation without commitment. *Journal of Economic Theory* 136: 1–27. [CrossRef]
- Hansen, Lars Peter, and Thomas J. Sargent. 2008. Robustness. Princeton: Princeton University Press.
- Hansen, Lars Peter, and Thomas J. Sargent. 2012. Three types of ambiguity. Journal of Monetary Economics 59: 422–45. [CrossRef]
- Hassler, John, Per Krusell, and Anthony Smith. 2016. Environmental Macroeconomics. In *Handbook of Macroeconomics*. Edited by John B. Taylor and Harald Uhlig. Amsterdam: Elsevier, pp. 1893–2008.

- Heal, Geoffrey, and Antony Millner. 2013. Uncertainty and Decision in Climate Change Economics. Technical Report 128. London: Centre for Climate Change Economics and Policy, University of Leeds. Available online: https://ideas.repec.org/p/lsg/lsgwps/wp108 .html (accessed on 28 September 2022).
- Hennlock, Magnus. 2009. *Robust Control in Global Warming Management: An Analytical Dynamic Integrated Assessment*. Technical Report 09-19. Washington, DC: Resources for the Future.
- Hotelling, Harald. 1931. The economics of exhaustible resources. *Journal of Political Economy* 39: 137–75. Available online: https://www.jstor.org/stable/1822328 (accessed on 28 September 2022). [CrossRef]
- Hsiang, Solomon, and Robert E. Kopp. 2018. An economists guide to climate change science. *Journal of Economic Perspectives* 32: 3–32. [CrossRef]
- Jacobson, David H. 1973. Optimal stochastic linear systems with exponential performance criteria and their relation to deterministic differential games. *IEEE Transactions on Automatic Control* 18: 124–31. [CrossRef]
- Judd, Kenneth L. 1985. Redistributive taxation in a simple perfect foresight model. Journal of Public Economics 28: 59–83. [CrossRef]
- Karantounias, Anastasios. 2020. *Doubts about the Model and Optimal Policy*. Technical Report 2012-12. Atlanta: Federal Reserve Bank of Atlanta. [CrossRef]
- Karantounias, Anastasios G. 2013. Managing pessimistic expectations and fiscal policy. Theoretical Economics 8: 193–231. [CrossRef]
- Kaufmann, Robert K., Michael L. Mann, Sucharita Gopal, Jackie A. Liederman, Peter D. Howe, Felix Pretis, Xiaojing Tang, and Michelle Gilmore. 2017. Spatial heterogeneity of climate change as an experiential basis for skepticism. *Proceedings of the National Acadamy* of Science of the United States of America 114: 67–71. [CrossRef]
- Kjellstrom, Tord. 2014. Productivity Losses Ignored in Economic Analysis of Climate Change. United Nations University. Available online: https://unu.edu/publications/articles/productivity-losses-ignored-in-economic-analysis-of-climate-change. html (accessed on 28 September 2022).
- Kjellstrom, Tord, R. Sari Kovats, Simon J. Lloyd, Tom Holt, and Richard S. J. Tol. 2009. The direct impact of climate change on regional labor productivity. *Arch Environ Occup Health* 64: 217–27. [CrossRef]
- Klibanoff, Peter, Massimo Marinacci, and Sujoy Mukerji. 2005. A smooth model of decision making under ambiguity. *Econometrica* 73: 1849–92. [CrossRef]
- Klibanoff, Peter, Massimo Marinacci, and Sujoy Mukerji. 2009. Recursive smooth ambiguity preferences. *Journal of Economic Theory* 144: 930–76. [CrossRef]
- Kullback, Solomon, and Richard A. Leibler. 1951. On information and sufficiency. Annals of Mathematical Statistics 22: 79-86. [CrossRef]
- Kunreuther, Howard, Ralph Ginsberg, Louis Miller, Philip Sagi, Paul Slovic, Bradley Borkan, and Norman Katz. 1978. *Disaster Insurance Protection: Public Policy Lessons*. New York: Wiley.
- Lemoine, Derek, and Christian P. Traeger. 2016. Ambiguous tipping points. *Journal of Economic Behavior and Organization* 132: 5–18. [CrossRef]
- Lifton, Robert J. 2017. *The Climate Swerve*. New York: The New Press. Available online: https://thenewpress.com/books/climateswerve (accessed on 28 September 2022).
- Li, Xin, Borghan Narajabad, and Ted Temzelides. 2016. Robust dynamic energy use and climate change. *Quantitative Economics* 7: 821–57. [CrossRef]
- Maccheroni, Fabio, Massimo Marinacci, and Aldo Rustichini. 2006. Ambiguity aversion, robustness, and the variational representation of preferences. *Econometrica* 74: 1468–98. [CrossRef]
- Maldonado, Franco, and Kevin P. Gallagher. 2022. *Climate Change and IMF Sustainability Analysis*. Technical report. Washington, DC: The Task Force on Climate, International Monetary Fund. Available online: https://www.bu.edu/gdp/files/2022/02/TF_WP_ 005_FIN.pdf (accessed on 28 September 2022).
- Marlon, Jennifer, Peter Howe, Matto Mildenberger, Anthony Leiserowitz, and Xinran Wang. 2020. Yale climate opinion maps u.s. 2020. Yale Program on Climate Change. Available online: https://climatecommunication.yale.edu/visualizations-data/ycom-us/ (accessed on 28 September 2022).
- Matthews, H. Damon, Susan Solomon, and Raymond Pierrehumbert. 2012. Cumulative carbon as a policy framework for achieving climate stabilization. *Philosophical Transactions of the Royal Society A* 370: 4365–79. [CrossRef]
- Meyer, Robert, and Howard Kunreuther. 2017. *The Ostrich Paradox*. Philadelphia: Wharton Digital Press. Available online: https://wsp.wharton.upenn.edu/book/ostrich-paradox/ (accessed on 28 September 2022).
- Millner, Anthony, Simon Dietz, and Geoffrey Heal. 2010. *Ambiguity and Climate Policy*. Technical Report 16050. Cambridge: National Bureau of Economic Research. Available online: https://www.nber.org/system/files/working_papers/w16050/w16050.pdf (accessed on 28 September 2022).
- Millner, Antony, Geoffrey Heal, and Simon Dietz. 2012. Scientific ambiguity and climate policy. *Environmental and Resource Economics* 55: 21–46. [CrossRef]
- Moore, Frances C., and Delavane B. Diaz. 2015. Temperature impacts on economic growth warrant stringent mitigation policy. *Nature Climate Change* 5: 127–31. [CrossRef]
- Njuki, Eric, Boris E. Bravo-Ureta, and Víctor E. Cabrera. 2020. Climatic effects and total factor productivity: Econometric evidence for wisconsin dairy farms. *European Review of Agricultural Economicss* 47: 1276–301. [CrossRef]
- Nordhaus, William D. 1993. Rolling the 'dice': An optimal transition path for controlling greenhouse gasses. *Resource and Energy Economics* 15: 27–50. [CrossRef]

- Nordhaus, William D. 2007. *Estimates of the Social Cost of Carbon: Background and Results from the Rice-2011 Model*. Technical Report 1826. Cowles Foundation Discussion Paper. New Haven and London: Yale University Press. [CrossRef]
- Nordhaus, William D. 2008. A Question of Balance: Weighting the Options on Global Warming Policies. New Haven and London: Yale University Press. Available online: https://www.jstor.org/stable/j.ctt1npzkh (accessed on 28 September 2022).
- Pascal, Blaise. 1670. Pensees. London: W F Trotter.
- Pindyck, Robert. 2007. Uncertainty in environmental economics. *Review of Environmental Economics and Policy* 1: 45–65. Available online: https://ssrn.com/abstract=949761 (accessed on 28 September 2022). [CrossRef]
- Pindyck, Robert S. 2013a. Climate change policy: What do the models tell us? *Journal of Economic Literature* 51: 860–72. Available online: https://www.researchgate.net/publication/275930356_Climate_Change_Policy_What_Do_the_Models_Tell_Us (accessed on 28 September 2022). [CrossRef]
- Pindyck, Robert S. 2013b. The climate policy dilemma. *Review of Environmental Economics and Policy* 7: 219–37. Available online: https://ideas.repec.org/a/oup/renvpo/v7y2013i2p219-237.html (accessed on 28 September 2022). [CrossRef]
- Pindyck, Robert S. 2017. The use and misuse of models for climate policy. *Review of Environmental Economics and Policy* 11: 100–14. Available online: https://ssrn.com/abstract=2596426 (accessed on 28 September 2022). [CrossRef]
- Rezai, Armon, and Rick van der Ploeg. 2017. Climate policies under climate model uncertainty: Max-min and min-max regret. *Energy Economics* 68: 4–16. [CrossRef]
- Rezai, Armon, and Rick van der Ploeg. 2019. The agnostic's response to climate deniers: Price carbon! *European Economic Review* 111: 70–84. [CrossRef]
- Roe, Gerard H., and Marcia B. Baker. 2007. Why is climate sensitivity so unpredictable? Science 318: 629–32. [CrossRef]
- Somanathan, E., Rohini Somanathan, Anant Sudarshan, and Meenu Tewari. 2018. The Impact of Temperature on Productivity and Labor Supply: Evidence from Indian Manufacturing. Energy Policy Institute, University of Chicago. Available online: https://bfi.uchicago.edu/working-paper/the-impact-of-temperature-on-productivity-and-labor-supply-evidence-fromindian-manufacturing/ (accessed on 28 September 2022).
- Stern, Lord Nicholas Stern. 2007. *The Economics of Climate Change: The Stern Review*. Cambridge: Cambridge University Press. Available online: https://www.lse.ac.uk/granthaminstitute/publication/the-economics-of-climate-change-the-stern-review/ (accessed on 28 September 2022).
- Strzalecki, Tomasz. 2011. Axiomatic foundation of multiplier preferences. Econometrica 79: 47–43. [CrossRef]
- Tetlow, Robert J., and Peter von zur Muehlen. 2009. Robustifying learnability. *Journal of Economic Dynamics and Control* 33: 296–316. [CrossRef]
- Than, Ker. 2015. Estimated Social Cost of Climate Change Not Be Accurate, Stanford Scientists Say. *Stanford News*. Available online: http://news.stanford.edu/2015/01/12/emissions-social-costs-011215/ (accessed on 28 September 2022).
- Traeger, Christian P. 2014. Why uncertainty matters—Discounting under intertemporal risk aversion and ambiguity. *Economic Theory* 56: 627–64. [CrossRef]
- Traeger, Christian P. 2018. Ace—Analytic Climate Economy (with Temperature and Uncertainty). Technical Report. Available online: https://ssrn.com/abstract=3307622 (accessed on 28 September 2022).
- van der Ploeg, Frederick. 1984. Economic policy rules for risk-sensitive decision making. Zeitschrift für Nationalökonomie 44: 207–35. [CrossRef]
- van der Ploeg, Frederick, and Cees Withagen. 2014. Growth, renewables, and the optimal carbon tax. *International Economic Review* 55: 283–311. [CrossRef]
- von Below, David. 2012. Optimal Carbon Taxes with Social and Private Discounting. Unpublished manuscript. Available online: https://ethz.ch/content/dam/ethz/special-interest/mtec/cer-eth/resource-econ-dam/documents/research/sured/ sured-2012/SURED-12_170_vonBelow.pdf (accessed on 28 September 2022).
- von zur Muehlen, Peter. 1982. Activist vs. Non-Activist Monetary Policy: Optimal Rules under Extreme Uncertainty. Available online: https://ssrn.com/abstract=2735777 (accessed on 28 September 2022).
- Weil, Philippe. 1990. Non-expected utility in macroeconomics. Quarterly Journal Economics CV 1: 29-42. [CrossRef]
- Weitzman, Martin L. 2007. A review of the *Stern Review* on the economics of climate change. *Journal of Economic Literature* 45: 703–24. [CrossRef]
- Weitzman, Martin L. 2009. On modeling and interpreting the economics of catastrophic climate change. *Review of Economics and Statistics* 91: 1–19. [CrossRef]
- Weitzman, Martin L. 2013. Tail-hedge discounting and the social cost of carbon. *Journal of Economic Literature* 51: 873–82. [CrossRef] Whittle, Peter. 1981. Risk sensitive linear quadratic gaussian control. *Advances in Applied Probability* 13: 764–77. [CrossRef]
- Wilbanks, Thomas J., Vatsal Bhatt, Daniel E. Bilello, Stanley R. Bull, James Ekmann, William C. Horak, Y. Joe Huang, Mark D. Levine, Michael J. Sale, David K. Schmalzer, and et al. 2008. *Effects of Climate Change on Energy Production and Use in the United States*. Washington, DC: US Climate Change Assessment Program: Synthesis and Assessment Product 4.5, National Technology and Science Council, US Climate Change Science Program, US Department of Energy. Available online: https://digitalcommons.unl.edu/usdoepub/12/ (accessed on 28 September 2022).
- Willner, Sven N., Nicole Glanemann, and Anders Levermann. 2021. Investment incentive reduced by climate damages can be restored by optimal policy. *Nature Communications* 12: 3245. [CrossRef]

Woodford, Michael. 2010. Robustly optimal monetary policy with near-rational expectations. *American Economic Review* 100: 274–303. [CrossRef]

Workman, Mark, Geoff Darch, Kate Dooley, Guy Lomax, James Maltby, and Hector Pollitt. 2021. Climate policy decision making in contexts of deep uncertainty - from optimisation to robustness. *Environmental Science and Policy* 120: 127–37. [CrossRef]

Zenios, Stavros A. 2022. The risks from climate change to sovereign debt. Climatic Change 172: 1. [CrossRef]

Zhang, Peng, Olivier Deschenes, Kyle C. Meng, and Junjie Zhang. 2017. *Temperature Effects on Productivity and Factor Reallocation: Evidence from a Half Million Chinese Manufacturing Plants*. Technical Report IZA DP No. 11132. Bonn: IZA Institute of Labor Economics. [CrossRef]

Zhu, Xiaodong. 1992. Optimal fiscal policy in a stochastic growth model. Journal of Economic Theory 58: 250-89. [CrossRef]